# A MULTI-PARAMETER EMPIRICAL MODEL FOR MESOPHILIC ANAEROBIC DIGESTION

by

### EMMANUEL CHUKWUMA OGBONNA

# A thesis submitted to the University of Hertfordshire

# in partial fulfilment of the requirements for the degree of

Doctor of Philosophy

July 2016

### Foreword

The work presented in this thesis is conducted at the School of Engineering and Technology, Science & Technology Research Institute, University of Hertfordshire, UK, from October 2010 to January 2015, under the supervision of Dr Rajnish Calay, Dr Rashid Ali, Dr Zhijun Peng and Dr Georgios Pissanidis.

The research work reported in this thesis sets out to develop a black-box (empirical) model for the mesophilic anaerobic digestion (AD) process. The goal is to have a better understanding of how physico-environmental operating parameters influence biogas and methane production. Experimental work is carried out to determine the effects of four operating parameters on biogas and methane production in a laboratory scale. In addition, the data acquired from the experiments is utilised to construct the mesophilic AD models for this research study, in MATLAB through System Identification method.

The introduction of this thesis in Chapter 1 presents the theme of the research work. It highlighted the irreversible impact of global energy production and consumption, especially, fossil fuels, on the environment and all life forms, as well as discussed the possible alternative energy sources that can mitigate these risks. In addition, this part discussed the environmental and health risks resulting from increasing municipal solid waste (MSW) generation due to rising human population and industrial advancement. Furthermore, it introduced the concept of AD, which is the process by which bacteria breakdown organic materials in the absence of oxygen to produce biogas and digestate (biofertiliser). Massarutto (2010) suggests that AD is the ideal technology for sustainable management of organic materials to produce a source of

renewable energy, and an effective waste management technique that provides economic, health and environmental benefits.

Following the introduction is a review of scientific articles relating to AD process (Chapter 2). This discussed some of the operating parameters that influence the activities of anaerobic bacteria. The improvement and current state of in AD are also reviewed in Chapter 3. Chapter 4 discusses the complex interaction of different species of anaerobic bacteria that constitute the four steps of digestion process as well as the experimental work conducted in this work. Chapter 5 discusses the collection and analysis of the experimental results, considering the effects of temperature, pH, mixing speed and pressure on biogas and methane production while Chapter 6 discusses the construction of black-box model through MATLAB System Identification Toolbox. Two nonlinear model structures, autoregressive with exogenous input (NARX) and Hammerstein-Wiener (NLHW) with different nonlinearity estimators and model orders are chosen by trial and error and utilised to estimate the models. The performance of the models is determined by comparing the simulated outputs of the estimated models and the output in the validation data. The approach is used to validate the estimated models by checking how well the simulated output of the models fits the measured output. The best models for biogas and methane production are chosen by comparing the outputs of the best NARX and NLHW models (each for biogas and methane production), and the validation data, as well as utilising the Akaike information criterion (AIC), to measure the quality of each model relative to each of the other models. The NLHW models mhw2 and mhws2 are chosen for biogas and methane production, respectively. The identified NLHW models mhw2 and mhws2 represent the behaviour of the production of biogas and methane, respectively, from mesophilic AD. Of all the candidate models studied, the nonlinear models provide a superior reproduction of the experimental data over the whole analysed period. (Chapter 6).

The thesis also discussed the scalability of black-box models. Finally, conclusions and recommendation for further research work are covered in Chapter 7.

This thesis presents the most vital outcomes of this work, which has been partially published as a conference paper in a journal, and the citation of the paper is as follows: Ogbonna E. C., Ali R., Pissanidis G. (2013): Simulation Model for Mesophilic Anaerobic Digestion Heating System. Renewable Energy Research and Applications (ICRERA), 2013

International Conference: 505-510, doi: 10.1109/ICRERA.2013.6749807. The full copy of this paper is located in Appendix C.

### Acknowledgement

I would like to express my profound gratitude to the almighty God who has provided for me in every way to make this PhD study a success. I would also like to express my special appreciation and thanks to my two supervisors, Dr Zhijun Peng and Dr Georgios Pissanidis for the encouragement, advice and guidance you provided for me throughout this study. I would like to thank Professor David Stuckey of Imperial College London for your support. I would also like to thank Professor Talip Alukaidey and Dr Iman Mansouri both of the University of Hertfordshire, for the encouragement, advice, inspiration and suggestions you provided for me throughout this work. In addition, I would like to express my gratitude to Mr Pete Pearce of Thames Water Innovation Centre for providing me with the inoculum I used in this study. Furthermore, I would like to thank all the members of staff and research students at the University of Hertfordshire who helped me in one way or the other to attain this goal.

I would like to express a special thanks and appreciation to my beloved wife Chizor for your continued support and encouragement, even in the moments of despair when there was no one else to turn to. Your prayer and love have sustained me thus far. I would also like to express my gratitude to my two daughters, Chinechem and Esomchi for their patience in those times I was too busy to attend to them. Finally, I would like to thank my family members and friends for their support and prayers. This work would have been more difficult without you.

#### Emmanuel C. Ogbonna

January 2016

### Abstract

Anaerobic digestion, which is the process by which bacteria breakdown organic matter to produce biogas (renewable energy source) and digestate (biofertiliser) in the absence of oxygen, proves to be the ideal concept not only for sustainable energy provision but also for effective organic waste management. However, the production amount of biogas to keep up with the global demand is limited by the underperformance in the system implementing the AD process. This underperformance is due to the difficulty in obtaining and maintaining the optimal operating parameters/states for anaerobic bacteria to thrive with regards to attaining a specific critical population number, which results in maximising the biogas production. This problem continues to exist as a result of insufficient knowledge of the interactions between the operating parameters and bacterial community. In addition, the lack of sufficient knowledge of the composition of bacterial groups that varies with changes in the operating parameters such as temperature, substrate and retention time. Without sufficient knowledge of the overall impact of the physico-environmental operating parameters on anaerobic bacterial growth and composition, significant improvement of biogas production may be difficult to attain.

In order to mitigate this problem, this study has presented a nonlinear multi-parameter system modelling of mesophilic AD. It utilised raw data sets generated from laboratory experimentation of the influence of four operating parameters, temperature, pH, mixing speed and pressure on biogas and methane production, signifying that this is a multiple input single output (MISO) system. Due to the nonlinear characteristics of the data, the nonlinear blackbox modelling technique is applied. The modelling is performed in MATLAB through System Identification approach. Two nonlinear model structures, autoregressive with exogenous input (NARX) and Hammerstein-Wiener (NLHW) with different nonlinearity estimators and model

orders are chosen by trial and error and utilised to estimate the models. The performance of the models is determined by comparing the simulated outputs of the estimated models and the output in the validation data. The approach is used to validate the estimated models by checking how well the simulated output of the models fits the measured output. The best models for biogas and methane production are chosen by comparing the outputs of the best NARX and NLHW models (each for biogas and methane production), and the validation data, as well as utilising the Akaike information criterion to measure the quality of each model relative to each of the other models. The NLHW models mhw2 and mhws2 are chosen for biogas and methane production, respectively. The identified NLHW models mhw2 and mhws2 represent the behaviour of the production of biogas and methane, respectively, from mesophilic AD. Among all the candidate models studied, the nonlinear models provide a superior reproduction of the experimental data over the whole analysed period. Furthermore, the models constructed in this study cannot be used for scale-up purpose because they are not able to satisfy the rules and criteria for applying dimensional analysis to scale-up.

## **Table of Contents**

Foreword	i
Acknowledg	gementiv
Abstract	V
Table of Co	ntentsvii
List of Figu	res xiii
List of Tabl	esxvii
List of Abbi	reviations xviii
Nomenclatu	ırexxi
Chapter 1 I	ntroduction1
1.1 Introdu	uction1
1.2 Backg	round1
1.2.1	Energy and environmental problems1
1.2.2	Municipal solid waste generation and the world population
1.2.3	Evolution of waste management concepts7
1.2.4	Motivation of anaerobic digestion for organic waste management
1.3 Proble	m statement11
1.4 Aim a	nd objectives12
1.5 Summ	ary13
Chapter 2 D	Development in Anaerobic Digestion: Literature Review14
2.1 Introdu	uction14

2.2 Opera	ating parameters that influence AD performance	14
2.2.1	pH	15
2.2.2	Temperature	16
2.2.3	Carbon to Nitrogen (C/N) ratio	18
2.2.4	Substrate characteristics	19
2.2.5	Hydraulic retention time (HRT)	21
2.2.6	Organic loading rate (OLR)	22
2.2.7	Mixing	23
2.3 Types	s of anaerobic digester	23
2.3.1	Feeding mode (batch and continuous)	24
2.3.2	Solid content (wet and dry)	27
2.4 Summ	nary	
Chapter 3	Improvement and current state in AD: Literature Review	29
3.1 Introc	duction	29
3.2 Pre-tr	reatment for process enhancement	29
3.3 Co-di	gestion of different substrates	31
3.4 Proce	ess modelling	33
3.4.1	Mathematical models description	34
3.4.2	Linear black-box models	40
3.4.3	Nonlinear black-box models	44
3.5 Summ	nary	48

Chapter 4	Experimental Apparatus and Methodology49
4.1 Introd	luction
4.2 Bacte	priological and biochemical processes in AD49
4.3 Furth	er review of the operating parameters54
4.4 Expe	rimental procedures
4.4.1	Preparation of bacterial culture media
4.4.2	Impact of variable temperature on biogas production60
4.4.3	Effect of variable pH on biogas production
4.4.4	Influence of digester pressure on biogas and methane production
4.4.5	Impact of mixing speed on biogas production67
4.4.6	Effect of simultaneous manipulation of operating parameters
4.4.7	Influence of co-digestion on biogas production68
4.5 Sum	nary70
Chapter 5	Results and Discussions71
5.1 Introd	luction71
5.2 The i	mpact of temperature on biogas and methane production71
5.2.1	Average rate and cumulative biogas production71
5.2.2	Total volume of biogas production74
5.2.3	Methane production at varying temperature75
5.2.4	Average total volume of methane production at different temperatures77
5.3 Effec	t of pH78

	5.3.1	Rate and cumulative biogas production at various pH values78
	5.3.2	Average total volume of biogas production at different pH values
	5.3.3	Methane production rate and cumulative methane production at different pH
	values	81
	5.3.4	Average total volume of methane production at different pH values
5	.4 Influer	nce of digester pressure
	5.4.1	Rate and cumulative biogas production at different digester pressures
	5.4.2	Average total volume of biogas production at variable digester pressure
	5.4.3	Methane production rate and cumulative methane yield at different digester
	pressure	e 87
	5.4.4	Average methane concentration at different digester pressures
	5.4.5	Average total methane production at different digester pressures
5	.5 Effect	of mixing speed90
	5.5.1	Average rate and cumulative biogas production at various mixing speeds91
	5.5.2	Average total volume of biogas production at variable mixing speeds
	5.5.3	Average rate and cumulative methane production at various mixing speeds94
	5.5.4	Average total volume of methane production at variable mixing speeds95
5	.6 Influe	nce of simultaneous manipulation of multiple inputs on biogas and methane
p	roduction	n96
	5.6.1	Biogas and methane production96
	5.6.1a	Average rate and average cumulative biogas production
	5.6.1b	Average rate and average cumulative methane production for Experiment3198

5.7 Influe	ence of co-digestion on biogas and methane production	100
5.7.1	Average rate of biogas production at different co-digestion ratio	100
5.7.2	Average methane production rate at different co-substrate ratio	102
5.7.3	Average cumulative methane production at different co-substrate ratio	103
5.7.4	Observed process characteristics	104
5.8 Sum	mary	105
Chapter 6	Black-Box Model Construction	106
6.1 Intro	duction	106
6.2 Syste	em Identification modelling method	107
6.2.1	Choice of a suitable mathematical model	111
6.2.2	Black-box model identification procedure	113
6.2.3	NARX model estimation	120
6.2.4	Hammerstein-Wiener model estimator	130
6.2.5	Comparison between the biogas and the methane production models	147
6.3 Biop	rocess Scale-up	148
6.3.1	Scale-up of Black-box Model	150
6.4 Sumr	mary,	151
Chapter 7	Conclusions	153
7.1 Introd	duction	153
7.2 Obse	erved Outcomes	154
7.3 Limit	tation of this study	162

7.4 Recommendations for Further Work	
References	
Appendices	
Appendix A: Raw Experimental Data	
Appendix B: MATLAB Modelling Codes	
Appendix C: Published Conference Paper	

# **List of Figures**

Fig. 2.1. Relative growth rate of different methanogenic bacteria	17
Fig. 2.2. Single-stage digester	25
Fig. 2.3. Two or multi-stage digester	26
Fig. 3.1. Classification of process models	34
Fig. 3.2. Monod specific growth rate	37
Fig.3.3. Block diagram of black-box process	
Fig.3.4. Classification of black-box models	40
Fig. 3.5. Different linear black-box models	43
Fig.4.1. Stages in AD process	51
Fig. 4.2. Bacterial culture preparation	60
Fig. 4.3. Cross-section of New Brunswick BioFlo 111 Digester	62
Fig. 4.4. Setup of BioFlo 111 Digester	63
Fig. 4.5. Setup of co-digestion experiment	69
Fig. 5.1. Average rate of biogas production at different temperatures	72
Fig. 5.2. Average cumulative biogas production at different temperatures	73
Fig. 5.3. Average total volume of biogas yield at different temperatures	74
Fig. 5.4. Average rate of methane proportion at different temperatures	75
Fig. 5.5. Average cumulative methane production at different temperatures	76
Fig. 5.6. Average total volume of methane yield at variable temperature	77
Fig. 5.7. Average rate of biogas production at different pH values	78
Fig. 5.8. Average cumulative biogas production at different pH values	80
Fig. 5.9. Average total volume of biogas production at different pH values	81
Fig. 5.10. Average methane production rate at different pH values	

Fig. 5.11. Average cumulative methane production at different pH values	33
Fig. 5.12. Average total volume of methane production at different pH values	34
Fig. 5.13. Average rate of biogas production at different pressures	35
Fig. 5.14. Average cumulative biogas production at different pressure values	36
Fig. 5.15. Average total biogas production at different pressure values	37
Fig. 5.16. Average methane production rate at different pressures	38
Fig. 5.17. Average cumulative methane production at variable pressure	38
Fig. 5.18. Average methane proportion at different pressures	39
Fig. 5.19. Average total methane production different pressures	<del>)</del> 0
Fig. 5.20. Average rate of biogas production at various mixing speed9	<b>)</b> 1
Fig.5.21. Average cumulative biogas production at different mixing speed9	<del>)</del> 2
Fig. 5.22. Average total volume of biogas production at different mixing speeds9	<del>)</del> 3
Fig. 5.23. Average rate of methane production at various mixing speeds	94
Fig. 5.24. Average cumulative methane production at different mixing speeds9	<del>)</del> 5
Fig.5.25. Average total volume of methane production at different mixing speeds9	<del>)</del> 6
Fig. 5.26. Average rate of biogas production at simultaneous variance of input parameters9	<del>)</del> 7
Fig. 5.27. Average cumulative biogas production at simultaneous adjustment of input	
parameters9	<del>)</del> 8
Fig. 5.28. Average rate of methane production at simultaneous adjustment of input	
parameters9	<del>)</del> 9
Fig. 5.29. Average cumulative methane production at simultaneous adjustment of the	
operating parameters	)0
Fig. 5.30. Average rate of biogas production at different co-substrate ratio	)1
Fig. 5.31. Average cumulative biogas production at different co-substrate ratio10	)2
Fig. 5.32. Average methane production rate at various substrate ratio	)3

Fig. 5.33. Average cumulative methane production at different co-substrate ratio104
Fig. 6.1. Block diagram for model identification106
Fig. 6.2a. Flowchart of steps in System Identification modelling method108
Fig. 6.2b. Flowchart of the actual black-box model identification procedure114
Fig. 6.3. Input-output data plot for biogas production116
Fig. 6.4. Input-output data plot for methane production117
Fig.6.5. Block diagram of NARX models (Ljung, 2015; Shariff et al., 2014)
Fig. 6.6. Evaluation of estimated NARX models with different model moder for biogas
production124
Fig. 6.7. Evaluation of estimated NARX models with different number of units in the
'wavelet' estimator for biogas production
Fig. 6.8. Evaluation of NARX models with different nonlinearity estimators for biogas
production
Fig. 6.9. Evaluation of NARX models with different model order for methane production.127
Fig. 6.10. Evaluation of estimated NARX models with different number of units in 'wavelet'
estimator for methane production
Fig. 6.11. Evaluation of NARX models with different nonlinearity estimators for methane
production
Fig. 6.12. Block diagram of Hammerstein-Wiener model130
Fig. 6.13. Evaluation of NLHW models with pwlinear for both nonlinearities for biogas
production
Fig.6.14 Evaluation of NLHW models with unitgain and saturation nonlinearities for biogas
production
Fig. 6.15. Evaluation of NLHW model with saturation and deadzone nonlinearities for biogas
production

Fig. 6.16. Comparison of the best NLHW model for biogas brod	uction136
Fig. 6.17. Evaluation of estimated NLHW models with different	nonlinearity combinentios
for methane production	
Fig. 6.18. Comparison of the estimated NARX and NLHW mode	els for biogas production.139
Fig. 6.19. Comparison of the estimated NARX and NLHW mode	els for methane production
Fig. 6.20a. Simulation of percentage error from models for bioga	as and methane production
Fig. 6.20b Error estimator	

### **List of Tables**

Table 3.1. List of the modified Monod models	37
Table 4.1 Biogas 5000 technical specifications	54
Table 4.2. New Brunswick BioFlo 111 technical specifications	55
Table 6.1a. Test for biogas production model mhw2 for variables of temperature	16
Table 6.1b. Test for methane production model mhws2 for variables of temperature	16
Table 6.2a. Test result for biogas production model mhw2 for variables of pH	16
Table 6.2b. Test result for methane production model mhws2 for variables of pH	16

# **List of Abbreviations**

AD	Anaerobic Digestion
ADM1	Anaerobic Digestion model one
CaO	Calcium oxide
CFD	Computational Fluid Dynamic
CH <sub>4</sub>	Methane
$CO_2$	Carbon dioxide
COD/N	Chemical Oxygen Demand/Nitrogen
C/N ratio	Carbon to Nitrogen ratio
CSTR	Continuous Stirred Tank Reactor
DA	Dimensional Analysis
DEFRA	Department for Environment, Food and Rural Affairs
DAE	Differential Algebraic Equation
DE	Differential Equation
DN	Dimensionless Number
EC	European Commission
EEA	European Environment Agency
EPSAC	Extended Prediction Self-Adaptive Control
EU	European Union
Fig.	Figure
FOA	Food and Agriculture Organisation
FW	Food Waste
GoF	Goodness-of-Fit

GUI	Graphical User Interface
H <sub>2</sub>	Hydrogen
HCL	Hydrochloric Acid
H <sub>2</sub> O	Water
HPA	Health Protection Agency
HRT	Hydraulic Retention Time
$H_2S$	Hydrogen Sulphide
IEA	International Energy Agency
IEO	Information Energy Administration
IWA	International Water Association
LAP	Laboratory Analytical Procedure
LCFA	Long Chain Fatty Acid
MIMO	Multiple Input Multiple Output
MISO	Multiple Input Single Output
МО	Measured Output
MPC	Model Predictive Control
MSW	Municipal Solid Waste
MVs	Manipulated Variables
$N_2$	Nitrogen gas
NaOH	Sodium Hydroxide
NARX	Nonlinear Auto Regressive with Exogenous input
NH <sub>3</sub>	Ammonia
NI	National Instrument
NLHW	Nonlinear Hammerstein-Wiener
NNFCC	National Non-Food Crop Centre

NWCC	National Wind Coordinating Committee
OECD	Organisation for Economic Co-operation and Development
OFMSW	Organic Fraction of Municipal Solid Waste
OLR	Organic Loading Rate
pН	Power of Hydrogen
PhD	Doctor of Philosophy
Ref	Reference
SCADA	Supervisory Control and Data Acquisition
SNH	Scottish Natural Heritage
SSR	Sum of Squares of the Regression
SST	Total Sum of Squares
TOC	Total organic carbon
TS	Total Solids
UNFPA	United Nations Fund for Population Activities
VFAs	Volatile Fatty Acids
VS	Volatile Solids
WID	Waste Incineration Directive
WIR	Waste Incineration (England and Wales) Regulations
WWS	Wastewater Sludge

## Nomenclature

α	Intersection of line meets the y-axis,
β	Slope of the equation and,
3	Variation or random error of the actual data
у	Output
Х	measure of time or dosage
А	Starting or initial quantity
В	Relative growth rate
D	Dilution rate
$h^{-1}$	Per hour
<b>V</b> =	Volume of digester
m <sup>3</sup>	Metre cube
F	Flow rate
mg	Milligram
t	Time
1	Length
μ	Specific growth rate
$\mu_{ m max}$	Maximum specific growth rate
S	Substrate concentration
ks	Half-velocity constant
Y	Bacterial yield coefficient
$G_{s,}G$	Biogas production
Т	Lag time
S	Second

Ts	Sampling time
Р	prediction horizon
<i>y</i> ( <i>t</i> )	Output at time t
f	nonlinear function
$\varepsilon(t)$	noise signals
<i>u</i> ( <i>t</i> )	Input at time t
$n_y$	maximum lags of past output
$n_u$	maximum lags of past input
w(t)	input internal variables
x(t)	output internal variables
mL	millilitre
rpm	revolution per minute
L	Litre
°C	degree celcius
kWh	Kilowatt hour
Press	Pressure
Temp	Temperature
MS	Mixing Speed

### Chapter 1 Introduction

#### **1.1 Introduction**

This chapter presents the research work theme, highlighting the impact of global energy production and consumption, especially, the effects of fossil fuels on the environment and all life forms. It also discussed the possible alternative energy sources that can mitigate the environmental and health risks linked to fossil fuels. In addition, it discussed the environmental and health risks due to increasing municipal solid waste (MSW) generation. Furthermore, the concept of AD is introduced, including its suitability for providing a sustainable energy source as well as implemented as an ideal organic waste management system. Finally, the problem statement, aim and objectives of this research work are discussed.

#### **1.2 Background**

#### **1.2.1** Energy and environmental problems

Energy and its impact on the environment are interrelated (Ahuja and Tatsutani, 2009). The environmental impacts of energy production and energy utilisation are irreversible (Sidik et al., 2013, Budiyono et al., 2013; Abdullahi et al., 2011; Espinoza-Escalante et al., 2009; Tejada and Gonzalez, 2006). First, energy consumption continues to escalate year after year due to increasing human population, rising living standards and rapid growth of energy intensive industries in emerging economies (OECD, 2011; Ahuja and Tatsutani, 2009). Global reserves of fossil fuels, our primary source of energy, are finite and are being depleted (Sidik et al., 2013; Budiyono et al., 2013, OECD, 2011).

Second, the conversion of the chemical energy of fossil fuels to useful work involves combustion, the products of which include greenhouse gases such as CO<sub>2</sub> and NO<sub>2</sub> (NRC,

2010; Tsoutsos et al., 2005). Emissions of these gases are not only harmful to many life forms but also contribute to the cause of climate change (Budiyono et al., 2013, OECD, 2011). With ever increasing energy consumption we are exhausting our reserves of fossil fuels, intensifying the deterioration of the quality of the air we breathe and increasing global warming.

In order to mitigate the two aforementioned risks, there is a need to explore alternative sources of energy that are sustainable. The potentials of sustainable energy sources such as solar, wind, geothermal and biomass, if fully harvested, could provide satisfactory solutions to the global energy and environmental challenges (EC, 1997; EC, 1995). The renewable energy sources are discussed as follows.

#### Solar energy:

Solar could be the world's ideal source of energy, it is free and inexhaustible (Tudorache and Kreindler, 2010). It has the capacity to provide more than 10,000 times the global annual energy requirement (Tsoutsos et al., 2005; Greenpeace, 2005). Not only is solar abundant and infinite, it also offers substantial environmental benefits relative to fossil fuels, our primary energy sources (Tsoutsos et al., 2005; Tsoutsos et al., 2003a; Tsoutsos, 2001; Boyle, 1996; Johansson and Burnham, 1993). However, solar systems are linked to a number of adverse environmental impacts such as land use, visual impact and effects on buildings (Hestnes, 1999; OECD/IEA, 1998; Boyle, 1996).

In addition, solar energy is not available at all time. To generate power from solar, there must be sunlight, which makes it impossible to produce power at night or even in some hours of daylight in places situated in the temperate, continental and polar climates. Solar energy in these climates decreases during the winter months as there are fewer sunlight hours, resulting in the reduction of the intensity of solar radiation. In these circumstances, energy must be stored or obtained elsewhere at night or during the winter months when there is no sunshine.

#### Wind energy:

The Wind is another potential source of energy that is gaining prominence in the world (UCS, 2013). It is one of the ideal sources of renewable energy (UCS, 2013). It is infinite, plentiful and produces no toxic contamination or greenhouse gases (UCS, 2013). Although the wind offers many possibilities to replace the conventional sources of energy, yet it has a number of undesirable impacts on the environment. These include land use that varies from site to site (NREL, 2012; NREL, 2010; Denholm et al., 2009; Michel et al., 2007), impact of wind turbines on wildlife and habitat, most especially bats and birds' death (NWCC, 2010) and effect on public health, especially sound and visual disturbances.

Like solar energy, wind energy has its own limitations; it is unpredictable. Wind turbines cannot generate electricity when any wind is blowing, thus making wind energy not suitable for a single source energy solution (Rasmussen, 2010). In addition, wind energy is sited typically in coastal and hilly areas (SNH, 2014). Cities situated within the plane topographies will not be able to harness wind energy as much as in coastal and hilly geographies, making wind energy non-universal.

#### **Geothermal energy:**

Geothermal energy is another source of renewable energy. It is from the heat confined inside the earth crust (Gagel et al., 2007; Barbier, 2002). It can be utilised to generate electricity when the temperature is above 150 °C. On the other hand, geothermal energy can be applied directly for space heating and industrial processes when the temperature is below 150 °C (Fridleifsson, 2003; Barbier 2002; Fridleifsson, 1996). Geothermal resources are estimated to be 2,000±140 Terawatts hour per annum (TWh/a) worth of electricity, as well as more than 7,000 TWh/a worth of heat that can be used for direct space heating or industrial applications (Fridleifsson, 2003; Fridleifsson, 2001; Stefansson, 1998). The International Energy Outlook 2013 (IEO, 2013) estimated that the total energy consumption of the world would be 184,634.73 TWh in 2020 and 240,318.22 TWh in 2040. This means that the global geothermal energy can only contribute about 5 % of the projected total world energy requirement for 2020 and 3.8 % for 2040 (IEO, 2013).

It is apparent that geothermal energy cannot meet the global energy requirement. There are a number of geological environments that support exploitable geothermal energy. The geothermal fields with a temperature above 150 °C suitable for electricity generation are mainly trapped in locations where there is young volcanism, seismic and magmatic action (Fridleifsson, 1996), which do not occur in every country. On the other hand, the geothermal resources with a temperature below 150 °C that could be utilised directly are found in many parts of the world (Fridleifsson, 1996). Furthermore, the exploitation of geothermal energy results to some environmental threats, such as emission of gases like, CO<sub>2</sub>, H<sub>2</sub>S, NH<sub>3</sub>, N<sub>2</sub>, H<sub>2</sub>, and CH<sub>4</sub> (Barbier, 2002; Fridleifsson, 2001), wildlife extinction, destruction of natural vegetation and land use (Barbier, 2002; Fridleifsson, 2001).

#### **Biomass energy:**

Biomass is also another source of renewable energy that has the promise to mitigate the impact of global energy production on the environment and resource conservation (Wang, 2014). The use of biomass to produce sustainable energy can replace fossil fuels and also minimises  $CO_2$ and  $CH_4$  emission, which contribute significantly to global warming (Kusch et al., 2011). The conversion of biomass to energy has a variety of techniques that are currently applied, which is broadly classified into thermochemical and biochemical/biological processes (Cheng et al., 2014; Cheng, 2013; Tonini, 2013; Caputo et al., 2005; McKendry, 2002). Thermochemical conversion process consists of combustion, gasification, pyrolysis, and liquefaction (Ciubota-Rosie et al., 2008; Caputo et al., 2005; Demirbas, 2004; McKendry, 2002; Demirbas, 2001; Demirbas, 1998; Overend, 1998; Solantausta, 1995). While biochemical/biological conversion method is made up of aerobic fermentation, anaerobic digestion (AD) and mechanical extraction (Ciubota-Rosie et al., 2008; Caputo et al., 2005; McKendry, 2002; Demirbas, 2001; WEC, 1994). However, AD is the ideal biomass energy conversion technology due the advantage of not only producing renewable energy source but also well suited for organic waste management in the municipalities and rural areas (Kusch et al., 2011).

Every energy conversion process, including biomass conversion concepts, have some degree of negative impacts on the environment. Biomass conversion techniques also have a number of environmental concerns, such as air pollution, deforestation, as well as the impact on the cultivation of food crops (Ciubota-Rosie et al., 2008). But, unlike any other source of sustainable energy, biomass energy can achieve a neutral CO<sub>2</sub> contribution to the environment (Ciubota-Rosie et al., 2008; Jefferson, 2006; Robu, 2005).

This is attained by creating a balance between the harvested biomass for energy and the consumption of the  $CO_2$  produced in the process of energy conversion and utilisation by new biomass plant growth (Ciubota-Rosie et al., 2008; Jefferson, 2006; Robu, 2005). According to Miner (2010), carbon neutral activity is the one that balances the release of  $CO_2$  into the atmosphere and the absorption from the atmosphere. The carbon neutral cycle is illustrated in Figure 1.1 (Miner, 2010).



Fig.1.1. Carbon neutral cycle of biomass conversion through AD

#### 1.2.2 Municipal solid waste generation and the world population

The United Nations (UN) in 2007 projected that the world population will increase by 2.5 billion between 2007 and 2050, that is, from 6.7 billion in 2007 to 9.2 billion in 2050 (UN, 2007). This increase is equivalent to the population of the world in 1950 and will be absorbed by the less developed parts of the world, whose population is likely to rise from 5.4 billion in 2007 to 7.9 billion in 2050, whilst the developed nations are expected to remain marginally constant at 1.2 billion people (UN, 2007). Similarly, UNFPA (2007) forecasted that about 5 billion people would live in the municipal areas by 2030.

As a result of the continuous growth of the world population as well as improvement in the standard of living of people and industrial advancement, the amount of municipal solid waste (MSW) generated is on a constant rise. The Organisation for Economic Corporation and Development (OECD, 2004), stated that between the year 1980 and 2000 the amount of municipal waste generated in its member states increased by about 54%. In another report, OECD stated that its member states are generating over 4 billion tonnes of MSW annually (OECD, 2013). In addition, it reported that in 2012 each person within the 28 EU member states generated an average of 492 kg of MSW (Eurostat, 2014). In the same report, each person in the UK generated an average of 472 kg of MSW (Eurostat, 2014).

Similarly, DEFRA reported that the UK generated 31.1 million tonnes of MSW in 2012 (Themelis and Bourtsalas, 2013). It is established that MSW is a global problem, but the improper waste management in developing countries increases the susceptibility to environmental and health hazards. These potential risks include air pollution, emission of greenhouse gases, underground water contamination, cholera, typhoid fever and bubonic plague (Antonis, 2013; Boadi and Kuitunen, 2005; DEFRA, 2004)

#### **1.2.3** Evolution of waste management concepts

Before the advent of industrial revolution, the nature of waste produced was mainly biodegradable, such as vegetable, human waste and ashes from the incineration of other waste materials (Chandler et al., 1997; Ascari et al., 1992). The management of waste was not complex due to the utilisation as fertiliser or soil conditioner on farmland (Sangodoyin and Ipadeola, 2000; Chandler et al., 1997; Ascari et al., 1992; Nightingale, 1954). The industrial evolution in the 19<sup>th</sup> century created a significant increase in the global economic activities, resulting in mass migration of people to the industrial cities from the rural areas (Ayomoh et

al., 2008). The increase in industrial activities and the constant growth in human population, especially in the cities led to a significant increase in the quantity of solid waste generation (Ziadat and Mott, 2005). As a result, waste management becomes a critical issue to humanity (Articlebase, 2011; Melosi, 2005; Chandler et al., 1997). The different techniques used to minimise the challenges of waste management include landfill, incineration, recycling and biological reprocessing.

Landfill and incineration are the oldest and most common form of waste management techniques (Themelis and Ulloa, 2007). Many countries, particularly the developing countries, still apply landfill as the main waste management system (EEA, 2009; Themelis and Ulloa, 2007; Massarutto, 2001; Buclet and Godard, 2000). Landfills had been linked to a number of environmental and health hazards, like air contamination (Davoli et al., 2010), strong odours and smells perceived over 1 km from landfill sites (HPA, 2011), congenital malformation and low birth weight among babies born to women living within 2 km from landfill sites (Forastiere et al., 2011).

Due to the risks associated with landfill, many countries instituted frameworks on landfill regulation. One of the most prominent frameworks legislated by the EU is the Waste Frameworks Directive (Directive 75/442/EEC, replaced by the modified Directive 2008/98/EC). The framework emphasises the reduction of waste that goes to landfills (EEA, 2009). Similarly, the United States Environmental Protection Agency (USEPA) on October 9, 1991, publicised a revised minimum technical requirement for MSW landfills (USEPA, 1991). This regulation led to the closure of many smaller landfills in the USA, reducing the total number from 6,500 in 1988 to 1,767 in 2003 (USEPA, 2003). In addition, the USEPA (1991)

framework include guidelines for remodelling old landfills, construction of new landfill systems and the type of waste that go into them (Daskalopoulos et al., 1997).

The remodelled and new landfill systems are equipped with impermeable liners and caps as well as a subsystem that collects and treats leachate. They also have gas wells and pipes that collect and transport landfills gas, mainly methane, to where it is utilised for electricity or heat generation (Themelis and Ulloa, 2007). Furthermore, the landfill systems consist of impervious layer covers applied when full to prevent rainwater as well as facilitate biodegradation process (Themelis and Ulloa, 2007).

The application of incineration to reduce the excess quantity of household and agricultural waste dates back to thousands of years ago (Petts, 1994). Incineration involves the combustion of solid waste to produce ash residue (Petts, 1994). It is an effective way of reducing the volume of MSW and the demand for landfill space. However, it is related to possible environmental and health risks. These include airborne emission of dioxins (hazardous chemicals) (HPS, 2009), Sulphur dioxide (SO<sub>2</sub>) (Roberts and Chen, 2006, also cited by Fewtrell, 2012) and nitrogen dioxide (NO<sub>2</sub>) (Forastiere et al., 2011), which can cause cancer and death.

Similarly, a number of countries enacted frameworks for incineration systems due to the potential dangers. The Waste Incineration Directive (WID) 2000/76/EC of European Parliament and of the Council (WID, 2000), and the Waste Incineration (England and Wales) Regulations (WIR) 2002 (WIR, 2002) are some of the frameworks instituted. The aim of the directives is to innovate incineration systems in order to minimise the environmental and health risks (DEFRA, 2009). The recycling method of waste management is not discussed further in this thesis, but the biological reprocessing is discussed as follows.

#### **1.2.4** Motivation of anaerobic digestion for organic waste management

Appropriate waste management practice is crucial for any sustainable society. It prevents air, soil and water pollution as well as improves public health, decreases greenhouse gas emission and preserves natural resources. The AD system, which is a type of biological reprocessing, is the ideal waste management technique (Bohn, 2010; Hartmann et al., 2004). AD is the process by which bacteria breakdown organic material to produce biogas (renewable energy source) and digestate (biofertiliser) in the absence of oxygen. It involves not only the collection and safe disposal of organic waste but also sustainable management of organic material to create a source of renewable energy as well as provide economic, health and environmental benefits (Massarutto, 2010). According to Bohn (2010), AD reduces greenhouse gas emission more than any other waste management systems. It is an effective method of recovering energy and nutrients from organic material (Hartmann et al., 2004). The following are some of the benefits of AD system (DEFRI, 2009).

- Contribution towards mitigation of climate change and other environmental targets;
- Treatment of biodegradable wastes to generate biogas, a renewable energy that can be utilised to produce electricity and heat from combined heat and power (CHP) or for vehicle fuel;
- Diversion of organic wastes from landfills and capturing of methane emission from organic wastes;
- Provision of organic fertiliser and soil conditioner for agriculture and land use; and
- A source of revenue generation to farmers and other practitioners, as excess energy and digestate are sold, which adds to the nation's gross domestic product (GDP).

#### **1.3 Problem statement**

The enhancement of biogas production is crucial in recent years due to increasing demand for biogas, resulting from the global quest for alternative sources of energy that are sustainable (Krustok et al., 2013; van Foreest, 2012; Rajendran et al., 2012; Ploechl et al., 2010). However, the production of biogas to keep up with the global demand is limited by the underperformance in the system implementing the AD process. This underperformance is due to the difficulty in obtaining and maintaining the optimal operating parameters/states for anaerobic bacteria to thrive with regards to attaining a specific critical population number, which results in maximising the rate of biogas production. This problem continues to exist as a result of insufficient knowledge of the interactions between the operating parameters and bacterial community. In addition, the lack of sufficient knowledge of the composition of bacterial groups that varies with changes in the operating parameters such as temperature, substrate and retention time. Without sufficient knowledge of the overall impact of the physico-environmental operating parameters on anaerobic bacterial growth and composition, significant improvement of biogas production may be difficult to attain.

In order to mitigate this problem, the construction of a suitable dynamic mathematical model is required that enables a better understanding of the relationship between the operating parameters and biogas production. This facilitates the means of predicting the behaviour of bacteria in AD, under different physico-environmental operating parameters

A number of AD models have been developed by previous studies (Batstone et al., 2002; Amon et al., 2007; Keymer and Schlicher, 2003; Angelidaki et al., 1999; Boyle 1976; Buswell and Mueller, 1952). However, the current models are mainly dependent on assumptions that

consequently result in neglect of real-life scenarios, which limits a more realistic study on AD process.

#### 1.4 Aim and objectives

An effective digester operation is one that maintains the optimal operating parameters/states for anaerobic bacteria to thrive with regards to attaining a specific critical population number which results in maximising the rate of biogas production.

The aim of this PhD study is to develop a black-box (empirical) model for mesophilic AD process, in order to understand better, how physico-environmental operating parameters influence biogas and methane production. More specific there is a need to consider the control limits for the controlled parameters necessary for improving methanogenesis reaction while ensuring an enhanced digestion operation.

In order to accomplish this goal, the following objectives are addressed.

- Identify and understand the performance variables of mesophilic AD system (Chapter 2);
- Conduct a critical review of the operating parameters that influence the production of biogas and methane under mesophilic condition. For this research work the following four operating parameters are considered; temperature, pH, mixing speed and pressure (Chapter 2);
- 3. Determine the limits for the four operating parameters and their impact on biogas and methane production. This includes a series of laboratory experiments with each of the operating parameters. The data obtained from the investigations are analysed and the

range of values for each operating parameter that recorded the highest performance in terms of biogas quantity production and quality are determined (Chapters 4 & 5);

- 4. Construct a black-box model each for biogas production and methane production. The limits of the four operating parameters considered are analysed further in order to establish their correlation with biogas and methane production. Subsequently, two nonlinear multi-parameter black-box models are constructed using System Identification method through MATLAB functions. The models are able to predict the behaviour of the real system with sufficient accuracy (Chapter 6); and
- 5. Determine the scalability of black-box models. This is to determine the possibility of scale-up of the black-box models constructed in this research study.

#### **1.5 Summary**

This chapter has presented the concept of this research work. It highlighted the impact of energy production and consumption on the environment and all live forms, as well as the global need for an alternative energy source that is sustainable. It also discussed the risks of increasing MSW on the environment and public health. In addition, the chapter introduced AD as the ideal technology for both a sustainable energy source and organic waste management. It then presented the problem statement, as well as the aim and objectives of this research work. The following chapter discusses the review of previous and relevant scientific articles in AD technology.
# Chapter 2 Development in Anaerobic Digestion: Literature Review

## 2.1 Introduction

The application of AD was recorded in China and India over 2000 years ago when it was used for treating animal manure (Nayono, 2010; Veenstra, 2000). However, not until the 19<sup>th</sup> century was AD applied for wastewater and solid waste treatment (Nayono, 2010; Residua, 2009). The reason for the gap was due to the perception of AD as a slow and ineffective technology for organic waste treatment, especially for the increasing wastewater volume in industrialised and densely populated municipalities (Nayono, 2010; Gijzen, 2002; Polprasert, 2001). The first properly constructed AD plant was in a leper colony in Bombay, India in 1859 (Abbasi et al., 2012; Khanal, 2008; Meynell, 1976). Similarly, the first account of AD system in England, UK, was in 1895, during which biogas was generated from sewage treatment plant and utilised as fuel for street lamps in Exeter (Residua, 2009; McCabe and Eckenfelder, 1957). AD technology has continued to gain prominence in many countries, particularly, in the recovery of renewable energy from household and municipal waste (Holm-Nielsen, 2009). In addition, AD contributes to sanitation improvement, greenhouse gas mitigation, and production of organic fertiliser (Wang, 2014; Johansen et al., 2013, Vaneeckhaute et al., 2013; Wang et al., 2011; Zheng et al., 2010). This chapter presents a discussion of the various physicoenvironmental operating parameters that influence the performance of AD process, as it relates to anaerobic bacteria activity and biogas production.

#### 2.2 Operating parameters that influence AD performance

There are a number of operating parameters that influence anaerobic bacterial activities such as specific growth rate, degradation rate, substrate utilisation and biogas production (Rajendren et al., 2012; Deublein and Steinhauser, 2008; Al Seadi et al., 2008). For improved performance of AD process, it is necessary to maintain the appropriate range of the operating parameters that support optimal growth of anaerobic bacteria (Rajendren et al., 2012; Deublein and Steinhauser, 2008; Al Seadi et al., 2008). In order to maintain the stability of anaerobic reaction, it is crucial to maintaining a balance between the rate of organic acid formation and methane production (Bernard, 2012; USEPA, 2008). Maintaining a proper equilibrium between acidogenic and methanogenic bacteria requires that the operating parameters are kept within their different optimal ranges (Bernard, 2012; USEPA, 2008). The operating parameters discussed in this thesis include pH, temperature, substrate characteristics, hydraulic retention time (HRT), organic loading rate (OLR), mixing and carbon to nitrogen (C/N) ratio (Rajendren et al., 2012; Deublein and Steinhauser, 2008; Yadvika et al., 2004; Gerardi, 2003).

# 2.2.1 pH

The performance of AD process is affected by variation of pH in the digester (Cioabla et al., 2012; Khalid et al., 2011; Romano and Zhang, 2011; Kumar, 2010; Wu et al., 2009). Each bacterial group in AD process has a specific range of pH values at which they optimally multiply and are most active (Khalid et al., 2011). Various studies have shown that anaerobic bacteria can function in a pH range of 5.5 - 8.5 (Nayono, 2009; Al Seadi et al., 2008; Lay et al., 1998; RISE-AT, 1998; Stronach et al., 1986). On the other hand, Cioabla et al., (2012) limited the pH range at which anaerobic bacteria can withstand to 6.5 – 8.0. However, it is reported that the optimum pH range for AD is 6.8 - 7.2 (Cioabla et al., 2012; Ward et al., 2008), which is the pH range for optimal growth of methanogenic bacteria. Other studies reported different optimal pH ranges for methanogenic bacterial growth, such as 6.8 - 7.6 (Nayono, 2009; Lay et al., 1998; Stronach et al., 1986), 7.0 - 8.0 (Al Seadi et al.'2008) and 6.5 - 8.2 (Lee et al., 2009b). The ideal optimal pH value for methanogenesis is 7.0, the neutral pH (Wang,

2014; Khalid et al., 2011; Yang and Okos, 1987; Huber et al., 1982), which should be the target pH value for biogas production. Furthermore, Hydrolytic and acidogenic bacteria are found to multiply optimally at lower pH levels of 5.5 and 6.5, respectively (Khalid et al., 2011; Kim et al., 2003).

Methanogenic bacteria are sensitive to the acidic environment; their growth, as well as methane production, are inhibited in acidic condition (Wang, 2014; Kangle et al. 2012). Organic acids produced during acidogenesis can decrease the pH level of the reaction below 5.0 and consequentially, limit the growth of methanogenic bacteria (Kangle et al. 2012). Since methanogenic bacteria utilise the organic acids, decrease in their population would lead to excess volatile fatty acids (VFAs) accumulation, resulting in digester sour – a situation whereby methanogenic bacteria fail to keep pace with acidogenic bacteria and the digester becomes acidic (Kozani, 2014). The acidic condition can be caused by any of or a combination of the following factors, temperature change, toxic compound or sudden addition of a large quantity of digestible organic matter. However, when there is less population of acidogenic bacteria than methanogenic bacteria in AD process, the pH value can increase beyond 8.0, which can cause an increase in the concentration of ammonia, and result in digester failure (Lusk, 1999). Since methanogenesis is the rate-limiting stage of AD, it is important to keep the pH value of digester within methanogenesis optimal range of 6.8 - 7.2.

#### 2.2.2 Temperature

Temperature is characterised as the most important parameter that influence AD process (Cioabla et al., 2012; Chae et al., 2008; Choorit and Wisarnwan, 2007). The reason for this is due to the impact of temperature on all the processes involved in AD, which includes physico-chemical properties of substrates in the digester as well as the kinetics and thermodynamics of

biological processes (Çalli, 2012; Kangle et al. 2012; Boe, 2006). The different groups of anaerobic methanogenic (methane forming) bacteria can multiply under various temperature limits (USFDA, 2012; Lamprecht, 2009; Bisschops et al., 2009; Lettinga et al., 2001). Figure 2.1 shows the rate of growth of different methanogens at varied temperature ranges (Lettinga et al., 2001).



Fig. 2.1. Relative growth rate of different methanogenic bacteria

As seen in Figure. 2.1, different methanogenic bacteria grow in varied temperature ranges. Psychrophilic methanogens can multiply under temperature between 0 - 20 °C, with optimum temperature range of 10 - 15 °C (Schmidt and Schaecheter, 2012; Willey et al., 2008; Lettinga et al., 2001; Hoyle, 2003; Temper et al, 1983; Winter et al, 1982; Morita, 1975). Mesophilic bacteria can populate at temperature between 10 - 45 °C, but grow optimally at temperature range of 35 - 40 °C (Schmidt and Schaecheter, 2012; Willey et al., 2008; Lettinga et al., 2001; Hoyle, 2003; Temper et al, 1983; Winter et al, 1982; Morita, 1975). Whilst thermophilic bacteria can withstand a higher temperature range of 25-65 °C, and the optimum temperature

range is between 50–60 °C (Schmidt and Schaecheter, 2012; Willey et al., 2008; Hoyle, 2003; Lettinga et al., 2001; Morita, 1975). In addition, AD process can operate at temperatures above 60 °C, which is referred to as extreme-thermophilic (Frock and Kelly, 2012; Abreu et al., 2010). Mackie and Bryant (1981) reported that low rates between these optimal temperatures could be attributed to lack of adaptability of the different methanogenic bacteria.

AD is predominantly operated at a mesophilic temperature range (Li et al., 2014; Wang et al., 2010). This is because there are more variety of bacteria that multiply at a mesophilic temperature range, thus mesophilic digestion is considered more stable than thermophilic and psychrophilic systems (Li et al., 2014; Wang et al., 2010). Similarly, mesophilic bacteria adapt to changing environmental conditions more than thermophilic bacteria (Li et al., 2014; Wang et al., 2010; Cheremisinoff 2008). In addition, thermophilic AD systems require more energy to maintain the required temperature, thereby making it more expensive to operate (OECD, 2010; Cheremisinoff 2008). However, thermophilic anaerobic process has faster degradation rate than mesophilic digestion, resulting in shorter HRT, higher rate of methane production and kill pathogens better than mesophilic AD process (Li et al., 2014; Zhao, 2011; Luostarinen, 2005; Mata-Alvarez, 2002; Lettinga et al., 2001; NREL, 1992).

## 2.2.3 Carbon to Nitrogen (C/N) ratio

The relationship between the proportion of carbon and nitrogen in a substrate is referred to as the carbon to nitrogen (C/N) ratio (Kangle et al., 2012). Different substrates have dissimilar C/N ratios. It is essential to maintain a suitable C/N ratio that can facilitate optimum bacterial growth in the digestion process (Kangle et al., 2012). Higher concentration of nitrogen in AD leads to excess ammonia production and can increase the pH level beyond 8.5, which is toxic to methanogenic bacteria (Kangle et al., 2012; Mata-Avarez, 2000). However, low nitrogen

concentration can result in acidic digester content, leading to insufficient ammonia production, which in effect, inhibits methane production (Kangle et al., 2012; Hartmann and Ahring, 2006; FOA, 1992). The nitrogen content in the substrate is converted to ammonia, which neutralises the VFAs produced by acidogenic bacteria, thereby maintaining the pH level at near neutral that is favourable for methanogenic bacterial growth and improved methane production (FOA, 1992; Speece and McCarty, 1964). Kimchie (1984) suggested that C/N ratios lower than 10:1 will be inhibitory to digestion process, but higher than 23:1 will not support optimum digestion process. In order to keep the optimal balance between carbon and nitrogen, substrates with high or low C/N ratios can be co-digested, such as animal manure and organic solid wastes (Kumar, 2012; Zaher et al., 2007).

# 2.2.4 Substrate characteristics

The performance of AD can be affected by the characteristics of the substrate such as composition, degradability, homogeneity, fluid dynamics, particle size and C/N ratio (Nayono, 2009; Steffen et al., 1998). Solid waste, especially the organic content of MSW may differ due to a number of reasons, including weather condition, cultural habits, seasons of the year and manner in which they are collected (Nayono, 2009).

The proportion of substrate composition (carbohydrates, proteins and fats) influences its biodegradability as well as methane yield potential (Hartmann and Ahring, 2006). Substrates that contain a greater amount of fats yield more methane than the same quantity of carbohydrates and proteins (Angelidaki et al., 1990; Hanaki et al., 1981). This is because more energy is stored in the bonds of fat molecules (Fernandes et al., 2009). When electrons migrate from the atom, like carbon, with a low affinity for electrons to atom, such as oxygen, with a high affinity for electrons, energy is given off (Fernandes et al., 2009). So, more energy is

released when fatty acids are oxidised than when the same process takes place in carbohydrate and proteins (Fernandes et al., 2009; Mosier et al., 2005). However, fat-rich substrates degrade slower than that of carbohydrates and proteins. This is due to the slower hydrolytic phase of fat catabolism compared to that of carbohydrates and proteins, which are more readily hydrolysed (Neves et al., 2008; Vidal et al., 2000).

The fluid dynamics of the substrate, which changes with seasons of the year, can influence the performance of anaerobic digester (Motte et al., 2013). Substrate with high water content may increase digester volume and require higher heat input per m<sup>3</sup> of digester content, thereby reducing digester effectiveness (Motte et al., 2013; Steffen et al., 1998). However, a substrate with low water content, meaning that the total solids (TS) of the substrate is high, can cause ineffective mixing performance, scum layer formation, solid sedimentation and clogging (Motte et al., 2013; Steffen et al., 1998). Steffen et al. (1998) suggested that for conventional Continuous Stirred Tank Reactor (CSTR), a type of digester, TS concentration of substrate should be about 6–10 %. TS is the ratio measured in percentage of the weight of solids left after the substrate sample was dried in an autoclave at 105 °C for approximately 24 hours and the weight of the original substrate sample (Sluiter et al., 2008). TS is represented mathematically in Eq. 2.1 (Sluiter et al., 2008):

$$\% \mathrm{TS} = \frac{((DPW + DSW) - DPW)}{oSW} \ge 100$$
(2.1)

Where,

DPW, DSW, OSW represent dry pan weight, dry sample weight and original sample weight, respectively.

Furthermore, the particle size of the substrate can significantly affect AD performance (Hajji and Rhachi, 2013; Hartmann and Ahring, 2006). Particle size reduction increases the available

specific surface area that facilitates biodegradation, especially in hydrolysis stage, resulting in an increase in methane production as well as a reduction in the quantity of residue generation (Hajji and Rhachi, 2013; Hartmann and Ahring, 2006; Palmowski and Müller, 2000; Hills and Nakano, 1984). Mshandete et al. (2006) found that by reducing the size of sisal fibre waste to 2mm, the methane production potential and total fibre biodegradation improved by approximately 20 % and 31–70 %, respectively. This has shown that pre-treating the particle size of the substrate can enhance both the performance of anaerobic digester as well as improve methane production.

# 2.2.5 Hydraulic retention time (HRT)

The hydraulic retention time (HRT) is the time it takes for complete digestion of substrate (Hartmann and Ahring, 2006; Lu and Ahring, 2005). The HRT of AD depends on some factors, which include process temperature, substrate composition, particle size, and digester configuration (Hartmann and Ahring, 2006; Lu and Ahring, 2005). The HRT for substrates digested in mesophilic AD process varies from 10 - 40 days, while that treated in thermophilic digester can take up to 3 - 14 days (Hartmann and Ahring, 2006; Lu and Ahring, 2006; Lu and Ahring, 2005). The HRT can be determined analytically from the relationship between digester volume and the volume of substrate fed per unit time as shown in Eq. 2.2 (Al Seadi et al., 2008).

$$HRT = \frac{V_R}{V}$$
(2.2)

Where,

HRT = hydraulic retention time (days)

 $V_R$  = digester volume ( $m^3$ )

V = volume of substrate fed per unit time  $(m^3/d)$ 

Due to the relatively slow growth of methanogenic bacteria, sufficient HRT is required for repopulation in order to make up for the bacteria discharged with the effluent (liquid discharged from the digester) (Zaher et al., 2007). Thus, maintaining bacterial population to cope with any inconsistency in the organic loading rate (OLR) (Zaher et al., 2007).

# 2.2.6 Organic loading rate (OLR)

OLR is the quantity of organic material (substrate) that is to be digested by a particular volume of the digester in a particular period of time (Al Seadi et al., 2008). Loading anaerobic digester above its sustainable capacity can cause accumulation of fatty acids, which inhibits AD process, resulting in process failure and low or no methane yield (Al Seadi et al., 2008; Vandevivere, 1999; Rise-at, 1998). OLR can be used as a measure of how much dry matter that can be loaded into a digester per volume of a unit time as presented in Eq. 2.3 (Al Seadi et al., 2008).

$$B_R = m \cdot \frac{c}{v_R} \tag{2.17}$$

Where,

 $B_R$  = organic load (kg/d ·  $m^3$ ) m = mass of substrate fed per time unit (kg/d) c = concentration of organic matter (%)  $V_R$  = digester volume ( $m^3$ )

However, improper OLR can affect biogas production (Al Seadi et al., 2008). As mentioned above, if the concentration of organic material loaded to the digester is high, it can lead to the high production of organic acids during acidogenesis, which decreases pH level below the acceptable level and in effect, decreases the population of methanogenic bacteria. Since methanogenic bacteria utilise the organic acids, a decrease in their population will result in VFAs accumulation and consequently digester sour (Lusk, 1999). However, the lack of sufficient organic material in the digester leads to a shortage of VFAs, as a result, reduces the population of methanogenic bacteria. The effect of the reduced number of methanogenic bacteria results in the rise of the digester pH beyond the optimum level, thereby, increases the ammonia content that leads to digester failure (Lusk, 1999).

# 2.2.7 Mixing

The anaerobic digesters require mixing/stirring to improve contact between bacteria and substrate, facilitating bacterial access to nutrients in substrate (Meroney and Colarado, 2009; Wards et al., 2008). Mixing prevents scum formation and thermal stratification in the digester (Meroney and Colarado, 2009; Karim et al., 2005). By mixing digester contents, a denser particulate such as heavy solids particles and sand are kept suspended to avoid sedimentation (Wards et al., 2008; Kaparaju et al., 2007).

There are different types of mixing systems, such as gas mixing, pumped mixing, long shafted paddle mixing, wall mounted draft tube mixing and submersible mixers (Wards et al., 2008; Karim et al., 2005; Burton and Turner, 2003). The AD process does not always require continuous mixing, in practice, intermittent mixing is often applied; which can vary from a few times a day to several times an hour. Excessive mixing upsets bacterial population (Lindmark, 2014). However, moderate mixing is preferable for improved digestion (Lindmark, 2014), which is supported by the findings of this study and reported in Chapter 5 of this thesis.

## 2.3 Types of anaerobic digester

The anaerobic digester is the main component of AD. It is an airtight tank that can be of any shape (cylindrical, rectangular, square and egg-shaped), where biodegradation of substrate and

biogas production take place (Ward et al., 2008). The design of anaerobic digester is required to contain the following basic objectives; to make a sustainable and continuous high organic load rate possible, to achieve a short HRT as possible and to optimise the production of methane (Khalid et al., 2011; Ward et al., 2008). The anaerobic digester can be broadly categorised according to the feeding mode (batch and continuous) and by the solid content of the substrate (dry and wet digestion) (Ward et al., 2008).

## **2.3.1** Feeding mode (batch and continuous)

The batch digester is considered the simplest to design, construct and operate (Khalid et al., 2011). The digester is fed with substrate mixed with water to form a slurry, then sealed and allowed for a length of time. During the HRT, biogas production progresses to a maximum and then decreases slowly as bacterial effectiveness decreases. Not only that the batch digester is simple to construct and operate, it is also not expensive. It can perform rapid digestion process and be utilised easily to measure the rate of digestion (Khalid et al., 2011; Koppar and Pullammanappallil, 2008; Weiland, 2006; Parawira et al., 2004; Vandeviere et al., 2002; Ouedraogo. 1999). However, batch digester has some limitations, including volume restriction, the inconsistent rate of biogas production and varying the proportion of methane content in biogas (Linke et al., 2006).

Unlike batch system, continuous fed digester requires daily loading of an organic substrate (Ward et al., 2008). The volume is designed large enough to contain the daily substrate feeding throughout the HRT (Ward et al., 2008). Continuous feed system can be divided into two types, namely, single-stage and two-stage or multistage digesters (Vandevivere et al., 2002; Lissens et al., 2001). In one-stage continuous feed system, all the four steps of AD process take place simultaneously in one digester. One major disadvantage of this system is that the entire

biochemical processes are kept under the same operating parameters, despite the fact that the growth rate of the various bacterial groups involved and their optimal pH ranges are different (Geradi, 2003; Vandevivere et al., 2002).



Fig. 2.2. Single-stage digester

However, a two-stage or multistage continuous feed system have the four stages of AD process take place in two separate phases in two digesters; hydrolysis/acidification take place in one digester and acetogenesis/methanogenesis processes occur in another digester. The product of the first stage (hydrolysis/acidification) in the first digester is fed into the second digester for the acetogenesis/methanogenesis step (De Baere, 2000; Schober et al., 1999). The separation of hydrolysis and acidification processes from acetogenesis and methanogenesis processes is mainly because they do not perform optimally under the same optimal pH range (Liu et al., 2006a; Liu et al., 2006b; Zoetemeyer et al., 1982).



Fig. 2.3. Two or multi-stage digester

The two or multi-stage system maintains better process stability than the one-stage system, particularly during the metabolism of substrates that can be easily hydrolysed (Ward et al. 2008; Bouallagui et al., 2005; Mata-Alvarez, 2002). Although multi-stage system costs more to construct and maintain, it is more effective compared to the single-stage method. A study reported by Nielsen et al, (2004), where a one-stage and a two-stage thermophilic system fed with cattle dung were compared. The result of the study shows that the two-stage system produced 6 - 8% higher specific methane as well as effectively removed 9% more volatile solids than the one-stage process. In another study, MSW was utilised as the substrate, a two-stage system was found to produce 21% methane more than a single-stage system (Liu et al. 2006a). The multi-stage system is capable of handling variation in OLR and quality as well as maintains optimal conditions for the entire process than single-stage system (Rapport et al., 2008). Despite the various advantages of the multi-stage system over the single-stage system,

yet the multi-stage system has not been widely applied for commercial purposes (Rapport et al., 2008; Vandevivere et al., 2002; de Baere, 2000). This is attributed to the added design complexity and extra costs in constructing and operating a multi-stage anaerobic digester (Rapport et al., 2008; Vandevivere et al., 2002; de Baere, 2000).

## 2.3.2 Solid content (wet and dry)

The AD system can also be categorised as wet or dry digestion system, depending on the concentration of the total solids of substrate (Ward et al. 2008). Wet AD system is intended to process a dilute substrate, like slurry, which has its total solids concentration less than 15%. Whilst dry AD system is designed to digest substrate that contains 15 - 40% total solids (Lissens et al., 2001).

In wet digestion system, if the incoming substrate contains total solids higher than 15%, it can be diluted to make the appropriate slurry by adding deionised water, recirculated process water, or co-digested with another substrate with a lower total solids concentration (Hartmann and Ahring, 2006). For instance, Organic Fraction of Municipal Solid Waste (OFMSW) can be blended with sewage sludge to form the suitable slurry for wet digestion (Hartmann and Ahring, 2006; Luning et al., 2003). Wet digestion system is useful in processing low solid organic materials such as food industry liquid waste and sewage sludge (Hartmann and Ahring, 2006). However, the disadvantages of wet digestion system include the requirement for continuous mixing of slurry in the digester, which results in extra energy consumption by the mixer; the digester volume may be reduced as heavy and inert materials accumulate at the digester base due to sedimentation (Banks and Stentiford, 2007; Vandevivere et al., 2002). In addition, it requires larger digester volume compared to dry digestion system, more robust

water pumping and piping and utilises more energy to heat the extra digester volume (Banks and Stentiford, 2007; Vandevivere et al., 2002).

The dry digestion system has the advantage of minimising the process risk posed by the suspension of fibrous organic materials, like straw or straw-containing animal dung in the digester, which is common with wet AD process (Kalia and Singh, 1998). Furthermore, dry digestion system requires less energy for mixing and heating of digester content, less digester volume as well as reduced water consumption (Kottner, 2002; Vandeveivere et al., 2002; Hoffmann, 2001). However, due to the high solid concentration of such substrates, dry AD system requires a higher technical equipment for substrate handling, mixing devices, pumping and pre-treatment compared to the wet system (Sigrid et al., 2011; FNR, 2009; Lissens et al., 2001).

### 2.4 Summary

This chapter has discussed a number of operating parameters that influence anaerobic bacterial activity and subsequently, biogas and methane production. It is found that in order to improve biogas production; the system is required to maintain the appropriate limits of the various operating parameters that support the optimal growth rate of anaerobic bacteria. In addition, the different categories of anaerobic digesters, which are characterised by either the feeding mode or the solid content of the substrate. The review of the relevant scientific articles is continued in the next chapter with a focus on the improvement and the current state in AD technology.

# Chapter 3 Improvement and current state in AD: Literature Review

## **3.1 Introduction**

Many studies have been going on for the past decades in the field of AD with the aim of improving biogas productivity (Syngellakis, 2014; Montgomery and Bochmann, 2014; Brebbia, 2010; Ahring, 2003; Van Lier et al., 2001). The review of the relevant scientific articles in AD technology discussed in Chapter 2 above covered the physico-environmental operating parameters and their influence on AD performance, as well as the different types of anaerobic digesters based on the feeding mode and substrate solid content. However, this chapter discusses the relevant improvements and current state in AD technology, which include pre-treatment processes, co-digestion of different organic materials and process modelling.

# 3.2 Pre-treatment for process enhancement

Pre-treatment is the means of improving biogas yield per volume or mass unit of input substrate, by treating a substrate in advance before loading it into a digester (Malatak, 2013). In AD, especially when digesting complex substrate, it is not the entire substrate that is converted to biogas. This is due to the inability of the bacteria to gain access to the whole substrate (Malatak, 2013). Simple substrates such as cellulose and hemicellulose are easily biodegradable in AD. However, when combined with lignin, like lingo-cellulosic, they become complex (long chain polymer) and reduce in biodegradability (Van Liere, 2011). In digesting complex substrates, hydrolysis becomes another limiting stage (Zhang and Cai, 2008; Zhang et al., 2007; Noike et al., 1985). Meaning that apart from methanogenesis, biogas production also depends on the rate of hydrolysis - biodegradability (Fernandes, 2009).

A number of pre-treatment methods have been developed for AD process, which can be applied separately or combined (Donoso-Bravo and Fdz-Polanco, 2013; Quinones et al., 2012; Luo et al., 2012; Bishnoi, 2012; Erden and Filibeli, 2009; Perez-Elvira et al., 2009). The methods are broadly characterised into mechanical, chemical, biological, thermal and ultrasonic.

Mechanical method is a means of pre-treatment that is utilised mainly to reduce substrate particle size, by grinding and pressing in order to break the cell walls (Quinones et al., 2012). The advantages of reducing substrate particle size include an increase in the available specific surface area that facilitates biodegradation, increase in methane production and a decrease in the quantity of residue generation.

Chemical is the second method of pre-treatment process, which is applied by acidifying or alkalising the substrate (Quinones et al., 2012). Mata-Alvarez et al. (2000) reported the pre-treatment of fibres with N<sub>a</sub>OH, NH<sub>4</sub>OH and a combination of both chemicals resulted in improving biogas and methane production. In addition, Lopez-Torres and Espinosa-Llorens (2008), reported that increase in methane potential occurred when fibre was mixed with CaO.

In biological pre-treatment, the substrate is broken down aerobically in advance before feeding it to the digester. This process has shown to reduce solid particles, improve degradation rate and methane production (Mata-Alvarez et al., 2000; Capela et al., 1999). In another development, enzymes were used to catabolise substrate prior to feeding it to the digester, which is found to enhance degradation, while improving biogas production (Donoso-Bravo and Fdz-Polanco, 2013; Luo et al., 2012; Quinones et al., 2012; Petersson et al., 2007).

The thermal process is another kind of pre-treatment method used in AD. It utilises high temperature  $(130 - 180 \,^{\circ}\text{C})$  in conjunction with pressure (5 - 8 bar) to break the cell structure of organic substrate (Buddle et al., 2008). De-coupling and hydrolysing the protein and starch, make the complex substrate more readily available for bacterial consumption (Bishnoi, 2012; McCartyet al., 1986; Haug et al., 1983). According to Jolis et al. (2008), thermal pre-treatment achieved a 55 to 60 % volatile solids destruction.

Furthermore, ultrasonic pre-treatment is applied in the catabolism of complex substrate like sewage sludge (Carrere et al., 2010). In this process, cell structure of sludge is mechanically disrupted due to induced cavitation (Carrere et al., 2010; Tiehm et al., 1997). It is found that cell disintegration changes the characteristics of sludge, resulting in an increase in biodegradability and improving volatile solid removal (Tiehm et al., 2001; Tiehm et al., 1997). In addition, it enhances biogas production (Erden and Filibeli, 2009; Perez-Elvira et al., 2009; Salsabil et al., 2009; Ward et al., 2008; Braguglia et al., 2008; Neis and Nickel, 2008; Xie et al., 2007; Bougrier et al., 2005; Bien et al., 2004; Chu et al., 2002; Onyeche et al., 2002; Wang, et al., 1999).

### 3.3 Co-digestion of different substrates

Co-digestion is a process in AD by which two or more substrates are homogenously blended and simultaneously digested (Agdag and Sponza, 2007). The AD was initially applied to a single substrate, single purpose treatment plant (Wu, 2007). However, studies have shown that AD process is more stable and achieves more potential benefits as a result of co-digestion (Wu, 2007; Hartmann and Ahring, 2005; Murto et al., 2004; Mata-Alvarez et al., 2000). Co-digestion enhances biogas production due to the supply of the required nutrients by the co-substrates involved (Mata-Alvarez et al., 2000). This achievement in nutrient balance leads to better digester performance and improved biogas production (Wu, 2007; Hartmann and Ahring, 2005). Murto et al. (2004) reported that co-digestion of solid slaughterhouse waste, fruit, vegetable and manure improved the buffering capacity of digester – the ability of digester to react to changes in pH. In other words, the buffering capacity of the digester is the measure of the amount of alkalinity present in the digester.

In addition, co-digestion is found to maintain the appropriate balance of C/N ratio in AD (Desai et al., 1994) and to reduce the concentration of nitrogen (Cuetos et al., 2008). For instance, the mixture of substrate with low nitrogen content and lipid is found to enhance biogas production (Castillo et al., 2006) as well as reduce the risk of acid accumulation and high concentration of ammonia that cause digester failure (Khalid et al., 2011; Castillo et al., 2006). In another study, co-digestion of the substrate that contains high moisture (liquid manure or sewage sludge) and substrate of poor moisture content is found to enhance the total solids (TS) content of digester (AgSTAR, 2012). Similarly, co-digestion involving substrate with high moisture content is found to dilute toxic compounds, thereby improving digester performance and biogas production (AgSTAR, 2012; Khalid et al., 2011; Braun, 2002).

Furthermore, a study by Macias-Corral et al., (2008) revealed that blending and digesting cow manure and organic fraction of municipal solid waste (OFMSW) enhances methane production. Other benefits derived from co-digestion include increase in biodegradation of organic materials (AgSTAR, 2012), stability and digestion rate improvement (Cornell et al., 2012) as well as higher mass conversion (Kangle et al., 2012), resulting in lower weight and volume of digestate (effluent). However, co-digestion has some disadvantages such as, an increase in digester effluent COD (chemical oxygen demand), need for additional pre-treatment, higher mixing requirement and higher energy requirement (Braun, 2002). Co-

digestion is a well-established practice in Europe, especially in Germany and Scandinavia (Appels et al., 2011; Environment Agency, 2010; Braun and Wellinger, 2009). Co-digestion plants for treating food waste and sewage sludge have been built in Germany, Switzerland and Denmark (Braun and Wellinger, 2009). The first co-digestion plant built in the UK is for the digestion of animal manure and food waste (Environment Agency, 2010). However, the co-digestion of food waste and sewage sludge is not in existence yet in the UK due to the different regulatory and management regimes guiding the digestion of the two substrates (Iacovidou et al., 2012; Environment Agency, 2010).

#### **3.4 Process modelling**

Process models have brought about simplification of processes and utilised to describe and predict the behaviour of processes as well as their required outputs (Sulaiman et al., 2011). Process models play an essential role in process understanding, process development, online diagnostic, and process automation (Craven et al. 2012; Sulaiman et al., 2011; Tham, 2000). Choosing the appropriate model depends on a number of factors such as the intended application, quality, quantity and the nature of the available experimental data (Craven et al. 2012; Lombardozzi and Sparks 2012; Müller et al., 2009; Bernacchi et al., 2009; Gómez et al., 2005; Hoffman et al., 2004; Boomen et al., 2002). Process models can be categorised as mathematical, statistical and quantitative. Mathematical models are further divided into mechanistic (white-box) and empirical (black-box) model as illustrated in Figure 3.1 (Vázquez-Cruz et al, 2014; Ljung, 2001).



Fig. 3.1. Classification of process models

# 3.4.1 Mathematical models description

The complexity associated with bioprocess, which includes AD process, means that modelling is required in order to describe and predict the behaviour of the process in real life. As mentioned above, mathematical models are classified into mechanistic (white-box) and empirical (black-box) models (Vázquez-Cruz et al, 2014; Ljung, 2001). However, there is also another type of mathematical model known as a gray-box model, which combines the characteristics of white-box and black-box models (Vázquez-Cruz et al, 2014; Ljung, 2014; Ljung, 2001). Henceforth in this thesis, mechanistic and empirical models are referred to as white-box and black-box models, respectively.

White-box model: This is the type of mathematical model that is constructed based on the prior knowledge of the chemical, physical and biological sub-processes of the system (Vázquez-Cruz et al, 2014; Ljung, 2001). It is constructed based on the kinetics and stoichiometry of the individual identified reaction (Fang, 2010). White-box model usually employs a state-space model construction method to define the internal structure of the complex biochemical and physico-chemical processes (Fang, 2010). White-box model can rely

on many assumptions and can be difficult to construct due to the complex nature of biochemical reactions (Yu et al., 2013; Fang, 2010)

The Monod kinetics model, which is one of the most used mathematical models for bioprocess is a white-box model (Craven et al. 2012; Liu et al., 2008; Zeng et al., 1998; Mitsdorffer, 1991; Glacken et al., 1989; Bree et al., 1988; Bergter, 1983; Chen and Hashimoto, 1978; Powell, 1967; Contois, 1959; Moser, 1958). The Monod kinetics model for bacterial growth depends on the relationship between specific growth rate and substrate concentration (Gerber and Span, 2008).

The relationship between the substrate flow rate and the volume of the digester is the dilution rate (Stanbury et al. 1995).

$$D = \frac{F}{V} \tag{3.1}$$

Where,

*D* is the dilution rate ( $h^{-1}$ ), which is the ratio of the volumetric flow rate of nutrient supplied to the digester and the volume of the digester *V* is the volume of digester ( $m^3$ )

*F* is the volumetric flow rate  $(m^{3}h^{-1})$ 

In addition, the rate of change of bacterial concentration is as follows:

$$\frac{\mathrm{d}X}{\mathrm{d}t} = (\mu - D) X \tag{3.2}$$

Where,

X is the bacterial concentration (mg/l)

*t* is the time of incubation (day)

 $\mu$  is the specific growth rate (day<sup>-1</sup>)

The bacterial growth is  $\mu X$ , while the bacterial output is DX.

Under steady-state conditions, the new substrate fed into the digester balances the loss of bacteria from the digester.

$$\frac{\mathrm{d}x}{\mathrm{d}t} = 0 \tag{3.3}$$

Then, Eq. 3.2 becomes,

$$\mu = D \tag{3.4}$$

According to Monod (1949), the nonlinear relationship between substrate concentration and specific bacterial growth rate is:

$$\mu = \mu_{\max} \frac{s}{\kappa_s + s} \tag{3.5}$$

Where,

 $\mu$  is the specific growth rate (day<sup>-1</sup>)

 $\mu_{\text{max}}$  is the maximum specific growth rate (day<sup>-1</sup>)

*S* is the substrate concentration (mg/l)

 $K_{\rm s}$  is the half-velocity constant (g/l), meaning the substrate concentration at one-half of the maximum growth rate ( $\mu_{\rm max}/2$ ).

The Monod model for bacterial growth shows that the specific rate of bacterial growth increases rapidly at low substrate concentration and vice versa until a bacterial saturation level is attained, as illustrated in Figure 3.2 (Gerber and Span, 2008). This indicates that the substrate concentration is the limiting factor for the bacterial growth rate. The saturation point is the maximum specific bacterial growth rate ( $\mu_{max}$ ).



Fig. 3.2. Monod specific growth rate

According to Contois (1959), the accuracy of the Monod model is high for pure bacterial cultures and simple substrates. In addition, Grady (1969) concluded that  $\mu_{max}$  is exclusive for individual bacterial cultures. However, Pfeffer (1974) suggested that the Monod model is insufficient to describe the degradation of complex substrate like municipal wastes. te Boekhorst et al. (1981) supported the finding of Pfeffer (1974) and added that the Monod model is sufficient for homogenous bacterial cultures, but cannot be used to describe processes involving heterogeneous bacterial cultures. Due to the shortfalls associated with Monod kinetics model, other authors modified the model as shown in Table 3.1.

Model	Author	Eq. No.
$\mu = \mu_{\max} \frac{S^n}{K_s + S^n}$	Moser, 1958	(3.6)
$\mu = \mu_{\max} \frac{S}{K_c + X + S} = \mu_{\max} \frac{S}{\frac{K_c \cdot X}{S} + 1}$	Contois, 1959	(3.7)
$\mu = \mu_{\max} \frac{(K+L+S)}{2 \cdot L} \cdot \left(1 - \sqrt{1 - \frac{4 \cdot L \cdot S^2}{(K+L+S)^2}}\right)$	Powell,1967	(3.8)
$\mu = \mu_{\max} \frac{S/S_1}{K + \frac{(1-K)\cdot S}{S_1}}$	Chen and Hashimoto, 1978	(3.9)
$\mu = \mu_{\max} \frac{s}{K_s + s} \cdot \left[1 - \exp\left(-\frac{t}{T}\right)\right]$	Bergter, 1983	(3.10)
$\mu = \mu_{\max} \frac{S^n}{S^n \cdot (1 + K_b \cdot G_s \cdot S^n)}$	Mitsdorffer, 1991	(3.11)

 Table 3.1. List of the modified Monod models

However, despite the various modifications of the Monod model for bacterial growth, it is concluded that these models with only one set of kinetic parameters are insufficient to describe biological processes of any retention time or degradation of complex substrates (Gerber and Span, 2008; Rao and Singh, 2004). Hence, the first-order models, which are also a type of white-box model, are developed (Gerber and Span, 2008; Rao and Singh, 2004). An example of the first-order model is the model developed by Rao and Singh, (2004).

$$k\left(\frac{\mathrm{d}s}{\mathrm{d}t} = -k\,.S\right) \tag{3.12}$$

Where,

k is the kinetics of bacterial growth due to enzyme activity

The first-order kinetics were applied in the hydrolytic step of the models developed by Bryers (1985), Angleidaki et al. (1999), Knobel and Lewis (2002) and Siegrist et al. (2002). However, Shin and Song (1995) applied the first-order kinetics for all the steps of the process. Although, the first-order models are easy to deal with, the accuracy is for confined requirements only (Hashimoto et al., 1981). Hashimoto et al. (1981) concluded that it is not possible to use first-order models for the prediction of optimum conditions of maximum biological activity or process failure.

Furthermore, the Anaerobic Digestion Model 1 (ADM1) developed by the International Water Association (IWA) task group in 2002 is a white-box, state-space model (Batstone et al., 2002). It utilised the theoretical concepts of material conservation and biochemical reactions to construct the structure of the model (Fang, 2010). The constituent bacterial population and processes of ADM1 are based on several assumptions in order to keep the model simple (Yu et al., 2013; Fang, 2010). In addition, the ADM1 requires several input parameters implemented in the form of differential algebraic equations (DAE) and differential equations (DE) (Yu et al., 2013; Batstone et al., 2002)

**Grey-box model:** This is a combination of the white-box and black-box model. It is applied when there is not enough theoretical knowledge of the system to construct a white-box model. Then, the construction of the model is based on the prior knowledge of the system dynamics and the estimation of the unknown model parameters from measured input-output data (Vázquez-Cruz et al, 2014; Krishna, 2010; Brus, 2005). The disadvantage of grey-box modelling is that it requires a large number of parameters for a detailed complex system such as AD process (Brus, 2005).

**Black-box model:** These models utilise data generated from the process to define the relationship between input and output variables (Vázquez-Cruz et al, 2014; Krishna, 2010). The block diagram of black-box model is shown in Figure 3.3.



Fig.3.3. Block diagram of black-box process

Black-box models depend on the availability of data for model estimation and validation (Vázquez-Cruz et al, 2014). According to Krishna (2010), the main disadvantage of black-box models is that the functions of the parameters lack deep physical significance – unable to describe process parameters such as mass transfer coefficients, heat and reaction kinetics. However, black-box models can adequately represent the trends in process behaviour and are as effective as the white-box models. In addition, black-box models have the advantage of

sufficiently modelling a process when knowledge about the process is vague or in the case of a complex process whereby it is impossible to solve the resulting equations (Krishna, 2010). Furthermore, in a number of situations, it is impractically feasible to apply white-box models due to financial and time constraints; hence, black-box models are utilised (Krishna, 2010).

The Black-box models are further classified into linear and nonlinear forms. The linear category is made up of the main transfer function and time-series models, and the nonlinear type is composed of neural network and time-series, as illustrated in Figure 3.4 (Vázquez-Cruz et al, 2014; André, 2013; Kumar and Zhao, 2011; Krishna, 2010; Stein et al., 2007; Aufhammeret et al., 2006).



Fig.3.4. Classification of black-box models

# 3.4.2 Linear black-box models

For general linear models in time-domain, the following equation describes the model.

$$y(t) = G(z^{-1}, \theta)u(t) + H(z^{-1}, \theta)\varepsilon(t)$$
 (3.13)

Where,

u(t) is the system input at time t

y(t) is the system output at time t

 $G(z^{-1}, \theta)$  is the transfer function of the deterministic part of the system

 $H(z^{-1}, \theta)$  is the transfer function of the stochastic part of the system

e(t) is the disturbance of the system which is usually zero-mean white noise

 $z^{-1}$  is the backward shift operator

The  $G(z^{-1}, \theta)$  specifies the relationship between the input and output variable. Whilst the  $H(z^{-1}, \theta)$  specifies the effects of the random disturbance on the output variable (Ljung, 2001).

For linear black-box models, G and H in Eq. 3.13 are rational transfer functions in the delay operator with unknown numerator and denominator polynomials defined by the following equations (Ljung, 2001):

$$G(z^{-1}, \theta) = \frac{B(z, \theta)}{A(z, \theta)F(z, \theta)}$$
(3.14)

Similarly,

$$H(z^{-1}, \theta) = \frac{C(z,\theta)}{A(z,\theta)D(z,\theta)}$$
(3.15)

The perimeter vector  $\theta$  is the set of model parameters that contain the coefficients  $b_i$ ,  $c_i$ ,  $d_i$  and  $f_i$  of the transfer functions, which are eliminated in the following equations to make them appear simpler. Hence, Eq. 3.13 becomes;

$$A(z)y(t) = \frac{B(z)}{F(z)}u(t-n) + \frac{C(z)}{D(z)}\varepsilon(t)$$
(3.16)

Where,

n is the system delay

A(z), B(z), C(z), D(z) and F(z) are polynomials in connection with the backward shift operator and described by the following equations.

$$A(z) = 1 + a_1 z^{-1} + \dots + a_{na} z^{-na}$$
(3.17)

$$B(z) = 1 + b_1 z^{-1} + \dots + b_{nb} z^{-nb}$$
(3.18)

$$C(z) = 1 + c_1 z^{-1} + \dots + c_{nc} z^{-nc}$$
(3.19)

$$D(z) = 1 + d_1 z^{-1} + \dots + d_{nd} z^{-nd}$$
(3.20)

$$F(z) = 1 + f_1 z^{-1} + \dots + f_{nf} z^{-nf}$$
(3.21)

Where *na*, *nb*, *nc*, *nd*, and *nf* are structural parameters.

However, when one or more of A(z), C(z), D(z) and F(z) are set to 1, then simpler models are created such as Box-Jenkins (BJ), output error (OE), autoregressive with exogenous terms (ARX) and autoregressive-moving average with exogenous terms (ARMAX) models. For instance, when C(z), D(z) and F(z) are set to one, ARX model is created.

$$A(z)y(t) = B(z)u(t-n) + \varepsilon(t)$$
(3.22)

When A(z), C(z) and D(z) are equal to 1, the resultant model is the OE model.

$$y(t) = \frac{B(z)}{F(z)}u(t-n) + \varepsilon(t)$$
(3.23)

When A(z) is equal to 1, the general linear model becomes the BJ model.

$$y(t) = \frac{B(z)}{F(z)}u(t-n) + \frac{C(z)}{D(z)}\varepsilon(t)$$
(3.24)

When D(z) and F(z) are equal to 1, the general linear model is reduced to ARMAX model.

$$A(z)y(t) = B(z)u(t-n) + C(z)\varepsilon(t)$$
(3.25)

Figure 3.5 represents the different linear black-box models.



Fig. 3.5. Different linear black-box models

Furthermore, the predictor associated with Eq. 3.16 is of "pseudo-linear" regression form (Sjoberg et al., 1995; Liung and Soderstrom, 1983).

$$\hat{\mathbf{y}}(t|\theta) = \theta^{\mathrm{T}} \,\varphi(t,\,\theta) \tag{3.26}$$

The regressors, which are the components of  $\varphi(t, \theta)$  are in the following general forms (Sjoberg et al., 1995):

- u(t-k) is associated with the *B*-polynomial
- y(t-k) is associated with the *A*-polynomial

- *ŷ<sub>u</sub>(t|θ)* is the simulated output from past input (*u*) only and is associated with the *F* polynomial
- $\varepsilon(t-k) = y(t-k) \hat{y}(t-k|\theta)$  is the prediction error associated with the *C*-polynomial
- $\varepsilon_u(t-k) = y(t-k) \hat{y}_u(t-k|\theta)$  is the simulation error associated with the *D*-polynomial

In addition, linear state-space models are common representations of dynamic models. They are similar to the ARX models, describing the same kind of linear relationship between inputs and outputs (Ljung, 2002).

$$x(t+1) = Ax(t) + Bu(t) + K\varepsilon(t)$$
(3.27)

$$y(t) = Cx(t) + Du(t) + \varepsilon(t)$$
(3.28)

Where,

x(t) is the vector of state variables

A, B, C and K are the matrices

If K = 0, it means that noise source  $\varepsilon(t)$  affects only the output. Also, if D = 0 it means that there is no direct influence from u(t) to y(t).

# 3.4.3 Nonlinear black-box models

The nonlinear black-box models are subdivided into time-series and neural network based models as shown in Figure 3.1. In the nonlinear time-series models, the behaviour of the process is modelled by combining the weighted cross-products and powers of the variables in the representation (Krishna, 2010). However, the perimeters of the functions remained linear, thereby facilitating the identification process.

Black-box models can be dynamic or static depending on their relationship with time. If the observed inputs (u) and outputs (y), form a dynamic system:

$$u^{t} = [u(1), u(2), ..., u(t)]$$
(3.29)

$$y^{t} = [y(1), y(2), ..., u(t)]$$
 (3.30)

Then, the general nonlinear dynamic black-box model is as follows:

$$y(t) = g(u^{t-1}, y^{t-1}) + \varepsilon(t)$$
(3.31)

Where,

y(t) is a vector that contains the outputs at time t

u(t) is a vector that contains the inputs at time t

 $g(u^{t-1}, y^{t-1})$  is the model structure to be selected

 $\varepsilon(t)$  is the noise on the output at time t

The additional term  $\varepsilon(t)$  in Eq. 3.31 denotes that the subsequent output, y(t), is not an exact function of the past data. However,  $\varepsilon(t)$  must be small so that given the past data,  $g(u^{t-1}, y^{t-1})$  is a good predictor of y(t).

In order to find the nonlinear function g in Eq. 3.31,  $g(u^{t-1}, y^{t-1})$  is parameterised with a finitedimensional parameter vector  $\theta$  (Ljung, 2001; Sjoberg et al., 1995). Then,  $g(u^{t-1}, y^{t-1})$  becomes  $g(u^{t-1}, y^{t-1}, \theta)$ . According to Sjoberg et al, (1995), this parameterisation is usually an approximation of the quality assessed by means of the fit between the model and the data.

Though, mapping the increasing number of past observations  $(u^t, y^t)$  into a finite dimensional vector  $\varphi(t)$  of fixed dimension,

$$g(u^{t-1}, y^{t-1}, \theta) = g(\varphi(t), \theta)$$
 (3.32)

Where,

$$\varphi(t) = \varphi(u^{t-1}, y^{t-1}) \tag{3.33}$$

In Eq. 3.33, the vector  $\varphi(t)$  is called regression vector, while its components on the right-hand of the equation are referred to as regressors (Sjoberg et al., 1995). Parameterising the regressors gives:

$$\varphi(t) = \varphi(u^{t-1}, y^{t-1}, \theta)$$
(3.34)

In addition, similar to the linear black-box regression structure in Eq. 3.26, the nonlinear case is of the following form.

$$\hat{\mathbf{y}}(t|\theta) = g(\varphi(t), \theta) \tag{3.35}$$

For the input and output case, u(t-k) and y(t-k) are measured variables. While  $\hat{y}_u(t|\theta)$ ,  $\varepsilon(t-k) = y(t-k) - \hat{y}(t-k|\theta)$  and  $\varepsilon_u(t-k) = y(t-k) - \hat{y}_u(t-k|\theta)$  are based on preceding outputs from the black-box model  $\hat{y}(t-k|\theta)$ . If the measured outputs y(t-k) in the regressors are substituted by the last simulated output  $\hat{y}(t-k|\theta)$  then, the output from Eq. 3.35 becomes equal to  $\hat{y}_u(t|\theta)$  (Sjoberg et al., 1995). The following are examples of nonlinear black-box model structures.

- Nonlinear autoregressive with exogenous terms (NARX) models use u(t k) and y(t k) as regressors
- Nonlinear output error (NOE) models use u(t-k) and  $\hat{y}_u(t-k|\theta)$  as regressors
- Nonlinear autoregressive moving average with exogenous terms (NARMAX) models use u(t-k), y(t-k) and  $\varepsilon(t-k|\theta)$  as regressors
- Nonlinear Box-Jenkins (NBJ) models use u(t k), ŷ(t k|θ), ε(t k|θ) and ε<sub>u</sub>(t k) as regressors
- Nonlinear Hammerstein-Wiener (NLHW) models use u(t k) and  $\hat{y}_u(t k|\theta)$  as regressors

However, the NOE, NARMAX and NBJ as well as the nonlinear state-space models are considered recurrent structures (Ali, 2010; Sjoberg et al., 1995). This is because the past

outputs from the models have to be fed back into the model computation, thereby required more considerable computation to fit the data (Ali, 2010; Sjoberg et al., 1995). In addition, it is difficult to check the conditions under which the obtained predictor model is stable (Sjoberg et al., 1995).

The black-box modelling method has been applied in a number of AD systems, such as the model of Beck (1980), which utilised black-box method for the identification of a single-input/single-output (SISO) model for volatile acid concentration as input and volumetric gas flow rate as output. Also, black-box models were used to model the combustion process of the internal combustion engine (Maass, 2011).

# Neural network

The neural network has gained prominence in nonlinear modelling (Billings, 2013a). It is a computational model that interconnects a large number of computational units called neurons, to form a network capable of performing complex computational tasks (Billings, 2013b; Sulaiman et al., 2011). Generally, with the use of certain learning algorithm, the neural network receives training to learn and to represent the data set (Billings, 2013a). The process of determining the weights that decide the strength of the connection between the neurons in the network and enable it to model the mechanism that yielded the data set involves learning a mathematical expression of the system (Billings, 2013a).

The neural network has the advantage of robustness, suitable for nonlinear models, very adaptive in nature and able to function even when there is a failure in one of the elements (Billings, 2013a). It does not require the prior knowledge of the interrelationships that existed between the input and output variables. Rather, what it needed to function is the specification

of the network design and enough quantity of steady input data (Yu et al., 2012). However, the limitation of a neural network for AD process model development is the difficulty in utilising the model to design a digester or scale up AD system (Yu, 2013). This is due to the various level of input-output relationships required to train the model for real life process application, which could be susceptible to overfitting or underfitting, as well as takes a long time to process large networks (Tabatabaei et al., 2010).

# 3.5 Summary

This chapter has presented some improvements made in AD technology. Pre-treatment is an important development in AD process as it reduces the substrate particle size, especially in hydrolysis. Consequently, it increases the available specific surface area that facilitates biodegradation, which in effect enhances biogas production, as well as reducing the quantity of residue generation. In addition, co-digestion of two or more substrates has shown to enhance biogas production. This is due to the combined positive effect achieved by the supply of the required nutrients from the co-substrates involved. Furthermore, the chapter discussed the different mathematical models, mainly the white-box and black-box models. The various mathematical models provide the possibility of describing and predicting the behaviour of AD process. The discussion in the next chapter focuses on the experiments conducted in this study.

# Chapter 4 Experimental Apparatus and Methodology

### 4.1 Introduction

The AD is a multi-step process involving several groups of anaerobic bacteria. These groups of bacteria promote a series of sequential biochemical reactions/processes in converting organic material to biogas (Baker and Evans, 2009). The biochemical processes consist of the following four main steps; hydrolysis, acidogenesis, acetogenesis and methanogenesis (Korres et al., 2013; Zieminski and Frac, 2012; Nwuche and Ugoji, 2010; Amani, 2011; Al Seadi et al., 2008). This chapter presents the discussion of the biochemical processes of AD that result in biogas production. It also discussed the theoretical chemical conversion of complex organic compounds to molecules of methane and carbon dioxide. Furthermore, the chapter describes the laboratory experiments conducted in this study, which investigates the influence of temperature, pH, mixing speed and pressure on biogas and methane production.

# 4.2 Bacteriological and biochemical processes in AD

The AD is the process by which bacteria breakdown organic matter in the absence of oxygen to produce biogas and digestate (NNFCC, 2011). Biogas is a mixture of methane (CH<sub>4</sub>), carbon dioxide (CO<sub>2</sub>) and traces of other gases, including ammonia (NH<sub>3</sub>), hydrogen (H<sub>2</sub>) and hydrogen sulphide (H<sub>2</sub>S) (NNFCC, 2011; Hiremath et al., 2009; Gallert and Winter, 2005; Kelleher et al., 2002; McKendry, 2002; Veeken et al., 2000).

The AD is a process that depends on the complex interaction of different species of bacteria (Kangle et al., 2012; Baker and Evans, 2009). In order to achieve a stable AD process and optimal biogas production, it is important to maintain the balance between the various bacterial groups (Wang, 2014). This balance could be affected by changes in the physico-environmental
conditions, which could result in accumulation of intermediary products (short-chained organic acid molecules) that may inhibit the entire AD process (Hooshyari et al., 2009; Wong et al., 2009; Liu et al, 2009; de Bok et al, 2004). The short-chained organic acid molecules are called volatile fatty acids (VFAs) (Marchaim, 1992).

The conversion of organic material to biogas using AD technology is a complex biochemical reaction that is composed of successive multi-step processes and parallel reactions (Korres et al., 2013; Kangle et al., 2012; Khalid et al., 2011; McCarty, 1982). The degradation process of complex organic polymers involves interrelated steps that facilitate the catabolism of organic material to smaller units. Each of the stages is performed by a specific group of bacteria, which successively break down the product of the previous step. The putrefaction of organic material takes place in four sequential steps, which are hydrolysis, acidogenesis, acetogenesis and methanogenesis illustrated in Figure 4.1 (Korres et al., 2013; Zieminski and Frac, 2012; Al Seadi et al., 2008; Veeken et al., 2000; Marchaim, 1992).

**Hydrolysis:** This is the first stage of AD process, at which organic polymers are broken down into soluble monomers by hydrolytic enzymes excreted by hydrolytic bacteria (Kangle et al., 2012). At this step, carbohydrates are hydrolysed into glucose, proteins converted into amino acids while lipids are transformed into glycerol and fatty acids (Zieminski and Frac, 2012; Nayono, 2009; Al Seadi, 2008). The conversion of glucose into simple sugar is illustrated in Eq. 4.1 (Korres et al., 2013; Ostrem, 2004; Themelis and Verma, 2004).

$$C_6H_{10}O_4 + 2H_2O \longrightarrow C_6H_{12}O_6 + 2H_2$$
 (4.1)



Fig.4.1. Stages in AD process

Acidogenesis: The second stage of AD process is also referred to as fermentation. The products of hydrolysis, simple sugars, amino acids and fatty acids are catabolised by acidogenic bacteria to form acetate, carbon dioxide, hydrogen, VFAs and alcohols. At this stage, acetate can be utilised by methanogenic bacteria and converted directly to methane (Nayono, 2009; Al Seadi, 2008). The following three chemical equations illustrate the three different conversions of

glucose by acidogenic bacteria into ethanol, propionate acid and acetic acid, as signifies in Eq. 4.2, Eq. 4.3 and Eq. 4.4, respectively (Korres et al., 2013; Ostrem, 2004; Themelis and Verma, 2004; Bilitewski et al., 1997; Marchaim, 1992).

$$C_6H_{12}O_6 \iff 2CH_3CH_2OH + 2CO_2 \tag{4.2}$$

$$C_6H_{12}O_6 + 2H_2 \iff 2CH_3CH_2COOH + 2H_2O$$

$$(4.3)$$

$$C_6H_{12}O_6 \longrightarrow 3CH_3COOH \tag{4.4}$$

Acetogenesis: At this stage, the products of acidogenesis that have not yet transformed to methane by methanogenic bacteria are converted into the methanogenic substrate by acetogenic bacteria. Low molecular VFAs are oxidised into acetate, hydrogen gas and carbon dioxide (Nayono, 2009; Al Seadi et al., 2008; Ostrem, 2004; Themelis and Verma, 2004). Also in acetogenesis, VFAs that have carbon chains longer than two units, as well as alcohols with carbon chains longer than one unit are oxidised into acetate or propionate and hydrogen gas (Al Seadi et al., 2008). The presence of hydrogen in this phase raises the hydrogen partial pressure that can inhibit the oxidation process (Al Seadi et al., 2008). For continuous conversion of VFAs into acetate, hydrogen gas and carbon dioxide, it is required to reduce the hydrogen partial pressure enough to thermodynamically let the process proceed (Nayono, 2009; Ostrem, 2004). Lowering the hydrogen partial pressure can be achieved by the presence of hydrogen-consuming bacteria in the digestion process, leading to the conversion of all the VFAs (Nayono, 2009; Ostrem, 2004). The concentration of hydrogen measured by its partial pressure can be an indicator of the performance of AD process (Mata-Alvarez, 2003). Examples of acetogenic reactions are illustrated in the conversion of propionate, glucose, ethanol and bicarbonate to acetate in Eq. 4.5, Eq. 4.6, Eq. 4.7 and Eq. 4.8, respectively (Korres et al., 2013; Ostrem, 2004; Marchaim, 1992; McInerney et al. 1981).

$$CH_3CH_2COO^- + 3H_2O \iff CH_2COO^- + H^+ + HCO_3^- + 2H_2$$

$$(4.5)$$

$$C_6H_{12}O_6 + 2H_2O \iff 2CH_3COOH + 2CO_2 + 4H_2$$

$$(4.6)$$

$$CH_3CH_2OH + 2H_2O \iff CH_3COO^- + 2H_2 + H^+$$

$$(4.7)$$

$$2HCO_3^- + 4H_2 + H^+ \longleftrightarrow CH_3COO^- + 2H_2O$$

$$(4.8)$$

**Methanogenesis:** This is the final stage of AD process; the VFAs are converted to methane and carbon dioxide, which is biogas, by the activities of methanogenic bacteria. The conversion of acetate to methane and carbon dioxide is presented in Eq. 4.9 and Eq. 4.10, while the transformation of alcohol into biogas is shown in Eq. 4.11. Furthermore, the conversion of hydrogen and carbon dioxide to biogas is presented in Eq. 4.12 (Korres et al., 2013; Nayono, 2009; Al Seadi et al., 2008; United Tech, 2003; Omstead et al., 1980; Thauer et al., 1977).

$$2CH_{3}CH_{3}OH + CO_{2} \longrightarrow 2CH_{3}COOH + CH_{4}$$

$$(4.9)$$

$$CH_3COOH \longrightarrow CH_4 + CO_2 \tag{4.10}$$

$$2C_2H_5OH + CO_2 \longrightarrow 2CH_3COOH + CH_4$$

$$(4.11)$$

$$CO_2 + 4H_2 \longrightarrow CH_4 + 2H_2O \tag{4.12}$$

The stability of anaerobic process depends on methanogenesis. This is due to the slower growth of methanogenic bacteria than other groups of anaerobic bacteria involved, besides very sensitive to changes in operating parameters (Wang, 2014; Zupancic and Grilc, 2012). Because of these reasons, methanogenesis is considered the rate-limiting step of AD process (Zupancic and Grilc, 2012). Methanogenic bacteria multiply and perform optimally at neutral pH condition (Nayono, 2009; Al Seadi et al., 2008; Gas Technology, 2003). Sudden temperature

change can inhibit the activity of methanogenic bacteria, resulting in less or no methane production (Çalli, 2012; Gao et al., 2011; Ahn and Forster, 2002). According to Davis and Cornwell (1998), the kinetics of methanogenesis determines the kinetics of the entire anaerobic biochemical process.

### 4.3 Further review of the operating parameters

The influence of operating parameters on biogas production is very challenging especially in the rural areas of developing nations (Pham et al., 2014; Cu et al., 2012). Millions of biogas production plants in countries like China, India, Tibet, Pakistan, Bangladesh and Vietnam are utilised mainly to produce biogas for cooking and lighting purposes (Bruun et al., 2014). Even in these simple biogas plants, the impacts of the operating parameters were critical. In addition to the review of the operating parameters in Chapter 2, the four operating parameters (temperature, pH, mixing speed and pressure) investigated in this research work are further discussed in this section.

For a reaction to occur, reactants require kinetic energy that facilitates their collusion with one another (Brown et al., 2009). Increasing the temperature of the reaction causes reactants' molecules to gain more kinetic energy that enables them to move about faster and collide frequently (Brown et al., 2009), thereby increasing the rate of reaction. Similarly, the rate of metabolism in bioprocess increases with temperature (Hoegh-Guldberg and Bruno, 2010). In AD process, the rate of anaerobic reaction increases as the operating temperature rises, but up to a certain temperature, known as the optimum temperature (Calli, 2012; Kent, 2000). Apart from increasing the rate of reaction by increasing the heat energy input, substrates utilised for biological process also require energy for the reaction to proceed (Starr and McMillan, 2014). The required energy, referred to as activation energy, is ideally supplied to the substrates by

the input heat energy (Dominiczak, 2007; Alters, 2000). In order to mitigate the influence of temperature, several simple biogas digesters in the countries like China and India are buried underground to maintain the temperature within the digesters (Pham et al., 2014, Kossman et al., 1997).

A number of researchers have investigated the effect of temperature on biogas production (Ma et al., 2013; Gavala et al., 2003; Chen and Hashimoto, 1978). Usman et al. (2012) investigated the influence of temperature on biogas production from two maize samples, which were digested at different temperatures, 30 °C, 45 °C, and 60 °C. The goal of the study was to determine the most suitable temperature for the process, that is, the temperature that achieved the highest biogas production from the given maize samples (Usman et al., 2012). Pandey and Soupir (2012) also conducted a study to determine how the given temperatures (25 °C, 37 °C, and 52.3 °C) influenced biogas production from AD of dairy manure. Furthermore, Zhao (2011) investigated how a change in temperature affected the production of biogas from the treatment of domestic wastewater. The results of the reviewed studies show that without the right digester temperature the production of biogas is limited and can make AD less reliable and unattractive as a means of sustainable energy source.

Similarly, the importance of stabilising the process pH within the optimum range in AD cannot be overemphasised. As discussed in section 2.2.1, the impact of pH variation on biogas production is very significant. It shows that a significant decrease in pH below 7.0, the neutral value, could result in the acidic slurry in the digester, which could inhibit biogas production. However, the significant rise in pH above the neutral level causes the slurry to become alkaline, thereby limiting or halting biogas production. For this reasons, many studies have been conducted with the goal of investigating the impact of pH variation on biogas production. Sumardiono et al. (2013) investigated the effect of COD/N (Chemical Oxygen Demand/Nitrogen) ratios of the substrate and pH control on biogas production from vinasse. The author found that at COD/N ratio of 600/7, the controlled pH sample had about 10% COD removal more than that of the fluctuating pH. The author concluded that the biogas production at the controlled neutral pH is more than the biogas generated at the fluctuating pH.

Another study was carried out utilising four similar samples of dairy wastewater, but with variable starting pH values (Kheiredine et al., 2014). It was found from the biodegradation test that the highest COD removal of 90.8% was recorded for the sample with initial pH of 7.0, followed by 79.64% for 9.5 pH, 63.75% for 5.5 pH and 49.11% for 4.0 pH. The experiment further shows that the highest methane content was produced by the sample that has the starting pH of 7.0, and it followed the same sequence as the results obtained from the biodegradation test, with negligible methane production as the pH decreases below 5.5 (Kheiredine et al., 2014). The biodegradation process signifies the degree and rate at which organic material is broken down by bacteria, and in this case, to produce methane and carbon dioxide (Kheiredine et al., 2014). Furthermore, Budiyono et al. (2013) conducted a similar biodegradation test with three samples of bioethanol waste containing different pH values at the start of the digestion process. The results obtained indicated that the sample that started with pH of 7.0 generated the highest biogas production of 3.81 mL/g COD, while the other two samples with initial pH of 6.0 and 8.0 produced 3.25 mL/g COD and 3.49 mL/g COD of biogas, respectively (Budiyono et al., 2013).

Similar to temperature and pH, the significance of mixing in AD to improve biogas production has been widely acknowledged (Sindall et al., 2013). However, relatively, less emphasis has

been made on exploring the specific impact of mixing speed on the production of biogas (Sindall et al., 2013). A number of researches were carried out to determine the effect of mixing speed on biogas production. A study conducted by Hoffmann et al. (2013), on the effect of mixing intensities on biogas production, utilising cow manure as substrate. In that study, four continuously stirred digesters containing similar samples at four different mixing speeds of 1500, 500, 250 and 50 rpm are operated. The author found that the performance of the digester stirred at 1500 rpm was adversely affected, leading to lower biogas production and higher VFA concentration. In contrast, the digester that operated at 500 rpm achieved the highest biogas production. Also observed is that the other two digesters operated at 250 and 50 rpm produced biogas marginally at the same rate, but higher than the 1500 rpm digester (Hoffmann et al., 2013).

Furthermore, another research utilised velocity gradient, through computational fluid dynamics (CFD) to determine the influence of mixing speed on biogas production. Velocity gradient is the velocity variance between adjacent layers of fluid (CEE, 2012). The author used velocity gradient as a proxy for mixing intensity of turbulence (Sindall et al., 2013). The results from the study showed that as the velocity gradient was lowered from 9.7  $s^{-1}$  to 7.2  $s^{-1}$ , which represented a reduction in the mixing speed from 100 to 50 rpm, increase in biogas production is recorded. Nevertheless, as the velocity gradient increased from 9.7  $s^{-1}$  to 14.3, for increasing mixing speed from 100 to 200 rpm, biogas production was found to decrease (Sindall et al., 2013).

The operating pressure is the last parameter investigated; however, less emphasis has been made on evaluating its effect on biogas and methane production (Chen et al., 2014). Research conducted by Abe and Horikoshi (2001) and cited by Chen et al. (2014), reported that

methanogenic bacteria can be metabolically active at a pressure up to 100 bar. In addition, it is reported that pressurised anaerobic digester can improve methane production, thereby, refining the quality of biogas (Chen et al., 2014; Lindeboom et al., 2011; Hayes et al., 1990). Increasing digester pressure liquefies the  $CO_2$  content of biogas, which decreases the overall biogas production, but improves the quality of biogas produced as it contains a higher proportion of methane (Lindeboom et al., 2011).

Similarly, a study performed by Chen et al. (2014) to investigate the impact of two different digester pressure, 1.5 and 9 bar, on biogas production, revealed that biogas generated by the 9 bar digester contained about 74.5 % methane. Whilst biogas produced by the 1.5 bar digester contained approximately 66.2 % methane. The outcome of this experimentation signifies that increasing the operational digester pressure can improve the methane content of the biogas produced.

The reason why CO<sub>2</sub> liquefied in the two studies cited is that it has a higher solubility in water compared to methane. From the solubility table, the solubility of CO<sub>2</sub> and methane in water (1ml/100ml) at 20 °C and 1atm is 0.782 and 0.032 g/kg, respectively (Yalkowsky et al., 2010). This means that CO<sub>2</sub> dissolves in water more than methane at the same temperature and pressure. According to Henry's law, the solubility of a gas in a liquid is directly proportional to the pressure of that gas above the surface of the solution (Yalkowsky et al., 2010). However, according to Japan-Agency for Marine-Earth Science and Technology (2007), it is difficult to explain the effects of pressure on complex metabolic systems, like AD, based on a simple volume law. Consequently, it is difficult to describe the influence of pressure on the actual production of biogas.

This study has shown the impact of each of the operating parameters evaluated not only on biogas production but also on the quality of the produced biogas. However, the following section presents the discussion of the laboratory experimentation of the combined effect of temperature, pH, mixing speed and pressure on biogas production and the methane content of the produced biogas under mesophilic condition.

### **4.4 Experimental procedures**

This section contains the methodology and steps of all the laboratory experimentations conducted in this research work. It covers the preparation of the bacteria culture media and the substrate (organic samples), as well as the experimental setups. A total of 27 experiments were performed, involving the four operating parameters and each repeated four times. The experimental design of 26 of them is in such a way that only one parameter is manipulated while the other three operating parameters are maintained at the same given value throughout the investigation for all the variables of the manipulated parameter. The remaining experiment is performed with all the four investigated operating parameters manipulated simultaneously in three different intervals of eight hours each. Furthermore, this research work investigated the impact of co-digestion of wastewater sludge (WWS) and food waste (FW) on biogas and methane production. The study provided more insight on the behaviour of mesophilic anaerobic bacteria and the response of biogas production. The obtained results are presented in Chapter 5, which are utilised to construct the process model. In addition, all the data generated from the experiments are found in appendix A.

### 4.4.1 Preparation of bacterial culture media

Bacteria culture is a medium by which bacteria is supplied with all the elements (nutrients) they require for growth (Bauman, 2007). The media for this study is prepared by weighing out

1.0 g of glucose into a conical flask, mixed with 90 mL of deionised water. The solution was inoculated with 10 mL of WWS, obtained from Thames Water Rye Meads Sewage Treatment Plant, Stanstead Abbortts, UK, to make up the solution to 100 mL. The vigorously stirred flask was covered airtight with a rubber cork. The anaerobic environment was achieved within the conical flask by supplying it with nitrogen through a tube for 20 minutes, which displaced the dissolved oxygen present. Afterwards, the prepared bacteria culture was placed in an incubator operating at 39 °C for 12 hours. The prepared bacteria culture media was utilised to inoculate all the samples in this study, except for samples for co-digestion tests. The setup for bacterial culture preparation is shown in Figure 4.2.



Fig. 4.2. Bacterial culture preparation

### 4.4.2 Impact of variable temperature on biogas production

The experimentation was carried out with four units of 2 L volume New Brunswick BioFlo 111 Batch/Continuous fermentors (digesters), operated in batch mode under mesophilic condition. The sample fed to the digesters contains a mixture of 10 g of glucose, 13 g of nutrient broth, 900 mL of deionised water and inoculated with 100 mL of the bacterial culture previously prepare, to make up the working volume to 1 L. In order to make the environment inside the digesters anaerobic (free from oxygen), nitrogen was supplied into the digesters for 30 minutes and dissolved oxygen contained in the digesters was displaced. The digesters were continuously stirred at 250 rpm while the digester pressure was controlled at 0.7 bar. The sample in the digester was retained for 24 hours.

The pH was maintained automatically at 7.2 by the supply of 2 M concentration of Sodium Hydroxide (NaOH) solution through the base peristaltic pump. The operating temperatures investigated are 32 °C, 34 °C, 36 °C, 38 °C, 40 °C and 42 °C. The biogas was collected through a tube inserted into the headspace of digesters to allow venting of biogas into a Tedlar gas sample bag attached to it. The cross-section of the digester and the experimental setup are shown in Figures 4.3 and Figures 4.4, respectively. The biogas and methane production were recorded every hour, using Biogas 5000 portable gas analyser connected to the gas collection tube. The experiments were repeated four times for each of the operating temperatures and the data obtained are analysed and discussed in Chapter 5.



Fig. 4.3. Cross-section of New Brunswick BioFlo 111 Digester



Fig. 4.4. Setup of BioFlo 111 Digester

The technical specifications for both Biogas 5000 gas analyser (Geotech, 2012) and New Brunswick BioFlo 111 Batch/Continuous fermentor (New Brunswick, 2010) are presented in Table 4.1 and Table 4.2, respectively.

Gas ranges					
Gases measured	CO <sub>2</sub> and CH <sub>4</sub>	By dual w channel	avelength infrared	sensor with reference	
Standard gas cells	Cell	Range	Typical accuracy (range : accuracy)	Typical accuracy (range : accuracy)	
	CH <sub>4</sub>	0-100%	0-70% : ±0.5% (vol)	70-100% : ±1.5% (vol)	
	CO <sub>4</sub>	0-100%	0-60% : ±0.5% (vol)	60-100% : ±1.5% (vol)	
Pump					
Flow	550 ml/min typ	550 ml/min typically			
Flow fail point	-200 mbar vacuum - user settable				
Maximum vacuum restart	-375 mbar approximately with flow rate of approx 80ml/min				
	Faci	ilities			
Temperature measurement	$-10^{\circ}$ C to $+75^{\circ}$ C with optional probe				
Temperature accuracy	±0.5°C with optional probe				
Flow measurement	Via Pitot tube, orifice plate, or anemometer				
Alarm	User selectable alarms				
Communications	Via USB lead or wireless Bluetooth				
Relative pressure measurement	±500 mbar				
Relative pressure accuracy	$\pm 4$ mbar typically (should be zeroed before reading) to $\pm 15$ mbar max				
Barometric pressure measurement	500 to 1500 m	500 to 1500 mbar, ±5 mbar accuracy			
Available memory	10 IDs *, 500 r	readings			
Environmental conditions					
Operating temperature range	-10°C to +50°C	$-10^{\circ}\text{C} \text{ to } +50^{\circ}\text{C}$			
Atmospheric pressure range	700 to 1200 mbar				
Relative humidity	0-95% noncondensing				
Certification rating					
ATEX	II 2G Ex ib IIA T1 Gb (Ta = $-10^{\circ}$ C to $+50^{\circ}$ C)				
MCERTS	MC/130240/00				
ISO17025	Calibration to	Calibration to UKAS certificate number 4533			
CSA	Ex ib IIA T1 (Ta= -10°C to +50°C) (Canada), AEx ib IIA T1 (Ta= -10°C				
	to +50°C) (US	A)			

### Table 4.1 Biogas 5000 technical specifications

Vessel	Working Volume	0.48-1.12L		
v esser	Total Volume			
Controller	Master Control	Controls 1-4 vessels 32 control loops per vessel: stores 10 recipes		
controller	Station	& 8 process variables per vessel for trend graphing. Includes an		
	Station	industrial touchscreen monitor/user interface. 3 built-in pumps &		
		connectors for all utilities & communications signals used by		
		fermentor/bioreactor 1		
	Utility Station	One each required for optional 2nd 3rd or 4th slave		
Ounty Station		fermentors or bioreactors. Each includes 3 built-in pumps &		
		connectors for all utilities & communications signals for its		
		individual fermentor/bioreactor		
	Touchscreen	15-inch industrial monitor canable of supporting up to 4		
	Interface/Display	formentors/bioreactors. One is standard with the		
Interface/Display		Master Control Station, Optional 2nd touchscreen available for use		
		with slave fermentors/bioreactors to replicate the image shown on		
		the Master display		
Tomporatura	Indication	Digital display in 0.1°C increments		
remperature	Dongo	From $5^{\circ}$ C above coolant temperature to $80^{\circ}$ C (cetting range: $A_{-}$		
	Kange	80°C).		
	Control	PI control employing PWM of heater and cooling water		
	Sensor	Platinum RTD probe		
Agitation	Drive	Permanent magnet motor with high torque input.		
	Indication	Digital display in 1 RPM increments.		
	Range	50-1200 RPM		
	Control	PI-controlled		
	Sensor	Optical photoplastic disc 500 lines/rev with quadrature output.		
	Impellers	2 six-bladed Rushton turbine impellers provided		
Exhaust	Filter	0.2µm disposable filter		
	Condenser	Stainless steel, water-cooled in headplate		
Aeration	4-Gas System	Up to 4 gases, including air, N2, CO2 & O2, delivered to ring		
		sparger		
	Sparger	Ring sparger		
	Inlet Filter	0.2µm absolute disposable filter		
	N2 Gas	For calibration of DO probe		
рН	Indication	Digital display in 0.01 pH increments		
	Range	2-12 pH		
	Control	P&I		
	Sensor	pH gel-filled probe		
DO	Indication	Digital display in 0.1% increments		
	Range	0-200%		
	Control	P&I, Agitation, O2 Enrichment. Also GasFlow Rate if		
		equipped with mass flow controller		
	Sensor	Polargraphic probe		
Other Sensors	Foam/Level	One foam/level sensor is standard		
	Options	Redox or second pH and second DO probes available		
Pumps	Pumps 1 & 2	Assignable peristaltic pumps		
-	•	Fixed speed (12 RPM) or variable duty cycle		
		Available control modes: Off, Prime, Base, Acid,		
		Foam, Levl2 Wet, Lvl2 Dry, Lvl 3 Wet or Lvl3 Dry.		
	Pump 3	Assignable peristaltic pump		
	•	Fixed speed (100 RPM) or variable duty cycle		
		Available control modes: Off, Prime, Base, Acid,		
		Foam, Levl2 Wet, Lvl2 Dry, Lvl 3 Wet or Lvl3 Dry.		
Utilities	Water	10 PSIG maximum, 50 µm filtration		
	Gas	10 PSIG maximum		
Ambient Operating Conditions		10-30 C, up to 80% relative humidity. non-condensing		

Table 4.2. New Brunswick BioFlo 111 technical specifications

### 4.4.3 Effect of variable pH on biogas production

The goal of the experiment is to determine how variable pH values influence biogas and methane production in the anaerobic biodegradation of glucose. The experimental setup and procedure are similar to that described in section 4.4.2, except that temperature was controlled at 39 °C. In addition, the operating pressure and mixing speed were maintained at 0.7 bar and 250 rpm, respectively, with HRT of 24 hours. The experiment was repeated four times for each of the investigated pH values; 5.0, 6.0, 7.0, and 8.0. The variable pH values were achieved by automatically adjusting the digester pH to the set-point through the addition of 2 M of hydrochloric (HCL) acid and 2 M of NaOH into the digesters. The biogas and methane production were measured every hour by gas analyser similar to that of the temperature discussed above. The outcome of the experiments is analysed and discussed in Chapter 5.

### 4.4.4 Influence of digester pressure on biogas and methane production

The investigation carried out to ascertain the impact of digester pressure on biogas and methane production is described in this section. The experimental procedure, setup and HRT are similar to that described in sections 4.4.1 and 4.4.2. The temperature, pH and mixing speed were regulated at 39 °C, 7.2 and 250 rpm, respectively. The initial digester pressure was set at 0.7 bar using vacuum/pressure pump and monitored by a pressure gauge. The experiment was repeated with different operation pressures, 0.5, 0.3 and 0.1 bar. In addition, four repetitions of the experiment for each of the given operation pressures were performed. The biogas production and methane proportion were monitored as described in section 4.4.2 and the findings are reported and discussed in Chapter 5.

### 4.4.5 Impact of mixing speed on biogas production

The effect of mixing speed on the biogas production and its methane concentration are also investigated in this study. Mixing as highlighted in section 2.2.7, facilitates contact between bacteria and nutrients from substrates (Meroney and Colarado, 2009; Wards et al., 2008), thereby improving digester performance. However, inappropriate mixing speed can affect the performance of digester as found in this study.

The methodology and experimental setup are similar to that discussed in section 4.4.3. The operating temperature was set at 39 °C, while the digester pH and pressure were maintained at 7.2 and 0.7 bar, respectively. The operating speed of the mixer was initially set at 60 rpm and the experiment was retained for 24 hours. Biogas was collected as usual through the tube inserted into the headspace of digester into a Tedlar gas sample bag while online biogas and methane production were monitored every hour via the gas analyser connected to the tube. The experiment was performed with three other operating mixing speeds; 300, 600 and 2000 rpm. The experiment for each of the mixing speeds was repeated four times and the data obtained is discussed in Chapter 5.

### 4.4.6 Effect of simultaneous manipulation of operating parameters

An experiment is conducted to evaluate the impact of simultaneously manipulating the operating parameters investigated in this study on biogas production. The experiment was repeated four times under mesophilic conditions. The data obtained from this investigation is utilised to test the estimated model. The parameters were manipulated simultaneously with three different values at eight hours interval. The temperature, pH, mixing speed and pressure were initially set at 35 °C, 8.5, 900 rpm and 0.3 bar, respectively. After eight hours, the values of the parameters were changed to 37 °C, 6.2, 500 rpm and 0.4 bar for temperature, pH, mixing

speed and pressure, respectively. Then at the 17<sup>th</sup> hour, the values of the parameters were altered again, in the same order, to 41 °C, 6.8, 120 rpm and 0.5 bar.

### 4.4.7 Influence of co-digestion on biogas production

As mentioned in section 3.1.2, co-digestion can improve nutrient balance, biodegradability, as well as digester performance and biogas production in AD. The methodology for the study was as follows:

### 1. Preparation of substrates

The substrates utilised for this experiment are wastewater sludge (WWS), obtained from Thames Water Rye Meads Sewage Treatment Plant, Stanstead Abbortts, Hertfordshire, and food waste (FW) collected from the cafeteria of the University of Hertfordshire, all in the UK. The activated sludge used as inoculum was obtained from Thames Water Rye Meads Sewage Treatment Plant. The FW consists of fish, rice, chicken, beef, bread, fresh vegetables and scrambled egg, which were ground with a grinding machine to reduce their particle size. All the FW samples collected throughout the study are of the same basic content.

### 2. Total solids (TS) determination

The TS content of both FW and WWS were determined by applying the Laboratory Analytical Procedure (LAP) methodology, described in LAP Determination of Total Solids in Biomass and Total Dissolved Solids in Liquid Process Sample (Sluiter et al., 2008, cited by Sluiter et al., 2010, Shekiro 111 et al., 2014 and Ritchie, 2014). For this study, TS was determined by drying the samples at 105 °C for 12 hours in an autoclave.

### 3. Co-digestion experiment

The experiment was performed with 3 L glass vessels (digesters), operated in batch mode, under mesophilic condition. The aim of this experimentation is to investigate the influence of the co-digestions of WWS and FW on the biogas production. The digester temperature was maintained at 39 °C and the mixing was performed by fabricated rotating shakers at 150 rpm. The volume of each sample in the digester was 2 L and the influent and effluent pH values for all the samples were noted. Furthermore, the environment within the digesters was made anaerobic by displacing the dissolved oxygen with nitrogen, which was supplied by a rubber tube inserted through the digester rubber cork. Each of the samples was retained in the digester for seven days. The experimental setup is illustrated in Figure 4.5 and biogas was harvested by a tube inserted into the headspace of digesters, which transferred the gas to a Tedlar gas sample bag. Gas analyser coupled to the tube monitored the rate of biogas production per day and the corresponding methane content. In order to achieve the ideal mixing ratio of the co-substrates, three different mass ratios of FW: WWS were tested at 30:70, 50:50 and 70:30. In addition, FW and WWS were tested as a single substrate, which was utilised as a reference for the mixed samples. Each of the experiments was repeated five times and the results are discussed in Chapter 5.



Fig. 4.5. Setup of co-digestion experiment

### 4.5 Summary

This chapter has discussed the bacteriological and biochemical processes that make up the AD process. These are successive multi-step and parallel reactions such as hydrolysis, acidogenesis, acetogenesis and methanogenesis, which are the action of a specific group of anaerobic bacteria for each step/process. Of all the processes methanogenesis is found to be the limiting step as it determines the effectiveness of the entire AD. The chapter also discussed the different experimentations to evaluate the effects of temperature, pH, mixing speed and pressure on the biogas production. Furthermore, it discussed the influence of the co-digestion of different ratios of FW and WWS on the biogas production. The data obtained from all the experiments conducted in this chapter are analysed and discussed in Chapter 5.

## Chapter 5 Results and Discussions

### **5.1 Introduction**

The experimental procedures for the impact of four operating parameters and co-digestion, under the mesophilic condition, on biogas and methane production, are discussed in Chapter 4. Because of the huge amount of data collected from the investigations conducted in this study, this chapter could not accommodate such amount of data, rather they are documented in appendix A. This chapter contains the analyses and discussions of the results collected from the experiments, which are presented in graphical forms.

### 5.2 The impact of temperature on biogas and methane production

This section discusses the relationship between temperature variation and biogas production in mesophilic AD of glucose. The effect of this relationship in terms of average biogas production, cumulative biogas production and total biogas production are plotted. Similarly, the methane content of the biogas at different digester temperatures are also presented in graphs.

### 5.2.1 Average rate and cumulative biogas production

The average biogas production rate and average cumulative biogas production are presented in Figures 5.1 and 5.2, respectively.



Fig. 5.1. Average rate of biogas production at different temperatures

Figure 5.1 presents the rate of biogas production from the digestion of glucose at 32, 34, 36, 38, 40 and 42 °C with HRT of 24 hours. As evident, the 36, 38, 40 and 42 °C digesters began biogas production at the 2<sup>nd</sup> hour, while the setups that operated at 34 °C, and 32 °C, commenced biogas production at the third and seventh hour, respectively. Indicating that the substrate and anaerobic bacteria received enough energy (activation energy) required for biodegradation to proceed from the 36, 38, 40 and 42 °C digesters within the second hour of the digestion process, which resulted in biogas yield within the second hour. Although the four digesters provided enough activation energy for the anaerobic reaction to commence, it is observed that the biogas production rate for 42 °C digester is the lowest at the beginning of the production, while the highest is recorded in the 40 °C digester, followed by the 38 °C digester and then the 36 °C digester. Indicating that 42 °C temperature is beyond the mesophilic optimal temperature range of the digester.

In addition, the 38, 40 and 42 °C digesters peaked at the ninth hour. Again, the digester that was operated at 42 °C generated the lowest peak, whereas the 40 °C digester produced the highest peak, followed by the 38 °C digester. The 36 °C digester is found to peak at the 12<sup>th</sup> hour, ahead of the 34 and 32 °C digesters that peaked at the 13<sup>th</sup> and 18<sup>th</sup> hour, respectively. This signifies that the 40 °C digester has the tendency to produce more biogas than the rest of the digesters within the first 10 hours of the digestion, followed by the digester that was operated at 38 °C.

Furthermore, the rate of biogas production remained positive for all the digesters up to the 20<sup>th</sup> hour, when the biogas production rate from the 42 °C digester became negative, meaning that the production of biogas was on the decline and lower than the production of the previous hour.



Fig. 5.2. Average cumulative biogas production at different temperatures

Similarly, the comparison of the average cumulative biogas produced at 32, 34, 36, 38, 40 and 42 °C as seen in Figure 5.2. It shows that the 40 °C digester-generated the highest average cumulative biogas production, followed by the 38 °C digester, then the third highest cumulative biogas is produced in the 36 °C digester. Whilst the 32 °C digester yielded the lowest average cumulative biogas, followed by the 34 °C digester and then the 42 °C digester. It is seen that the average cumulative biogas produced in the 42 °C digester is not the lowest, however, Figure 5.2 shows a significant drop in biogas production as digester temperature was elevated from 40 to 42 °C. This means that the temperature range that yielded the highest average cumulative biogas is 38 - 40 °C.

### 5.2.2 Total volume of biogas production

The average total volume of biogas production from the impact of variable operating temperatures is presented in Figure 5.3.



Fig. 5.3. Average total volume of biogas yield at different temperatures

As seen in Figure 5.3, as digester temperature increased from 32 to 39 °C, the average total biogas volume production increased, although not a linear relationship. As the temperature was further elevated from 39 to 40 °C, the production of biogas is relatively constant. However, as the temperature marginally increased beyond 40 °C, the volume of biogas yield is seen to decline. Implying that increase in temperature within the mesophilic range, but not beyond the higher optimal value, can improve biogas yield, but further temperature increase beyond the optimal range can cause a decline in biogas production. In addition, it is seen in Figure 5.3 that the digester temperature rose beyond 40 °C, supporting the reports of previous studies (Babaee et al., 2013; Usman et al., 2012; Chae et al., 2008; El-Mashad et al., 2004). Hence, it is deduced from Figures 5.1, 5.2 and 5.3 that the optimum temperature range for AD of glucose under the mesophilic condition for this research work is 37 - 40 °C.

### 5.2.3 Methane production at varying temperature

The relationship between variable temperatures and methane production is shown Figures 5.4 and 5.5 for average rate and average cumulative methane production, respectively.



Fig. 5.4. Average rate of methane proportion at different temperatures

The rate of methane production presented in Figure 5.4 shows a similar behaviour to that of the biogas production rate in Figure 5.1. However, as seen in Figure 5.4, the 36, 38, 40 and 42 °C digesters began methane production at the third hour, while the setups that operated at 34 and 32 °C, commenced methane production at the fourth and eighth hour, respectively. In addition, the 38, 40 and 42 °C digesters peaked at the ninth hour. The digester that was operated at 40 °C produced the highest peak, followed by the 38 °C digester. The 36 and 34 °C digesters reached their peak at the 12<sup>th</sup> hour, ahead of the 42 °C digester, which peaked at the 13<sup>th</sup> hour. Then the 32 °C digester peaked at the 18<sup>th</sup> hour.

Furthermore, the rate of methane production remained positive for all the digesters up to the 19<sup>th</sup> hour, when the methane production rate from the 42 °C digester became negative. Other digesters, except the 34 °C digester, also produced negative methane production rate but at different times. The negative methane production rate means that the production of methane was on the decline and is lower than the methane production of the previous hour.



Fig. 5.5. Average cumulative methane production at different temperatures

The digester with a temperature of 36 °C provided the third highest average cumulative methane production, then the 42 °C digester. However, the 32 °C digester generated the least average cumulative methane after the 34 °C digester.

### 5.2.4 Average total volume of methane production at different temperatures

Figure 5.6 shows the average total methane production relative to temperature variation.



Fig. 5.6. Average total volume of methane yield at variable temperature

As seen in Figure 5.6, the impact of temperature on the total volume of methane yield is similar to the effect of temperature on the total volume of biogas production in Figure 5.3. It shows that as the operating temperature of the digesters increased from 32 to 39 °C, the total volume of methane production increased, though not a linear relationship. It is also seen that further increase in temperature from 39 to 40 °C, resulted in a relatively constant methane production. However, as the digester operating temperature slightly increased beyond 40 °C, the volume of

methane yield dropped. Indicating that 42 °C is above the temperature limit of which it supports the growth of the methanogenic bacteria.

### 5.3 Effect of pH

This section discussed the results obtained from the investigation conducted on the influence of pH variation on biogas production and methane concentration of the biogas.

### 5.3.1 Rate and cumulative biogas production at various pH values

The result of the experiment to determine the effects of pH variation on biogas and methane production are discussed in this section. The rate of biogas production and the cumulative biogas production are presented in Figures 5.7 and Figures 5.8, respectively.



Fig. 5.7. Average rate of biogas production at different pH values

As seen in Figure 5.7, all the digesters commenced biogas production at the same time, except the digester that was operated at pH of 5.0. The reason for this is that all the digesters were operated at the same temperature of 39 °C. The digester that operated at pH of 5.0 recorded no biogas yield, as seen in both Figures 5.7 and 5.8 because the activity of methanogenic bacteria was halted due to the acidic condition. This finding supports the reports by previous studies, which states that AD failure occurs when the pH inside the digester is below 5.5 (Nayono, 2009; Al Seadi et al., 2008).

In addition, it is seen in Figure 5.7 that the rate of biogas production for pH6 digester is higher than the pH7 and pH8 digesters between the second and third hour. Afterwards, pH7 digester significantly increased biogas production rate to about 80% higher than the pH6 and pH8 digesters. The pH7 digester peaked at the fourth hour and remained at that peak for two hours, and then took a sharp drop between the sixth and seventh hour. Finally, the rate of biogas production for pH7 digester reached a steady state at the 10<sup>th</sup> hour. However, the pH6 and pH8 digesters came to steady state in the 21<sup>st</sup> and 19<sup>th</sup> hour, respectively. This indicates that the organic conversion of substrate to biogas was almost completed at the 10<sup>th</sup>, 21<sup>st</sup> and 19<sup>th</sup> hour for pH7, pH6 and pH8 digesters, respectively. It took the pH7 digester 10 hours to reach a steady state, meaning that the HRT to complete organic conversion process was about half of the time it took pH6 and pH8 digesters. Apart from the pH5 digester that produced no biogas, it is seen that pH6 digester recorded the lowest biogas production rate, followed by the pH8 digester while the highest rate is generated by the digester that was operated at pH of 7.0.

Similarly, the influence of pH on the cumulative biogas production presented in Figure 5.8, shows that the highest cumulative biogas yield is recorded by the pH7 digester, then followed

by the pH8 digester, whereas the pH6 digester produced the lowest biogas production apart from the pH5 digester that recorded no biogas production.



Fig. 5.8. Average cumulative biogas production at different pH values

### 5.3.2 Average total volume of biogas production at different pH values

This section discusses the impact of variable pH values on the total volume of biogas production. As seen in Figure 5.9, the average total volume of biogas yield decreased as pH value decreased as well as increased from the pH of 7. In addition, it is evident in Figure 5.9 that there was no biogas production until about 5.5 pH due to the acidic condition of the digester content, which inhibits methanogenic bacterial activity. This result supports the previous studies that suggest that the ideal pH value for methanogenesis is 7.0 and the optimal pH range is 6.8 - 7.2.



Fig. 5.9. Average total volume of biogas production at different pH values

# 5.3.3 Methane production rate and cumulative methane production at different pH values

The average methane production rate and cumulative methane production are presented in Figure 5.10 and Figure 5.11



Fig. 5.10. Average methane production rate at different pH values

The methane production rate at variable pH as seen in Figure 5.10 is similar to the biogas production rate at variable pH seen in Figure 5.7. All the digesters commenced methane production at the same time, apart from the pH5 digester that produced no methane at all due to the acidic condition. In addition, in Figure 5.10, the rate of methane production for pH7 digester peaked about the sixth hour and then took a sharp drop. Finally, the rate of methane production for pH7 digester reached almost the steady state at the 19<sup>th</sup> hour. However, it is seen that the methane production rate for pH6 and pH8 digesters are close on average.



Fig. 5.11. Average cumulative methane production at different pH values

As seen in Figure 5.11, the average cumulative methane production is highest at digester pH of 7.0. In addition, the cumulative methane production for pH6 digester is slightly higher than the pH8 digester. Whilst the pH5 digester did not produce methane.

5.3.4 Average total volume of methane production at different pH values



Fig. 5.12. Average total volume of methane production at different pH values

Figure 5.12 shows that from pH of 5.0 to 5.5 no methane was produced, but as digester pH increased from 5.5 to 6.8, the average total volume of methane production increased. However, methane production is seen to be relatively stable between pH 6.8 and 7.2. As digester pH exceeded 7.2 the average total volume of methane production is seen to decline, which supports the suggestion that the optimum mesophilic pH range for methanogenesis is 6.8 - 7.2.

It is not that no gas was produced at pH of 5.0, rather, there is no detection of methane in the generated gas due to the inhibition of methanogenic bacteria by the acidic condition in the digester. No further evaluation was carried out in the produced gas in order to determine its composition. Since there is no methane detection on the gas produced at this pH, no biogas is accounted for, as biogas is primarily the combination of methane and carbon dioxide. Hence, the absence of methane production in AD process means that no biogas is produced.

### **5.4 Influence of digester pressure**

This section discusses the experimental results from the influence of digester pressure variation on biogas and methane production.

### 5.4.1 Rate and cumulative biogas production at different digester pressures

The average biogas production rate and average cumulative biogas production under different digester pressures are presented in Figure 5.13 and 5.14, respectively.



Fig. 5.13. Average rate of biogas production at different pressures

As seen in Figure 5.13, the rate of biogas production did not follow a particular pattern with respect to the varied operating pressures. It is difficult to explain why there is no particular response of biogas production rate to the increase or decrease in digester pressure. This result is in agreement with the work of Japan-Agency for Marine-Earth Science and Technology (2007), which highlights the difficulty in explaining the influence of pressure on complex metabolic systems, which includes the AD process.


Fig. 5.14. Average cumulative biogas production at different pressure values

It is seen in Figure 5.14 that there is a marginal difference in the cumulative biogas yield of the different samples at as the digester operating pressure increases. This might be because of the small difference in the operating pressure. However, the studies by Chen et al. (2014), Lindeboom et al. (2011) and Hayes et al. (1990) shows that there is a decrease in biogas production as operating pressure increases. This is because part of the generated  $CO_2$  in biogas liquefies with an increase in digester operating pressure, thereby reducing the overall biogas production while increasing the concentration of methane in the biogas. The phenomenon is based on gas solubility; the  $CO_2$  being more soluble in water than methane liquefies at the same temperature and pressure (Yalkowsky et al., 2010), resulting in more concentration of methane in the biogas.

### 5.4.2 Average total volume of biogas production at variable digester pressure

The average total volume of biogas production at variable digester pressure is presented in Figure 5.15.



Fig. 5.15. Average total biogas production at different pressure values

As seen in Figure 5.15, the total volume of biogas yield marginally increases as digester pressure decreases from 0.7 to 0.1 bar. In Figure 5.15, it is seen that the 0.1 bar digester generated the highest total biogas production, followed by the 0.3 bar digester while the 0.7 bar digester yielded the lowest biogas behind the 0.5 bar digester. The reason for this is also based on the solubility of gases.

# 5.4.3 Methane production rate and cumulative methane yield at different digester pressure

The relationship between pressure and methane production rate and cumulative methane production are presented in Figure 5.16 and Figure 5.17, respectively.



Figure 5.16 shows a similar behaviour as the biogas production rate at various digester pressure, with no clear pattern. This is also attributed to the difficulty in describing the effects of pressure on the complex metabolic process.



Fig. 5.17. Average cumulative methane production at variable pressure

As seen in Figure 5.17, the variation of digester pressure marginally influenced the amount of methane production. This is similar to the behaviour of biogas to the given operating pressures. The small difference in the digester operating pressures utilised in this study may have contributed in the reason why there is no significant change in cumulative methane production with an increase in the digester operating pressure.

# 5.4.4 Average methane concentration at different digester pressures

The influence of variable digester pressure on the methane proportion is seen in Figure 5.18.



Fig. 5.18. Average methane proportion at different pressures

As seen in Figure 5.18, an increase in digester pressure resulted in an increase in methane proportion. Though the increment is marginal due to the small increase in the digester operating pressure. As the digester pressure increases from 0.1 to 0.7 bar, the methane proportion is found

to increase. The significant of this result highlights the importance of digester pressure in AD, with regards to improving the methane concentration in biogas.

#### 5.4.5 Average total methane production at different digester pressures

The influence of variable digester pressure on methane production is seen in Figure 5.19.



Fig. 5.19. Average total methane production different pressures

As seen in Figure 5.19, methane production increases as digester pressure increase from 0.1 to 0.7 bar. Though the increment is marginal due to the small increase in the digester operating pressure. The significant of this result highlights the importance of digester pressure in AD, with regards to improving methane production.

# 5.5 Effect of mixing speed

This section discusses the impact of various mixing speed on biogas and methane production.

#### 5.5.1 Average rate and cumulative biogas production at various mixing speeds

The average biogas production rate and cumulative biogas production relative to different mixing speeds are presented in Figures 5.20 and 5.21, respectively.



Fig. 5.20. Average rate of biogas production at various mixing speed

As seen in Figure 5.20, the 60 and 300 rpm digesters maintained a similar rate of biogas production throughout the digestion process. In addition, the 60 and 300 rpm digesters produced the highest peak between the eighth and the .11<sup>th</sup> hour of retention time. It is seen in Figure 5. 20 that the rate of biogas production for 600 rpm digester is higher than the 60, 300, and 2000 rpm digesters between the 12<sup>th</sup> and 20<sup>th</sup> hour. Furthermore, the plot shows the continuous decline in biogas production from the 2000 rpm digester for two hours, between the 16<sup>th</sup> and 18<sup>th</sup>, which is indicated by the negative biogas production rate. However, the biogas production picked up again for another two hours, before reaching the steady state at the 20<sup>th</sup> hour, signifying that no more increase in biogas production by the 2000 rpm digester.



Fig.5.21. Average cumulative biogas production at different mixing speed

Similarly, in Figure 5.21 the 60, 300 and 600 rpm digesters maintained parallel biogas production pattern between the third and 10<sup>th</sup> hour while the 2000 rpm digester produced little or no biogas before the fifth hour. The 2000 rpm digester improved between the fifth and 15<sup>th</sup> hour, where the biogas production peaked between the 14<sup>th</sup> and 15<sup>th</sup> hour. It then gradually declined before reaching the steady state at the 20<sup>th</sup> hour. The cumulative biogas production by the 600 rpm digester is lower than the 60 and 300 rpm digesters and reached its biogas production peak at about the 20<sup>th</sup> hour before it the gradual decline. Contrarily, the cumulative gas generated by the 300 rpm digester is the highest for all the samples, but marginally higher than the 60 rpm digester. Making 300 rpm the optimal mixing speed relative to the other mixing speed investigated and for the quantity of the samples utilised in this study. It means that there is a marginal increase in biogas production as the mixing speed increased from 60 to 300 rpm. However, a further increase in the mixing speed to 600 rpm shows a decrease in the biogas production. In addition, as the mixing speed is further increased to 2000 rpm, there is even

much more reduction in biogas production relative to the biogas produced by the 60 and 300 rpm digesters.

### 5.5.2 Average total volume of biogas production at variable mixing speeds

The bar chart in Figure 5.22 presents the average total volume of biogas production at different operating mixing speeds.



Fig. 5.22. Average total volume of biogas production at different mixing speeds

The average total volume of biogas production at different mixing speeds as seen in Figure 5.22, shows that the digester that was operated at 300 rpm produced the highest volume of biogas, followed by the 60 rpm digester. However, the 2000 rpm digester recorded the lowest total volume of biogas production, behind the 600 rpm digester. For this study, the optimal mixing speed range is 150 - 300 rpm. Further study is required in this area, with the goal of investigating the relationship between various mixing speeds and a specific quantity and total solids of substrates.

#### 5.5.3 Average rate and cumulative methane production at various mixing speeds

The average biogas production rate and cumulative methane production relative to different mixing speeds are presented in Figures 5.23 and 5.24, respectively.



Fig. 5.23. Average rate of methane production at various mixing speeds

As seen in Figure 5.23, the 60 and 300 rpm digesters maintained nearly a similar pattern throughout the digestion process. In addition, the 60 and 300 rpm digesters produced the highest peak at the fourth hour. It is seen in Figure 5.23 that while the rate of methane production for the 60, 300 and 600 rpm digesters was declining, the methane production rate for the 2000 rpm digesters was increasing just after the fourth hour. Furthermore, the plot shows that the methane production rate for the 2000 rpm digesters was negative from the 16<sup>th</sup> hour, indicating the continuous decline in methane production from the 2000 rpm digester.

The influence of variable digester mixing speed on the cumulative methane production is presented in Figure 5.24.



Fig. 5.24. Average cumulative methane production at different mixing speeds

As seen in Figure 5.24, the cumulative methane production to the different mixing speeds is similar to the cumulative biogas production was seen in Figure 5.21. This is because the methanogenic bacterial activity is responsible for the behaviour of the methane and biogas production.

# 5.5.4 Average total volume of methane production at variable mixing speeds

The average total volume of methane produced at various mixing speeds is presented in Figure 5.25. It shows a similar pattern as the average total volume of biogas production see in Figure 5.22. The 300 rpm digester produced the highest methane, followed by the 60 rpm digester, whereas, the 2000 rpm produced the lowest average total volume of methane. In addition, as the mixing speed increased from 300 rpm, the methane production is seen to decline, indicating that the optimal mixing speed is exceeded.



Fig.5.25. Average total volume of methane production at different mixing speeds

# 5.6 Influence of simultaneous manipulation of multiple inputs on biogas and methane production

The experiment is conducted to explore the significance of simultaneous manipulation of four operating parameters. The operating parameters are temperature, pH, mixing speed and pressure. The operating parameters were simultaneously altered at three different intervals of eight hours each. The data obtained from the experiment is used to test the black-box developed in this research study.

### 5.6.1 Biogas and methane production

The initial value for temperature, pH, mixing speed and pressure for the experiment are set at 35 °C, 7.6, 700 rpm and 0.3 bar. This is followed by another parameter adjustment after eight hours to 37 °C, 5.4, 500 rpm and 0.4 bar, and the final amendment made to temperature, pH, mixing speed and pressure are 41 °C, 6.8, 120 rpm and 0.5 bar, respectively. The impacts of

the simultaneous alteration of the operating parameters on biogas and methane production are presented in the following plots.

# 5.6.1a Average rate and average cumulative biogas production

The effect of this experimental design on biogas production rate is presented in Figure 5.26.



Fig. 5.26. Average rate of biogas production at simultaneous variance of input parameters

As seen in Figure 5.26, the rate of biogas production gradually increased from the fourth hour and peaked after two hours at the sixth hour. It then declines and becomes negative, indicating a continuous decline in biogas production that is lower than the biogas production of the biogas production recorded in the previous hour. Furthermore, the rate of production picked up again but was still on the negative axis for another 4 hours before biogas production ceased and did not resume until the end of the experiment. The reason why biogas production was halted from the 13<sup>th</sup> hour was due to the reduction of the digester pH to 5.4, at which the digester content becomes acidic and inhibited the activity of the methanogenic bacteria, consequently halting

methane production that combines with  $CO_2$  to forms biogas. Despite increasing the pH to 6.8 at the 17<sup>th</sup> hour, the system did not recover for the rest of the experiment as seen in Figure 5.26. Indicating that pH inhibition of methanogenic bacteria is difficult to reverse. This result supports the study of Kangle et al. (2012).

The average cumulative biogas production is presented in Figure 5.27.



parameters

As seen in Figure 5.27, the average cumulative biogas production peaked at the eighth hour. It then decreased to the 13<sup>th</sup> hour at which biogas production is halted. The reason for the termination of biogas production is similar to the one given for the biogas production rate in.

# 5.6.1b Average rate and average cumulative methane production for Experiment31

The influence of the adjustment of the operating parameters on methane production is presented and discussed in this section. In Figures 5.28, the average methane production rate is presented.



Fig. 5.28. Average rate of methane production at simultaneous adjustment of input parameters

The response of methane to the simultaneous adjustment of the operating parameters is similar to that of biogas pattern is seen in Figure 5.26. The rate of methane production as seen in Figure 5.28 reached the peak at the sixth hour. It then decreased to the minimum on the negative axis at the tenth hour. Then the rate of methane production increased again up to the 12<sup>th</sup> hour when it halted and did not recover. This is due to the acid condition caused by the adjustment of digester pH to 5.4, which inhibited the methanogenic bacteria.

Furthermore, the average cumulative methane production is presented in Figure 5.29.



Fig. 5.29. Average cumulative methane production at simultaneous adjustment of the operating parameters

Similarly, the average cumulative methane production as seen in Figure 5.29 increased rapidly and peaked at the seventh hour as observed in Figure 5.27. It then dropped to the zero level at the 13<sup>th</sup> hour when methane production ceased, because of the same reason stated for the biogas production rate.

# 5.7 Influence of co-digestion on biogas and methane production

This section discusses the significance of the results obtained from the investigations carried out to determine the effect of co-digestion on biogas and methane production. Samples of Food waste (FW) and Wastewater sludge (WWS) are co-digested in the following ratio: FW only represented as (S1); WWS only (S2); 0.3FW+0.7WWS (S3); 0.5FW+0.5WWS (S4) and 0.7FW+0.3WWS (S5). The results obtained are presented in the following plots.

# 5.7.1 Average rate of biogas production at different co-digestion ratio

The biogas production rate is presented in Figure 5.30.



Fig. 5.30. Average rate of biogas production at different co-substrate ratio

As seen in Figure 5.30, the average rate of biogas production from co-digestion of substrates at different ratios followed nearly a similar pattern. This indicates that the response of the anaerobic bacteria to the operating parameters of all the samples is similar. The biogas production rate peaked on the fifth day for all the samples and the decreased. However, the S4 sample recorded the highest biogas production rate, followed by the S5 digester and the S3 digester. Whilst the S1 sample generated the lowest biogas production rate, followed by the S2 sample, which is the single-substrate samples. The result shows that co-substrates samples performed better than the single-substrate to biogas was nearly completed, thereby resulting in a continuous decline in the rate of biogas production.



Fig. 5.31. Average cumulative biogas production at different co-substrate ratio

Similarly, Figure 5.31 shows that S4 digester produced the highest cumulative biogas followed by the S5 digester and then again, the S3 digester. Furthermore, it was observed that the digestion of WWS as a single substrate generated the lowest average cumulative biogas while the second lowest is produced by the digestion of FW only substrate, that is, SI and S2, respectively. Implying that co-digestion of FW and SSW improved the digestion process, as suppose to the digestion of FW and WWS as single substrates. In addition, S4 was found to be the best co-substrate ratio for co-digestion of FW and WWS for improved biogas production for this study.

# 5.7.2 Average methane production rate at different co-substrate ratio

As seen in Figure 5.32, the average rate of methane production from co-digestion of substrates at different ratios followed nearly a similar pattern. This indicates that the response of the anaerobic bacteria to the operating parameters in all the samples is similar. The methane production rate peaked at the fifth day for all the samples and the decreased. However, the S4

sample recorded the highest methane production rate, followed by the S5 digester and the S3 digester. Whilst the S2 sample generated the lowest biogas production rate, followed by the S1 sample, which is the single-substrate samples. The result shows that co-substrates samples performed better than the single-substrate samples. The negative rate of methane production indicates that the conversion of substrate to methane was nearly completed, thereby resulting in a continuous decline in the rate of methane production.



Fig. 5.32. Average methane production rate at various substrate ratio

#### 5.7.3 Average cumulative methane production at different co-substrate ratio

The average cumulative methane production as seen in Figure 5.33, shows that all the samples recorded their individual highest methane production on the fifth day. It also indicated that the co-substrate digesters performed better than the single-substrate digesters. Similarly, figure 5.33 shows that S4 digester produced the highest cumulative methane, which implies that S4 is the best co-substrate ratio for the co-digestion of FW and WWS for this study. This study has shown that co-substrate digestion offers better digester performance than the single-

substrate digestion, which supports the work previously conducted on co-digestion (Wu, 2007; Mata-Alvarez et al., 2000).



Fig. 5.33. Average cumulative methane production at different co-substrate ratio

# 5.7.4 Observed process characteristics

The observation of the responses of biogas and methane production to variable temperature, pH, mixing speed and pressure plotted and discussed in this chapter indicate that mesophilic AD process is a nonlinear dynamic system. The properties of the nonlinear system include the following.

1. The mesophilic AD process does not satisfy both superposition and homogeneity principles.

**Superposition** - For a system *S* to satisfy superposition principle (Cuff, 2012):

If  $y_1 = Sx_1$  and  $y_2 = Sx_2$ , then  $y_1 + y_2 = S(x_1 + x_2)$ 

**Homogeneity** - For a system *S* to satisfy homogeneity principle (Cuff, 2012):

If  $\mathbf{y} = S\mathbf{x}$ , and k is a constant then  $\mathbf{yk} = S(\mathbf{xk})$ 

2. The system has multiple equilibrium points. The curves of the plots in this chapter signify the multiple equilibrium points. The equilibrium point is a point  $x_0$  in the state space of autonomous system  $\dot{x} = f(x)$ , when the state x reaches  $x_0$  and stays at  $x_0$  for all times (Canon, 2016). The equilibrium points for a nonlinear system is the solution to the equation:

$$f(x_n)=0$$

Where, n nonlinear equations in n unknowns have to be solved and the solution is between 0 and infinity.

3. The pattern of some plots of the system demonstrated randomness irrespective of the deterministic nature of the system.

# 5.8 Summary

This chapter has presented the results of the experimentations conducted in this research work. It is found that the biogas production is influenced by the variation in the operating temperature, pH, mixing speed and pressure. The outcome of this study has also shown that for improved biogas production, all the operating parameters need to be maintained at their individual optimum range. In addition, the behaviour of the system under consideration shows that it is a nonlinear process. Furthermore, this chapter highlighted the importance of co-digestion of substrates as well as the importance of the appropriate ratio of the co-substrates in achieving improved biogas production. The next chapter presents the modelling of nonlinear multiparameter models for mesophilic AD process, which is based on the response of biogas and methane production to the overall impact of the operating parameters considered in this study.

# Chapter 6 Black-Box Model Construction

# **6.1 Introduction**

Chapters 4 and 5 discussed the method, generation and analysis of the data from the experimental investigation of the influence of temperature, pH, mixing speed and pressure on biogas and methane production. More specific, Chapter 4 discussed the experimental study to determine the effects of the multi-parameter input on biogas and methane production, while Chapter 5 present an analysis of the implications of these effects based on the data obtained from the experimentations - how the input variables influenced the production of biogas and methane. However, the discussion in this chapter centres on the construction of multivariable nonlinear mathematical models for mesophilic AD process from experimental measurement considering a multiple input single output (MISO) configuration. The purpose of the models is to predict the production of biogas and methane under mesophilic conditions. The models in this thesis are constructed using System Identification method. Figure 6.1 shows the block diagram of the model identification process.



Fig. 6.1. Block diagram for model identification (Schoukens et al., 2012)

# 6.2 System Identification modelling method

System Identification is a method of estimating mathematical models of a dynamic system from measured input-output data (Schoukens et al., 2012). It is implemented under MATLAB functions (Ljung, 2015; Ljung, 2010). The procedure of estimating a model of a dynamic system from measured input-output data (based on the System Identification algorithms) consists of three main phases (Ljung, 2015).

- Design of the experimental setup and data collection,
- Selection of the most appropriate model structure, and
- Identification of method to select a particular model in the set based on the information in the input-output data

In more detail, System Identification modelling method is composed of a number of steps as shown in the flowchart in Figure 6.2a.



Fig. 6.2a. Flowchart of steps in System Identification modelling method

- I. A Good design of experimental setup and data collection ensures that the relevant variables are measured and collected with sufficient accuracy in order to reflect the behaviour of the dynamic system under consideration. The input-output data from the dynamic system, which is acquired through experimentation, is used by System Identification Toolbox to estimate the values of adjustable parameters in a given model structure (Ljung, 2015). The quality of a model depends on the how well the data collected from experimental setup reflects the behaviour of the dynamic system (Ljung, 2015).
- II. The data organisation involves splitting of sample data into two parts; one part is used for model estimation and the other part is used for model validation purpose (Ljung, 2015). The bigger the size of the data set for model estimation, the more information is obtained about the system and the better the model fit the data (Tangirala, 2015; Villanverde and Banga, 2013). Data organisation also include the separation of output vector and sampling time from the estimation and validation data sets, which are the information required for the construction of discrete-time models (Ljung, 2015). Furthermore, the organisation of data also includes the conversion of estimation and validation data sets into iddata objects, which is the representation of time-domain data and the required data structures by MATLAB for System Identification (Ljung, 2015)
- III. The model structure is the mathematical relationship between input and output variables, with unknown parameters (Ljung, 2015). System Identification method requires the selection of model structure in order to determine the numerical values of the parameters of the model (Ljung, 2015).

- IV. Model order is an important component in System Identification modelling method, which is represented by one or more numbers whose definition is a function of the model structure (Ljung, 2015; Pillonetto et al., 2013). The model order represents the set of a number of poles and number zeros, as well as delays for defining the type of model structured used for model identification (Ljung, 2015; Sugiki, 2014).
- V. In model estimation, System Identification Toolbox uses estimation algorithms and the selected model structure to determine the numerical values of model parameters (Ljung, 2015).
- VI. Model validation is used to determine how close the model is to the actual system (mesophilic AD process). This involves comparing the simulated output of the model and output of the measured data (Ljung, 2015; Proctor et al. 2013; Rahiman et al., 2009).
- VII. Model evaluation is used to analyse the suitability of the estimated model with respect to how close the simulated output of the model compares with output of measured data of the actual system for a specific stimulus (Vazquez-Cruz et al., 2014). If the evaluation shows that the model output compares "reasonably" (accuracy evaluation) with the measurement of the actual system, based on the desired accuracy figure, then the modelling process is complete. However, if the results of the model evaluation are not satisfactory, then it is necessary to reconsider the previous steps of the model construction as seen in Figure 6.2a.

It is essential to consider the various levels of accuracy of data and information utilised in constructing the models in this study, especially when evaluating the quality of the models, which are affected by:

- The assumptions based on views from general knowledge and discussions with experts in the field;
- Knowledge of the stoichiometry and analytical relationships guiding the processes;
- The relationships between input and output variables being experimentally determined and presented in the form of data; and
- Statistical correlations that might exist

# 6.2.1 Choice of a suitable mathematical model

The choice of a suitable mathematical model is an important prerequisite to constructing a satisfactory model of a bioprocess. In section 3.2, the two main types of mathematical model (white-box and black-box models) were discussed. White-box models require an in-depth knowledge of the system under consideration in order to represent the system in a sufficiently detailed manner to reproduce the process behaviour (Krishna, 2010); however, the knowledge about AD process is limited. This is due to the poor understanding of the bacterial community responsible for AD process, particularly with respect to the composition of the bacterial community and the bacterial response to different substrates and operations (Li, 2013; Riviere et al., 2009).

Monod kinetic white-box model is inadequate to describe AD process because of the presence of a single set of kinetic parameters involved (Vázquez-Cruz et al, 2014). Again, white-box models such as ADM1 that are based on the first principal is computationally intensive to fit because of the requirement for a large number of kinetic parameters implemented in the form of DAE and DE set (Yu et al., 2013).

Furthermore, the financial and time constraint, which limits the construction of white-box models, especially when dealing with a complex process, such as AD, whose process knowledge is vague. According to Heams (2006), white-box models have little or no significance in AD because of their inadequacy to consider all the stages of the anaerobic process, operating parameters and history of operational data. Another drawback of white-box models is the lack of consideration for the actual physico-environmental operating parameters that influence anaerobic bacterial activity. Leading to the inability of the white-box model to describe AD process when physico-environmental parameters need to be considered for a more realistic study of the operation. Due to these limitations, it may not be practically feasible to apply white-box models to predict the production of biogas and methane from measured input-output data for this study.

However, black-box models are based on the availability of input-output data for model identification (Vázquez-Cruz et al, 2014). They require no in-depth knowledge of the system under consideration; thereby can be applied when there is little or no knowledge about the process. Though the parameters of the functions of black-box models lack physical significance, yet black-box models can adequately represent the trend in system behaviour even in a complex process where it is not possible to solve the resultant equations (Krishna, 2010). In addition, black-box models can adequately predict biogas and methane production from input-output data. Therefore, black-box models are considered for the construction of the models for mesophilic AD process from measured MISO data for this study.

## 6.2.2 Black-box model identification procedure

Given an overview of how the System Identification Toolbox "works" to construct models based on input-output data sets as seen in Figure 6.2a. The flowchart in Figure 6.2b shows the actual procedure followed in this study to estimate the multi-input-single-output (MISO) nonlinear black-box models from experimental data, which are discussed in more specific details as applied to this research work in the following steps. The models are estimated in MATLAB command line, and the execution codes for various processes are documented in Appendix B.





Fig. 6.2b. Flowchart of the actual black-box model identification procedure

### Step 1: Design of experimental setup and data collection

The design of experimental setup for this research study, which is discussed in Chapter 4, is utilised to investigate the influence of multi-input parameters on biogas and methane production. The data set utilised for the model identification, which is in time-domain, contains 624 data samples, collected from 26 experiments with a sample time of 1 hour for a 24-hour duration, as described in Chapter 5. The inputs u(t), are composed of 26 variables: eight variables of temperature (in °C), six variables of pH, six variables of mixing speed (in rpm) and six variables of pressure (in bar). The output vector y(t) contains a single variable: biogas or methane.

#### **Step 2: Data organisation**

The input-output data collected from the experiments are loaded into MATLAB Workspace. The input data sets contain a 4-by-25 matrix for each set and the output data is a vector. The output vector, as well as the time vector, are extracted from the 26 experiments in MATLAB command line. An iddata object is created for each of the 26 experimental data sets with the following syntax (Ljung, 2015).

 $iddata_object = iddata(y, u, Ts)$ 

This creates an iddata object containing a time-domain output variable *y*, and input variable *u*. Where Ts specifies the sampling time of the experimental data (Ljung, 2015). An iddata object is a basic object and the required data structure by MATLAB for handling input-output variables in System Identification Toolbox (Ljung, 2015).

#### Step 3: Data quality analysis and data selection

The quality of the data collected from the experiments is analysed using the input-output data plot shown in Figure 6.3 and Figure 6.4 for biogas and methane production, respectively.



Fig. 6.3. Input-output data plot for biogas production

The plot seen in Figure 6.3 is used to inspect and clean data. The plot shows that the output data generated from the experiment with pH of 5 is zeros all through due to the none production of methane, which is the major constituent of biogas, resulting from the acid state of the digester sludge. Therefore, the data from the experiment with pH of 5 is dropped, leaving the total experimental data used for the estimation and validation of the nonlinear black-box models to 25.

Similarly, the plot in Figure 6.4 is used to inspect and clean the input-output data for methane production.



Fig. 6.4. Input-output data plot for methane production

The plot in Figure 6.4 shows that the output data generated from the experiment with pH of 5 is zeros all through due to non-production of methane at the acid state of digester sludge. Therefore, the data from the experiment with pH of 5 is also dropped, leaving the total experimental data used for the estimation and validation of the nonlinear black-box models to 25.

The 25 iddata objects from the 25 experiments for both biogas and methane production are split into two unequal subsets, 17 iddata objects are used for model estimation and eight iddata objects are used for model validation. All the iddata objects for model estimation are merged into a single iddata object; de for biogas production and ze for methane production. Similarly, all the iddata objects for model validation are merged into one iddata object, dv for biogas production and zv for methane production, with the following syntax (Ljung, 2015).

de or dv = merge (iddata1, iddata2 .... IddataN)

The merged estimation and validation iddata objects contain individual iddata objects that represent all the four input parameters.

# **Step 4: Model structure selection**

The choice of a suitable model structure is based on the understanding of the System Identification method as well as the dynamic systems under consideration (Murray-Smith, 2015). As discussed in section 6.2.1, mesophilic AD process is a nonlinear system, which can be adequately represented by black-box models. There are a variety of nonlinear black-box model structures that are available for modelling mesophilic AD process, such as NARX, NOE, NARMAX, NLHW and NBJ, discussed in section 3.2.1. However, NARX and NLHW model structures are often used for nonlinear black-box model identification (Rankovi'c et al., 2012). The NARX and NLHW are the two model structures used to estimate the models for biogas and methane production for this research work. The model structures have different nonlinearity estimators that are specific to the selected model structures.

The output plot for NARX and NLHW models do not show a good Fit with the incorrect complexity of the nonlinearity estimators (Alexandridis and Zapranis, 2014; Zhang, 1997, cited in Matlab 2016a). The complexity of the nonlinearity estimators is specified using the NumberOfUnits (number of units in nonlinear estimator) (Alexandridis and Zapranis, 2014; Zhang, 1997, cited in Matlab 2016a). This is a positive integer that can be chosen automatically when the number of units is determined from estimation data. Or interactively, when the number of units is determined during model estimation (Alexandridis and Zapranis, 2014;

Zhang, 1997, cited in Matlab 2016a). A higher number of units signifies a more complex nonlinearity estimator. A nonlinear estimator with a fewer number of units will result in underfitted model (Alexandridis and Zapranis, 2014). However, specifying more number of units will lead to overfitted model (Alexandridis and Zapranis, 2014).

Selecting the appropriate nonlinearity estimator complexity requires validating a low complex model first and then progressively increase the complexity and validate accordingly (Zhang, 1997, cited in Matlab 2016a). The quality of the models reduces as the nonlinearity estimator becomes too complex (Zhang, 1997, cited in Matlab 2016a). The reduction in quality of the performance of the models is only seen if independent estimation and validation data sets are used (Alexandridis and Zapranis, 2014; Zhang, 1997, cited in Matlab 2016a). Therefore, it is essential to select the suitable number of units in nonlinearity estimators for a particular model estimation. According to Alexandridis and Zapranis (2014), the simplest way to select the optimal number of units is by trial and error.

#### **Step 5: Model orders selection**

Model order is used for estimating nonlinear black-box models (Ljung, 2015). It is a set of positive integers that represent the poles, zeros and delays used for defining the regressor configuration for NARX models, and the linear subsystem transfer function for NLHW models (Ljung, 2015; Sugiki, 2014).

The model order for NARX models are in the form [na nb nk], where na signifies the number of past output terms used to predict the current output, nb represents the number of past input terms used to predict the current output, and nk specifies the delay from input to the output in terms of the number of samples (Ljung, 2015).

Similarly, the model order for NLHW models are in the form [na nf nk], where na signifies the number of past output terms used to predict the current output, nf specifies the number of poles of the linear transfer function, and nk specifies the delay from input to the output in terms of the number of samples (Ljung, 2015).

A poor model Fit can be as a result of the incorrect model order. Ideally, the lowest-order model that adequately describes the system dynamics well is preferred (Ljung, 2015; Sugiki, 2014). If the model order is too low, it can result in model underfitting. Similarly, If a model order is too high it can cause a model to be overfitted (Ljung, 2015; Sugiki, 2014). Both underfitting and overfitting are undesirable in a model, however, overfitting has more impact than underfitting in that it causes the model to fit the noise, instead of the actual behaviour of the system under consideration. The selection of model orders is by trial and error (Ljung, 2015; Sugiki, 2014).

The remaining steps in model identification, including model estimation, model validation and model evaluation are discussed more in the following sections as necessary due to the description being specific to the selected model structures.

#### 6.2.3 NARX model estimation

The NARX model is characterised by regressors and a nonlinear estimator (Ljung, 2015; Shariff et al., 2014). The nonlinear estimator is composed of a parallel combination of nonlinear and linear functions. The regressors are described as delayed inputs and outputs from the nonlinearity identification, which are inputs to the nonlinear estimator functions, as shown in Figure 6.5 (Ljung, 2015; Shariff et al., 2014).



Fig.6.5. Block diagram of NARX models (Ljung, 2015; Shariff et al., 2014)

The NARX model is described by the following equation (Shariff et al., 2014).

$$y(t) = f[y(t-1), \dots, y(t-n_y), \dots, u(t), u(t-1), u(t-n_u)] + \varepsilon(t)$$
(6.1)

Where,

y(t) is the output u(t) is the input  $\varepsilon(t)$  is the noise signals  $n_y$  is the maximum lags of past output  $n_u$  is the maximum lags of past inputs f is the nonlinear function

u(t - 1) and y(t - 1) are the regressors

# Step 6a: NARX model estimation for biogas and methane production

Two important elements, model order and nonlinearity estimators, which are required for the estimation of black-box models, are chosen by trial and error. The NARX models are estimated in MATLAB command line with the following syntax.

M = narx (iddata, model order, nonlinearity estimator),
where M is the NARX models.

The Fit (in %) of the estimated model is the mean square error (MSE) between the measured data and the simulated output of the model (Ljung, 2015). A model of perfect fit (no error), means that the Fit is 100 %. Similarly, a model that is not able to describe any of the variation of the output and only the mean level corresponds to a Fit of 0% (Ljung, 2015).

To get a measure of how good the Fit of the estimated model to the measured data is, the simulated output of the model is compared to the output of the measured data (Ljung, 2008; Ljung, 2015). This approach is used to validate the estimated models by checking how well the simulated output of the models fits the measured output (Ljung, 2008; Ljung, 2015). The model output plot comparing the simulated output of the models and the measured output, the percentage of the output the model reproduces, which is the 'Best Fit', is computed using the following equation (Ljung, 2008):

Best Fit = 
$$\left(1 - \frac{|y - \hat{y}|}{|y - \bar{y}|}\right) \ge 100$$
 (6.3)

Where, y is the measured output,  $\hat{y}$  is the simulated output, and  $\bar{y}$  is the mean of y. The Fit ranges from 100%, which indicates a perfect Fit, to 0% that corresponds to a very bad fit. The measured data is reffered to as the validation data.

Comparing the quality of different estimated models, a model is considered to describe the data well when the Fit to the validation data is high. If an estimated model has a higher Fit to the validation data than another model, then the first model is considered to describe the real system better. The comparison between the simulated output of the model and the output in the validation data is computed by the 'compare' command in MATLAB with the following syntax (Ljung, 2015).

Compare(M), where M is the estimated model

As seen in the plots in Figure 6.3, the responses of biogas production to the variables of Temperature, pH, Mixing speed and Pressure, indicate that mesophilic AD system is in the region of the third order. The first model order chosen for the estimation of NARX models for this study is [ones(1,1), ones(1,4), ones(1,4)], which is equivalent to [1 1 1] = [na nb nk]. It means that the output variable is predicted by the output and all the four input variables, and it is being delayed by one sample. With the MATLAB default 'wavelet' nonlinearity estimator, the function nlarx, which is used to facilitate the estimation of NARX models (Shariff et al., 2014), and the model order, the NARX model mx1 is estimated.

Similarly, five more models are estimated using the following five different model order:

[ones(1,1), ones(1,4), 2\*ones(1,4)] for model mx2;

[2\*ones(1,1), ones(1,4), 2\*ones(1,4)] for model mx3;

[2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)] for model mx4;

[3\*ones(1,1), ones(1,4), ones(1,4)] for model mx5; and

[3\*ones(1,1), 3\*ones(1,4), ones(1,4)] for model mx6.

The six estimated models are evaluated by comparing their simulated output and the output in the validation data, dv, as shown in Figure 6.6.



Fig. 6.6. Evaluation of estimated NARX models with different model moder for biogas production

As seen in Figure 6.6, the result of the evaluation shows that the best model is mx3 as the model has a higher Fit (79.63%) to the validation data than other estimated models. For the estimated model mx3, the number of units in 'wavelet' is automatically chosen by the estimation algorithm, which is 7. In order to explicitly specified the number of units in the 'wavelet', 7 is replaced with 6, 8 and 9, and are used to estimate models mx7, mx8, and mx9, respectively, with the same model order [2\*ones(1,1), ones(1,4), 2\*ones(1,4)] used to estimate model mx3. The models are evaluated with the validation data as shown in Figure 6.7.



Fig. 6.7. Evaluation of estimated NARX models with different number of units in the 'wavelet' estimator for biogas production

The result of the evaluation as seen in Figure 6.7 shows that there is about 5% improvement of the previously estimated model mx3 when the number of units in the 'wavelet' is increased from 7 to 9. This indicates that model mx9 is a better model than models mx3, mx7 and mx8, and the best performing NARX model with 'wavelet' estimator for biogas production for this study.

Apart from 'wavelet' estimator two other nonlinearity estimators 'treepartition' and 'sigmoidnet' are used to estimate two new NARX models, mx10 and mx11, respectively. The models are estimated with the estimation data (de), model order [2\*ones(1,1), ones(1,4), 2\*ones(1,4)] and their respective nonlinearity estimator. The estimated NARX models mx9, mx10 and mx11 are evaluated with the validation data as presented in Figure 6.8.



Fig. 6.8. Evaluation of NARX models with different nonlinearity estimators for biogas production

As seen in Figure 6.7, the evaluation shows that the model mx9 with 'wavelet' nonlinearity estimator performed better than the models mx10 and mx11 with 'treepartition' and 'sigmoidnet' nonlinearity estimators, respectively. Therefore, the estimated model mx9 is considered the best NARX model that describes the dynamics of mesophilic AD for biogas production for this research work.

Similarly, estimating the NARX model for methane production, involed all the steps followed to estimate the NARX model for biogas production. The data acquired from methane production, a constituent of biogas production, is split into two subsets, 17 data sets for the estimation data (ze) and eight data sets for the validation data (zv). Six different NARX models are estimated with 'wavelet' nonlinearity estimators and the following model order:

[ones(1,1), ones(1,4), ones(1,4)] for model ms1;

[ones(1,1), ones(1,4), 2\*ones(1,4)] for model ms2;

[2\*ones(1,1), ones(1,4), 2\*ones(1,4)] for model ms3;

[2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)] for model ms4;

[3\*ones(1,1), ones(1,4), ones(1,4)] for model ms5; and

[3\*ones(1,1), 3\*ones(1,4), ones(1,4)] for model ms6.

The estimated models are evaluated by comparing their simulated output and the output in the validation data, zv, as shown in Figure 6.9.



Fig. 6.9. Evaluation of NARX models with different model order for methane production

As seen in Figure 6.9, the result of the evaluation shows that the best model is ms4 as the model has a higher Fit (78.38%) to the validation data than other estimated models. The number of units in the 'wavelet' estimator is automatically chosen by the estimation algorithm as 10 for model ms4. In order to explicitly specified the number of units in 'wavelet', 10 is replaced with 9, 11 and 12, and are used to estimate models ms7, ms8, and ms9, respectively, with the same model order [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)] used to estimate model ms4. The models are evaluated with the validation data as shown in Figure 6.10.



Fig. 6.10. Evaluation of estimated NARX models with different number of units in 'wavelet' estimator for methane production

The result of the evaluation as seen in Figure 6.10, shows that there is a marginal improvement of about 1% on the previously estimated model ms4 when the number of units in 'wavelet' is reduced to 9. This indicates that model ms7 is a better model than models ms4,

ms8 and ms9, and the best performing NARX model with 'wavelet' estimator for methane production for this study.

Likewise, apart from 'wavelet' estimator two other nonlinearity estimators 'treepartition' and 'sigmoidnet' are used to estimate two new NARX models, ms10 and ms11, respectively. The models are estimated with the estimation data (ze), model order [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)] and their respective nonlinearity estimator. The estimated NARX models ms7, ms10 and ms11 are evaluated with the validation data as presented in Figure 6.11.



Fig. 6.11. Evaluation of NARX models with different nonlinearity estimators for methane production

The result of the evaluation as seen in Figure 6.11 shows that the model ms7 with 'wavelet' nonlinearity estimator performed better than the models ms10 and ms11 with 'treepartition'

and 'sigmoidnet' nonlinearity estimators, respectively, when compared with the validation data. Therefore, the estimated model ms7 is considered the best NARX model that describes the dynamics of mesophilic AD for methane production for this research work.

### 6.2.4 Hammerstein-Wiener model estimator

The Nonlinear Hammerstein-Wiener (NLHW) model consists of three serial blocks, such that a dynamic linear block is placed in between the input and output static nonlinear blocks, as presented in Figure 6.12 (Abbasi-Asl et al., 2012).



Fig. 6.12. Block diagram of Hammerstein-Wiener model (Abbasi-Asl et al., 2012)

The three components of Hammerstein-Wiener model are represented with the following three mathematical expressions (Abbasi-Asl et al., 2012).

For the static input nonlinear function,

$$w(t) = f(u(t)) \tag{6.4}$$

Where,

w(t) is an input to the linear function B/F.

For the dynamic linear block,

$$x(t) = \frac{B}{F}w(t) \tag{6.5}$$

Where,

x(t) represents a linear transfer function. The dimension of x(t) is also the same as y(t), while *B* and *F* have similar characteristics as polynomial in the linear OE model.

Considering the inputs  $n_u$  and outputs  $n_y$ , the dynamic linear block is a transfer function that has the matrix entries as (Abbasi-Asl et al., 2012):

Transfer function 
$$= \frac{B_{ji}(q)}{F_{ji}(q)}$$
 (6.6)

Where,

 $j = 1, 2, ..., n_y$  and  $i = 1, 2, ..., n_u$ , represent possible number of inputs and outputs, respectively

The output static nonlinear function is expressed mathematically as:

$$y(t) = h(x(t)) \tag{6.7}$$

The Eq. 6.7 is a nonlinear function that converts the output of linear block to the system output. Both w(t) and x(t) are internal variables that define the input and output of the linear block. The *f* function is known as the input nonlinearity because it acts on the linear block. While the *h* function signifies the output nonlinearity, because it acts on the output port of the linear block (Abbasi-Asl et al., 2012). However, in the absence of the input nonlinear block, the model becomes a Wiener model. Similarly, if the output nonlinear block is not present, then the model becomes a Hammerstein model (Choo et al., 2012).

#### Step 5b: NLHW model estimation for biogas and methane production

The NLHW models are estimated with the syntax (Ljung et al., 2007):

Mhw = nlhw(Data, Orders, InputNonlinearity, OutputNonlinearity);

Where Mhw is the NLHW model, model Order = [nb bf, nk], which specifies the orders and delay of the linear function, InputNonlinearity and OutputNonlinearity specify the nonlinearity estimators for two nonlinear blocks.

Similar to the NARX model estimation, the initial model order used for estimating the NLHW model mhw1 is [ones(1,4), ones(1,4), zeros(1,4)]. This means that in the linear block, the output is the sum of the four first order transfer function driven by the four inputs. The nonlinearity estimator is the pwlinear for both InputNonlinearity and OutputNonlinearity. The model mhw1 for biogas production is estimated using MATLAB command line. Two more NLHW models are estimated with the same InputNonlinearity and OutputNonlinearity, data de and the following model order:

[2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)] for model mhw2; and

[3\*ones(1,1), ones(1,4), ones(1,4)] for model mmh3

The estimated models are evaluated by comparing their simulated outputs to the output in the validation data, dv, as shown in Figure 6.13.



Fig. 6.13. Evaluation of NLHW models with pwlinear for both nonlinearities for biogas production

As seen in Figure 6.13, the result of the evaluation shows that the NLHW model mhw2, which has a Fit of 93.05% to the validation data, is considered a better model than models mhw1 and mhw3. This is because the Fit of mhw2 to the validation data is higher than that of mhw1 and mhw2.

Again, three other NLHW models, mhw4, mhw5 and mhw6, are estimated with the data de and model orders similar to the ones used to estimate models mhw1, mhw2 and mhw3 but with different nonlinearities for both InputNonlinearity and OutputNonlinearity - unitgain and saturation, respectively. The estimated models are evaluated by comparing their simulated outputs to the output in the validation data, dv, as shown in Figure 6.14.



Fig.6.14 Evaluation of NLHW models with unitgain and saturation nonlinearities for biogas production

The result of the evaluation as seen in Figure 6.14 reveals that model mhw5 has a higher Fit to the validation data than mhw4 and mhw6, indicating a better NLHW model.

Furthermore, the InputNonlinearity and OutputNonlinearity are replaced with saturation and deadzone, respectively, and used to estimate models mhw7, mhw8 and mhw9. The estimated models are evaluated with the validation data dv as presented in Figure 6.15.



Fig. 6.15. Evaluation of NLHW model with saturation and deadzone nonlinearities for biogas production

The result of the evaluation as seen in Figure 6.15 reveals that model mhw8 has a higher Fit to the validation data than mhw7 and mhw9, which signifies a better NLHW model.

The selection of the best NLHW model for biogas production is performed by comparing the simulated output of the best model from each of the set of the nonlinearity combinations presented in Figure 13, Figure 14 and Figure 15 (models mhw2, mhw5 and mhw8) and the output in the validation data as plotted in Figure 6.16.



Fig. 6.16. Comparison of the best NLHW model for biogas broduction

The result of the analysis seen in Figure 6.16 shows that the three estimated NLHW models performed very well in comparison to the output in the validation data. In addition, the three models are estimated with identical model order, [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)]. However, model mhw2 marginally describes the dynamic system better than the other two models, indicated by a slightly higher Fit to the validation data than models mhw5 and mhw8. Therefore, model mhw2 is the best NLHW model for biogas production for this study.

The estimation of NLHW model for methane production is performed using similar syntax used to estimate the best of the NLHW models for biogas production from each set of the nonlinearity combinations. Using the same model order, the three estimated NLHW models for methane production are evaluated by comparing their simulated output and the output in the validation data zv as shown in Figure 6.17.



Fig. 6.17. Evaluation of estimated NLHW models with different nonlinearity combinentios for methane production

As seen in Figure 6.17, the evaluation indicates that model mhws2 is marginally better than models mhws5 and mhws8, because model mhws2 better fits the validation data (96.04% vs. 94.79% and 94.37%). Therefore, model mhws2 is the best performing NLHW model for methane production for this research work.

## Step 6: Post-estimation analysis – model selection

The selection of the models for this study is carried out by evaluating the quality of the best estimated NARX and NLHW models for biogas and methane production, and choosing the one that best describes the mesophilic AD system behaviour within the acceptable bounds. This research study utilises two different Goodness of Fit (GoF) approaches to measure the quality of NARX and NLHW models, in order to determine the best model each that describes well the production of biogas and methane. These approaches include the following:

- Comparing simulated output to measured output; and
- Comparing models using Akaike Information Criterion (AIC);

# Comparing simulated output to measured output:

This approach is used already in this study to compare the performance of different models within the same model structure. The NARX and NLHW models for biogas and methane production are compared using the 'compare' command in MATLAB, in order to determine which model best fits the validation data.

For biogas production, the simulated outputs of the estimated NARX model (mx9) and NLHW model (mhw2) are compared to the output in the validation model (dv) as presented in Figure 6.18.



As seen in Figure 6.18, the evaluation indicates that NLHW model mhw2 is better than NARX model mx9, because model mhw2 better fits the validation data (93.05% vs. 85.60%). Therefore, comparing the simulated outputs of the estimated models NARX and NLHW and the output in the validation data, model mhw2 is the best model that describes the system better for biogas production for this research work.

Similarly, the simulated outputs of estimated NARX model ms7 and NLHW model mhws2, and the output in the validation data are compared for methane production and the result is presented in Figure 6.19.



Fig. 6.19. Comparison of the estimated NARX and NLHW models for methane production

The result of how well the different models (NARX model ms7 and NLHW model mhws2) fits the validation data, as seen in Figure 6.19, indicates that NLHW model mhws2 is better than NARX model ms7 because model mhw2 better fits the validation data (96.04% vs. 79.41%). Once again, comparing the simulated outputs with the measured output approach, the NLHW model mhws2 is the best model for methane production for this study.

#### **Comparing models using Akaike Information Criterion (AIC):**

The second approach used to select the best models for this research study is the AIC, which is utilised to compare the quality of the best NARX model and the best NLHW model for biogas and methane production The AIC provides a measure of model quality by simulating the situation where the model is tested on a different data set (Ljung, 2008). Akaike's theory states that the most accurate model is the one with the smallest AIC (Ljung, 2008, Ljung, 2015). The AIC is defined by the following equation (Ljung, 2008, Ljung, 2015):

$$AIC = \log V + \frac{2d}{N} \tag{6.8}$$

Where V is the loss function, d is the number of estimated parameters, and N is the number of values in the estimation data set.

The loss function *V* is defined by the following equation (Ljung, 2008):

$$V = \det\left(\frac{1}{N}\sum_{1}^{N}\varepsilon(t,\theta_{N})(\varepsilon(t,\theta_{N}))^{T}\right)$$
(6.9)

where  $\theta_N$  represents the estimated parameters.

For d << N:

$$AIC = \log\left(V(1 + \frac{2d}{N})\right) \tag{6.10}$$

Computing AIC in MATLAB, the aic command is used to compare the different models for biogas and methane production and the syntax is as follows (Ljund, 2008).

AIC = aic(m1, m2) for biogas and methane production.

For biogas production, the comparison of NARX model mx9 and NLHW model mhw2 using AIC approach indicates that NLHW model mhw2 is a more accurate model than NARX as it gives the smaller AIC (9.6273 vs. 13.8054).

Similarly, the execution of the aic command indicates that NLHW model mhws2 is a more accurate model than NARX ms7 as it gives the smaller AIC (8.0308 vs. 13.2119) for methane production.

The measure of the quality of the NARX and NLHW models for biogas and methane production using the two GoF approaches indicates that NLHW model mhw2 and NLHW model nhws2 are the best models for biogas production and methane production, respectively, for this study. Therefore, NLHW model mhw2 is chosen as the model for biogas production, and NLHW model mhws2 chosen for methane production.

The mathematical expression for NLHW models is as follows:

$$y(t) = \left[\frac{B(z)}{F(z)}\right]u(t) + \varepsilon(t)$$
(6.11)

Where, y(t) is the output (biogas or methane production), u(t) is the input (operating parameters) and  $\varepsilon(t)$  is the noise signals

For biogas production, the domain-mathematical expression from the estimated nonlinear multi-parameter model, NLHW model mhw2, is as follows:

$$G_{b}(z) = \frac{1 - 15.8 z^{-1}}{1 - 1.935 z^{-1} - 0.9871 z^{-2}} * \text{Temp} + \frac{0.8907 - z^{-1}}{1 - 1.963 z^{-1} + 0.9837 z^{-2}} * \text{pH} + \frac{-0.7488 + z^{-1}}{1 - 1.811 z^{-1} + 0.8417 z^{-2}} * \text{MS} + \frac{-0.12 + z^{-1}}{1 - 1.947 z^{-1} + 0.9988 z^{-2}} * \text{Press} + \varepsilon(t)$$
(6.12)

Similarly, for methane production, the domain-mathematical expression from the estimated nonlinear multi-parameter model, NLHW model mhws2, is as follows:

$$G_m(z) = \frac{1 - 0.7134 z^{-1}}{1 - 1.876 z^{-1} - 0.9137 z^{-2}} * \text{Temp} +$$

$$\frac{-0.698 - z^{-1}}{1 - 1.913 z^{-1} + 0.9218 z^{-2}} * pH + \frac{1 - 0.9726 + z^{-1}}{1 - 1.993 z^{-1} + z^{-2}} * MS + \frac{1 - 0.2323 + z^{-1}}{1 - 1.908 z^{-1} + 0.9307 z^{-2}} * Press + \varepsilon(t)$$
(6.13)

Where, Temp (in °C), pH, MS (in rpm) and Press (in the bar) are temperature, pH, mixing speed and pressure respectively, while  $G_b(z)$  and  $G_m(z)$  (in mL) represent biogas and methane production, respectively. The following are a range of the values of the operating parameters at which the Eq. 6.12 and Eq. 6.13 are based on. 34<Temp<42 °C; 5.0<pH<8.0; 60<MS<2000 rpm; and 0.1<Press<0.7

## **Step 7: Model testing: simulation error analysis**

Simulation error occurs as a result of inaccurate input data, inaccurate physical model and limited accuracy of the solutions of the governing equations (Kocak, 2012; Christie, 2005; Oberkamfa, 2002).

The input error is the error in the data utilised to specify the problem. It is the error from the experimental error, which is classified as random or systematic errors (Christie et al., 2005; Oberkamfa, 2002). Random and systematic errors are problems associated with measurements (Christie et al., 2005; Oberkamfa, 2002). The physical model error is due to the effects of the phenomena, which are inadequately represented in the simulation (Kocak, 2012; Christie et al., 2005). Solution error is the difference between the approximate solution of the equations obtained with the numerical algorithms used in the simulation and the exact mathematical solution of the governing equations for the model (Kocak, 2012; Christie et al., 2005; Oberkamfa, 2002).

It is assumed that the models constructed in this study are not affected by the measurement and physical errors. This is due to the care is taken in the experimental setup and acquisition of data, as well as following the principle and procedure for identifying nonlinear black-box models using System Identification method. However, there is the possibility of error that may exist in the constructed models.

To analyse the error, the Root Mean Square Error (RMSE) is used. The RMSE is a commonly used measure of the difference between the measured outputs of a system in consideration and the outputs predicted by a model of the system (Chai and Draxler, 2014). The individual differences are called residuals, and the RMSE is used to combine the errors into a single measure of productive power (Chai and Draxler, 2014).

The RMSE of a model prediction with respect to the estimated output is the square root of the mean squared error, which is computed as follows (Chai and Draxler, 2014).

$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (\hat{y}_i - y_i)^2}{n}} (mL)$$
(6.14)

Where, *n* is samples of model errors, *y* is measured output values and  $\hat{y}$  is simulated output values

The percentage error of the system for biogas and methane production is calculated by dividing the RMSE by the measured output values, as follows.

$$\% Error = \frac{RMSE}{\sum_{i=1}^{n}(y)} \ge 100$$
(6.15)

The Simulink model as shown in Figure 6.20a, which represents Eq. 6.15 and Eq. 6.16, is used to simulate some intervals of a range of variables of two of the operating parameters at which

the models are based on. The internal part of error estimator in Figure 6.16 is shown in Figure 6.21. The variables of temperature and pH which are used experimentally to generate the measured output values (biogas or methane), are also utilised to simulate the models for biogas and methane production, in order to compute the RMSE and the percentage error between the simulated output values of the models mhw2 and mhws2, and the measured output values of biogas and methane production.



Fig. 6.20a. Simulation of percentage error from models for biogas and methane production



Fig. 6.20b Error estimator

The following Tables show the result of the Simulink model, containing simulated output, measured output and percentage error for biogas and methane production.

Temperature (°C)	39	43
Simulated output (mL)	2809	2310
Measured outpit (mL)	2752	2460
Percentage error (%)	2.1	-6.5

Table 6.1a. Test for biogas production model mhw2 for variables of<br/>temperature

# Table 6.1b. Test for methane production model mhws2 for variables of temperature

Temperature (°C)	39	43
Simulated output (mL)	1830	1522
Measured outpit (mL)	1875	1553
Percentage error (%)	-2.4	-2.0

Table 6.2a. Test result for biogas production model mhw2 for variables of pH

рН	7.3	8.3
Simulated output (mL)	2620	2240
Measured outpit (mL)	2570	2149
Percentage error (%)	1.9	4.2

Table 6.2b. Test result for methane	production model mhws2 for variables of p	Η
	1 1	

рН	7.3	8.3
Simulated output (mL)	1691	854
Measured outpit (mL)	1734	827
Percentage error (%)	-2.5	3.3

The analysis of simulation error as seen in Tables 6.1a - 6.2b, compares the simulated and measured output for biogas and methane production. It shows that the percentage error for biogas and methane production are less than 7% and 4%, respectively, signifying a small simulation error. The negative percentage error in some of the values of the tests indicates that the measured output is higher than the simulated output at those variables and vice verse.

#### 6.2.5 Comparison between the biogas and the methane production models

The biogas and methane production models are constructed using the same nonlinear blackbox model structure, NLHW models. The similarities of biogas production and methane production model support the response of biogas and methane production as seen in the plots in Chapter 5. This goes to strengthen the claim that methanogenesis is the limiting stage in AD process. Meaning that the behaviour of methanogenic bacteria determines the performance of AD process with regards to biogas production.

Although the NLHW models produced a high fit to the respective estimation data for biogas and methane production, there is the possibility of the following weaknesses in these models.

- 1. It is uncertain that these models can fit well data from the four input parameters that are significantly higher or lower than the data used in this study.
- 2. If the production of biogas and methane are considerably influenced by other operating parameters outside temperature, pH, mixing speed and pressure, it is unlikely that these models will give a good fit to the data, or able to predict the behaviour of mesophilic AD system well.

#### 6.3 Bioprocess Scale-up

There are several ways of developing and implementing biological processes, which are operated at different scales, such as laboratory, pilot and large scales. Many laboratory-scale processes that are used for research work are developed with the goal of screening and developing better bacterial strains, improving existing culture and performing the process more effectively with enhanced productivity (Najafpour, 2006). Usually, the production design of large-scale bioprocesses is developed in the laboratory, where all the factors that influence the intended output production are evaluated through experiments (Najafpour, 2006). Many factors, such as physical and biochemical, influence the production of biogas and methane as discussed in Chapter 2. The results from the laboratory process are transferred to the pilot-scale for verification by a relatively simple scale-up, after which they are transferred to large-scale (Galindo and Ramirez, 2013; Najafpour, 2006).

In other to scale-up a bioprocess, a useful technique called dimensional analysis (DA) is applied (Galindo and Ramirez, 2013; Moreira and Wallece, 2012; Najafpour, 2006). DA is the analysis of the relationship between various physical quantities by identifying their basic dimensions and units, as well as tracking these dimensions as calculations or comparisons (Gibbings, 2011). In DA, only quantities with similar dimensions are subtracted, added or compared (Barnett, 2007). Applying DA to scale-up requires certain rules and criteria to be followed. These criteria include maintaining geometric similarities and the keeping dimensionless numbers (DNs) constant (Galindo and Ramirez, 2013; Moreira and Wallece, 2012; Najafpour, 2006).

One way of obtaining DNs is from a structured grouping of all the parameters and variables that influence bioprocess outcome (Galindo and Ramirez, 2013). However, this method is

prone to errors as the omission of one important variable can differ the outcome (Galindo and Ramirez, 2013). The other way to achieve the DNs is to work with the ratios of time constants, forces and velocities (Galindo and Ramirez, 2013). This method is less prone to errors and often used (Galindo and Ramirez, 2013). The DNs from the ratios include Reynolds number or momentum factors, power consumption per unit volume of liquid, impeller tip velocity, liquid mixing and recirculation time and volumetric of mass transfer coefficient, which are discussed in detail in the books written by Simpson and Sastry (2013), Moreira and Wallece (2012) and Najafpour (2006).

According to Najafpour (2006), the geometric similarities and constant power per unit volume have been applied in the scale-up of most fermentation processes for alcohol and organic acid production, which are some of the products of acidogenesis, the second stage of methane production in AD process. However, there are some limitations associated with applying DA for scale-up of bioprocess (Moreira and Wallece, 2012; Najafpour, 2006). It is practically impossible to keep all the DNs constant during scale-up (Moreira and Wallece, 2012; Najafpour, 2006). For instance, mixing and mixing time is a major concern in applying DA for scale-up, because large-scale digesters may not have uniform mixing, whilst mixing is not a problem in lab-scale and pilot-scale digesters (Najafpour, 2006).

In applying DA for scale-up, the operating parameters are well controlled and certain rules have to be followed in changing the digester size to meet special criteria (Najafpour, 2006). In addition, maintaining only digester similarities and keeping some DNs constant may not satisfy the requirements of bioprocess design (Moreira and Wallece, 2012; Najafpour, 2006). The basic concept of DA, which is the geometric similarity, is usually maintained in scale-up from

laboratory to pilot scale. However, in the large-scale, it presents a real design challenge because the configurations of the digester are often changed (Moreira and Wallece, 2012).

#### 6.3.1 Scale-up of Black-box Model

The black-box approach is based on laboratory-scale or pilot-scale investigations, which focuses mainly on the manipulated and measured parameters (Heams, 2006). These parameters, such as physical, chemical and biochemical phenomena, can form the major factors of scale-up (Steel, 2007; Heams, 2006). The advantage of black-box model approach is that it allows for faster development of the determined configuration, by considering the target as several small black-box systems (Heams, 2006).

However, in opposition to nonlinear black-box model approach to scale-up, Heams (2006) reports that this approach offers less assurance as there are many physical, chemical, biochemical and microbiological phenomena that are not well explained (Heams, 2006). Meaning that the rules to maintaining the geometric similarities as well as keeping the DNs constant are not followed in nonlinear black-box model scale-up. Additionally, some authors reported that nonlinear black-box approach limits applications to the experimental conditions that generated the model, thereby, allows only the interpolation of the results and consequently, difficult to use for scale-up (Steel, 2007; Cybulski et al., 2001; Wentzel and Ekama, 1997).

The nonlinear black-box models constructed in this study focuses only on four manipulated parameters; temperature, pH, mixing speed and pressure. Although the geometric similarities required to apply DA to scale-up can be maintained, it is not possible to account for all the physical and biochemical parameters involved in a bacterial growth environment that are not considered in this study. It means that the nonlinear black-box models constructed in this study

lack the basis to apply DA to scale-up. Since the models are not able to satisfy the rules and criteria for DA, they cannot be used for the scale-up purpose.

One of the ways it can be possible to apply nonlinear black-box model to scale-up is to make assumptions about all the manipulated physical and biochemical operating parameters not considered in this study. Then the nonlinear black-box models can satisfy the criteria to apply DA to scale-up, however, the model would not be considered a black-box, rather, a grey-box model. Another way is if the digester geometric increase produces a corresponding linear output, then the target can be considered as a multiple of laboratory linear black-box model. Further study is encouraged in this area to explore the possibility of nonlinear black-box model scale-up in mesophilic AD process.

# 6.4 Summary,

This chapter has described how the nonlinear multi-parameter black-box models for biogas and methane production are constructed. The process includes the selection of black-box model technique due to inadequate knowledge of AD process. The estimation and validation of the experimental data are performed in MATLAB, through System Identification toolbox. The identified NLHW models mhw2 and mhws2 represent the behaviour of the production of biogas and methane, respectively, from mesophilic AD. Of all the candidate models studied, the nonlinear models provide a superior reproduction of the experimental data over the whole analysed period. Other appealing features of these HLHW models reside in the simplicity of the nonlinearities considered and the possibility to include new nonlinearities, as well as in its easy implementation. The models are prone to possible weaknesses and cannot be applied for scale-up purpose due to the inability of black-box models to satisfy the dimensional analysis criteria.

# Chapter 7 Conclusions

# 7.1 Introduction

The study is set out to develop a black-box model for mesophilic AD process, in order to understand better, how physico-environmental operating parameters influence biogas and methane production. The research has also sought out to consider the control limits for the controlled parameters necessary for improving methanogenesis reaction while ensuring an enhanced digestion operation. The review of the relevant academic articles showed that much progress has been made in AD technology. However, much work is still needed to improve the knowledge of AD, specifically, the interactions between the operating parameters and bacterial community. In addition, the composition of bacterial groups that varies with changes in the operating parameters.

This research work sought to provide solution to the following objectives:

- Identify and understand the performance variables of mesophilic AD system (Chapter 2);
- A critical review of the operating parameters that influence the performance of biogas and methane production was undertaken. For this research work the following four operating parameters are considered; temperature, pH, mixing speed and pressure (Chapter 2);
- Determine the limits for the four operating parameters and their impact on biogas and methane production. This includes a series of laboratory experiments with each of the operating parameters. The data obtained from the investigations are analysed and the range of values for each operating parameter that recorded the highest performance in terms of biogas and methane production (Chapters 4 & 5);

- Construct a black-box model each for biogas and methane production. The limits of the four operating parameters considered are analysed further in order to establish their correlation with biogas and methane production. Two nonlinear multi-parameter black-box models are constructed through System Identification method in MATLAB. The models are able to predict the behaviour of the real system with sufficient accuracy (Chapter 6); and
- Determine the scalability of black-box models. The black-box models constructed in this study focused only on four manipulated parameters. Although the models can maintain the geometric similarities required to apply DA to scale-up, it cannot account for all the other physical and biochemical parameters involved in the bacterial growth environment that are not considered in this study, making it difficult to use the models for scale-up (Chapter 6).

# 7.2 Observed Outcomes

The main observed outcomes with respect to the individual objectives of the study are chapter specific. This section unifies the observed outcomes to provide a solution to the aforementioned objectives of this study.

- Identify and understand the performance variables of mesophilic AD system through a critical review of the operating parameters that influenced the process performance:
- i. Influence of the operating parameters on AD performance: Various operating parameters influence the performance of AD system. Different operating parameters have a different degree of impact on biogas and methane production. The temperature is considered the most significant parameter in AD process as it influences the entire biochemical process. Anaerobic bacteria are grouped based on their temperature

preference and tolerance. The different types of AD are characterised by their bacterial group, which are based on the temperature favourite of the bacterial group. These include psychrophilic, mesophilic and thermophilic AD, which operates at 0 - 20 °C, 10 - 45 °C and 45 - 65 °C, respectively. The mesophilic AD process, which is the type of AD process considered in this study, performs optimally at a temperature range of 38 - 40 °C (Chapter 3).

Similarly, pH is an important parameter in AD, however, its variability does not impact on the entire anaerobic reactions. This is because different bacterial groups of the multistep process of AD have variable pH preferences, irrespective of the type of AD implemented. In Chapter 5 of this thesis, it is found that there is no record of methane production at pH of 5.0. This is because the activity of methane forming bacteria is inhibited in acid condition. However, acidogenesis thrive in acidic condition up to a low pH of 4.0 (Chapter 5).

Furthermore, some of the operating parameters, such as mixing/mixing speed, hydraulic retention time, pressure and substrate characteristics; do not actually halt biogas and methane production as in the case of temperature and pH. However, they influence the enhancement of biogas and methane production (Chapter 5).

**ii. Anaerobic digestion process improvement:** Various enhancements have been achieved in AD technology in other to improve biogas and methane production. These include pre-treatment, co-digestion and system modelling.

- iii. Co-digestion of two or more substrates The simultaneous digestion of two or more different substrates improves digestion process as investigated in this research work. In co-digestion, the participating substrates supply the required nutrients, which improves the performance of AD. It is found that improving the nutrient balance resulted in improved biodegradation, process stability and biogas production. The experiments reported in Chapter 5 show that better biodegradation and conversion of substrates to biogas is achieved by co-digestion of wastewater sludge and food waste compared to the digestion of only wastewater sludge or food waste. The study also found that co-digestion of substrates produced more methane than the single-substrate digestion, thus, improving the quality of biogas. This indicates that co-digestion offers better process performance than single-substrate digestion (Chapter 5).
- iv. System Modelling A number of previous studies (Chapter 2) reviewed show that modelling is an important tool for a complex system like AD process. It is used to describe and predict the behaviour and output of systems in real life. Mathematical modelling is generally classified into white-box models and black-box models. However, when there is the availability of some information about the system under consideration but not enough to apply the white-box models, then the known information is combined with the input/output data generated from physical experimentation, which is known as a grey-box model. In other words, grey-box model is the combination of the white-box and black-box models (Chapter 3). The black-box models are selected (Chapter 6). Moreover, some of the previous models constructed for predicting biogas production are reviewed and the outcome contributed to shaping the basis for developing the models for this study (Chapter 3).

- Determining the limits for some key parameters that influence mesophilic AD: Several factors, which are categorised into the physical, biological and chemical need to be considered for achieving effective mesophilic AD. This study considers the bacteriological and biochemical processes involved in AD (Chapter 4). In addition, it investigated the impacts of temperature, pH, mixing speed and pressure on AD under mesophilic conditions. The goal is to measure and compare the biogas and methane production for all the variables of temperature, pH, mixing speed and pressure, in order to determine the variables that achieved the highest biogas and methane production (Chapter 4).
- i. Bacteriological and biochemical processes involve in anaerobic digestion: The process of AD involves complex interactions of diverse species of bacteria organised in four different steps that function almost simultaneously. The four steps include hydrolysis, acidogenesis, acetogenesis and methanogenesis. In hydrolysis, complex organic polymers such as carbohydrates, fats and proteins are broken down to simple sugars, lipids and amino acids. The acidogenic bacteria convert the products of hydrolysis into hydrogen, carbon dioxide, alcohols and VFAs. At this stage, acetic acid contained in the VFAs are converted directly to methane by methanogenic bacteria. The remaining products of acidogenesis are transferred to the acetogenic stage, where they are converted to mainly acetic acid and finally, the methanogenic bacteria utilise the acetic acid as a substrate to produce methane.

Methanogenesis is considered the rate-limiting stage of the AD process, thereby determining the stability of the entire process. Methanogenic bacteria are very sensitive to changes in operating parameters. Since they limit the kinetics of the entire anaerobic
biochemical process, the operating parameters of the whole process need to be maintained at the optimal range of methanogenesis (Chapter 4).

## ii. Investigation of the influence of multiple operating parameters:

**Temperature:** The influence of mesophilic digestion of glucose for all the following variables of temperature, 32, 34, 36, 38, 40 and 42 °C on biogas and methane production is studied. Each of the experiments was kept at a constant temperature for the whole duration of 24 hours. The results show that the digester set at 32 °C for the entire duration generated the lowest biogas and methane production. Whereas the highest biogas and methane production are from the digester maintained at 40 °C, followed by the 38 and 36 °C digesters. The production of biogas and methane are found to decline as the digester temperature marginally exceeds 40 °C, indicating that the 42 °C temperature is beyond the temperature that supports the growth of mesophilic anaerobic bacteria.

From this study, the optimum temperature range is between 38 - 40 °C. This means that in order to achieve a constant improved biogas production in mesophilic AD, digester temperature needs to be maintained within the optimal temperature range for the entire duration of the process (Chapter 4).

**pH:** The influence of pH variation on the biogas and methane production was also studied. Different pH value is maintained for each digester throughout the retention time of 24 hours. The pH values investigated include 5.0, 6.0, 7.0 and 8.0. It is found that the pH of 5.0 inhibited the methanogenic process, thereby no methane is produced. However, the digester that was operated at pH of 7.0 generated the highest biogas and

methane production. The result also shows that as the pH moves below or beyond 7.0, the biogas and methane production decline. Indicating that as closer the pH value is to 7.0, the neutral value, the better the performance of the

methanogenic process. Thus, for continuously improved biogas and methane production, the pH needs to be kept close to the neutral value (Chapter 4).

**Mixing speed:** Mixing is important for optimal AD performance; however, excessive mixing can cause a number of problems such as lowering the surface tension of solution in the digester, and accumulation of solids over liquids within the digester. Results from the experimentation of the influence of mixing speed variation on biogas production show that digesters that operated at minimal or moderate mixing speed performed better than the digester with vigorous mixing. Among all the four mixing speeds experimented, the 60, 300 and 600 rpm are found to produce more biogas and methane than the 2000 rpm digester. In addition, the digester that operated at 2000 rpm mixing speed delayed biogas and methane production at the initial stage of the process due to the excessiveness of the mixing. The vigorous mixing does not only impact negatively on the performance of AD, but it also increases the cost of energy used by the mixing system, thereby increasing the overall operating cost of the system and making the AD technology less attractive (Chapter 4).

**Pressure:** The results of the experiment carried out to evaluate the influence of digester operating pressure on biogas and methane production shows that the 0.7 bar digester produced less biogas than the 0.1 bar digester. However, the methane production by the 0.7 bar digester is more than the 0.1 bar digester. This is because more carbon dioxide content in biogas from 0.7 bar digester is liquefied compared to biogas produced by

0.1 bar digester. It is therefore recommended that appropriate digester operating pressure need to be chosen, but not exceeding 100 bar, which is the maximum operating pressure that methanogenic bacteria can tolerate (Chapter 4).

• Constructing multi-parameter models for biogas and methane production:

This study has presented two nonlinear multiparameter models for mesophilic AD for predicting the biogas production and the methane production. The modelling utilised raw data generated from lab experiments of the influence of four operating parameters, temperature, pH, mixing speed and pressure on the biogas and methane production. Due to the nonlinear characteristics of the acquired data, the nonlinear black-box modelling technique is used to construct the models.

The black-box modelling offers some advantages over the white-box modelling technique; it requires no in-debt knowledge of the system under consideration, as well as adequately predict the biogas and methane production from input-output data. In addition, black-box modelling considers all the stages of the anaerobic process, operating parameters and history of operational data unlike the white-box modelling (Chapter 6).

Two nonlinear model structures, autoregressive with exogenous input (NARX) and Hammerstein-Wiener (NLHW) with different nonlinearity estimators and model orders are chosen by trial and error and utilised to estimate the models. The performance of the models is determined by comparing the simulated outputs of the estimated models and the output in the validation data. The approach is used to validate the estimated models by checking how well the simulated output of the models fits the measured output. The best models for biogas and methane production are chosen by comparing the outputs of the best NARX and NLHW models (each for biogas and methane production), and the validation data, as well as utilising the Akaike information criterion. The NLHW models mhw2 and mhws2 are chosen for biogas and methane production, respectively. The identified NLHW models mhw2 and mhws2 represent the behaviour of the production of biogas and methane, respectively, from mesophilic AD. Of all the candidate models studied, the nonlinear models provide a superior reproduction of the experimental data over the whole analysed period. (Chapter 6).

The favourable outcomes indicate that black-box simple models can estimate biogas and methane production, as well as contributed to improving the knowledge of mesophilic AD process. The constructed models can replicate the mesophilic AD process and forecast its response for a given multiple operating parameters used in the study with sufficient accuracy. Although the models may not have been the perfect representation of the behaviour of mesophilic AD process with regards to predicting the biogas and methane production. However, this study has opened up the opportunity for further research on modelling mesophilic AD process with multiple parameters. It is then inferred that this research work on the construction of mesophilic AD models for predicting biogas and methane production is successful.

• Determining the scalability of black-box model.

This study considered only four operating parameters in the construction of the blackbox models. Although geometric similarities required to apply DA (Dimensional Analysis) to scale-up can be achieved, but it is not possible to account for all the other physical and biochemical operating parameters involved in a bacterial growth environment that are not considered in this study.

In addition, with the nonlinear black-box models, it is practically impossible to keep all the DNs (Dimensionless Numbers) constant, meaning that the models constructed in this study lack the basis to apply DA to scale-up. Since the models are not able to satisfy the rules and criteria for DA, they cannot be used for the scale-up purpose (Chapter 6).

## 7.3 Limitation of this study

The study has presented an empirical contribution to improving mesophilic AD process, which is achieved through experimentation and applying System Identification Toolbox in MATLAB to construct nonlinear black-box models. This research work encountered possible limitations that need to be considered. The limitations are categorised in terms of methodological and longitudinal effects that influenced the experimentation and interpretation of the study.

- 1. Methodological limitations:
- Sample size The design of the experiment is such that only one operating parameter is manipulated at a time while the other three operating parameters are maintained at the same value throughout the investigation for all the variables of the manipulated operating parameter. This methodology is repeated for all the variables of temperature, pH, mixing speed and pressure. As a result, the number of sample units used for this study is reduced. The study is not able to capture the response of biogas and methane production if the design of the experiment is different. Because of the reduced sample size, it is difficult to find a comprehensive relationship between operating parameters and biogas as well as methane production.

• Lack of available and reliable data – The scope of analysis for this study is limited due to the insufficient evaluation of produced biogas sample to identify the entire gas composition. The reason for this is due to the gas analysed used in this study not equipped enough to identify and measure the proportion of other gases like carbon dioxide, hydrogen and ammonia present in biogas sample. Not only did it limit the scope of analysis conducted in this research work, but also significantly hindered the study of the relationship between operating parameters and other gases contained in biogas sample.

In addition, there is the possibility of lack of data reliability, which is likely to be as a result of biogas sample contamination, resulting from transporting biogas sample from the laboratory in the University of Hertfordshire, where it is produced, to the laboratory in Imperial College London, where it is analysed. Another reason that could be responsible for data unreliability, in this research work, is the possible laboratory incompatibility that may alter the environmental conditions of the biogas samples.

- Mode of data collection The sample interval for the experiments performed in this research work could also contribute to the possible limitations in this study. Reducing sample interval from 60 minutes (used in this study) to 60 seconds, would increase the sample size and reduce any possible random error effect. Also, the manual acquisition of data, due to the unavailability of computerised data acquisition system, may likewise affect the accuracy of data.
- 2. Researcher limitations:

- Longitudinal effects The availability of time to conduct a thorough study is constrained by the due date of this research work. It did not only limit the scope of the study but also reduced the number of experiments conducted (sample size), which could have provided more information about mesophilic AD, thereby, improving the accuracy of the outcome of this study.
- **Insufficient equipment**: The lack of sufficient equipment is one of the major factors that limited this study. It adversely affected almost every stage of this research work, from experimental setup to sample size, data collection and data analysis.

## 7.4 Recommendations for Further Work

In other to construct models that can reflect the behaviour of mesophilic AD with improved predictive accuracy, additional studies are required, which include the following.

- a) A further study is needed with the design of experimental setup in such a way that a variable of the operating parameters is investigated at a time with all the possible combinations of all the variables of other operating parameters. This methodology would require many experiments to be conducted. The purpose is to increase the sample size, as well as generate the best combination of the variables of the operating parameters at which the optimal production of biogas and methane is attained. In addition, it will increase the amount of information that can be used to improve the accuracy and robustness of the models.
- b) Further investigation consisting of more operating parameters that influence biogas and methane production is encouraged. The models developed in this study are limited in their ability to predict the behaviour of mesophilic AD due to the influence of other operating parameters that are not considered in this study. Thus, a study that

incorporates more operating parameters would produce a model that is more dynamic and more robust, with improved predictive accuracy.

- c) Additional study is also required to determine the entire gas content of biogas, as well evaluate how the individual gases respond to different combinations of the variables of the operating parameters. The study can be undertaken in conjunction with the previous recommended further studies or as a separate study utilising already acquired data. The proposed study is set out to find what combination of the variables of the operating parameters encourages the production of any of the gases that constitute biogas. The findings of the study can then be utilised to construct a model that improves methane production.
- d) Further study is required to explore the possibility of black-box model scale-up, as it relates to mesophilic AD process.

This study has developed two pilot nonlinear multi-parameter black-box models for mesophilic AD process, which are based on the input-output data. This has demonstrated the need for developing multi-parameter models that reflect the behaviour of mesophilic AD process with sufficient accuracy. With adequate models in place, it is possible to evaluate the actual system, in order to reduce process disturbance that reduces or inhibits process efficiency.

Based on the models constructed in this research work, further studies are possible with the goal of improving biogas and methane production. The enhancement of methane production through mesophilic AD process is an appropriate answer to the global quest for alternative

energy source that is sustainable, as well as a satisfactory solution for environmental and health risks associated with fossils fuels.

## References

Abbasi T., Tauseef S. M., Abbasi S. A. (2012): Biogas Energy. SpringerBriefs in Environmental Science, Volume 2; 2012, ISBN: 978-1-4614-1039-3; p 11-23. Available online at: http://www.springerlink.com/content/978-1-4614-1040-39

Abe F., Horikoshi K. (2001): The biotechnological potential of piezophiles. *Trends Biotechnol.* 2001, *19*, 102–108.

Abreu A. A., Alves J. I., Pererie M. A., Karakashev D., Alves M. M., Angelidaki I. (2010): Engineered heat treated methanogenic granules: A promising biotechnological approach for extreme thermophilic biohydrogen prediction. Bioresources Technology. Volume 101, Issue 24, December 2010, Pages 9577-9586.

Agdag, O. N., Sponza, D. T. (2007): Co-digestion of mixed industrial sludge with municipal solid wastes in anaerobic simulated landfilling bioreactors. J. Hazard. Mat. 140, 75–85.

AgSTAR (2012): Increasing Anaerobic Digester Performance with Co-digestion. September 2012. Available online at: http://www.epa.gov/agstar/documents/codigestion.pdf

Ahn J. and Forster C.F. (2002): The effect of temperature variations on the performance of mesophilic and thermophilic anaerobic filters treating a simulated paper mill wastewater; *Process Biochemistry* 37: 589–594.

Ahring B. K (2003): Perspectives for Anaerobic Digestion. Biomethanation I. Advances in biochemical engineering/biotechnology, vol. 81Sringer, Berlin (2003), pp. 1–30.

Ahuja D. and Tatsutani M. (2009): Sustainable energy for developing countries », *S.A.P.I.EN.S* [Online], 2.1. 2009, Online since 07 April 2009. Available online at: URL http://sapiens.revues.org/823

Al Seadi T., Rutz D., Prassl H., Köttner M., Finsterwalder T., Volk S., and Janssen R., (2008): Biogas Handbook. Esbjerg, Denmark, ISBN: 978-87-992962-0-0

Alexandridis A. and Zapranis A. (2014): Wavelet Neural Network: With Applications in Financial Engineering, Chaos, and Classification. Publisher: John Wiley & Sons, May 2014, ISBN: 9781118596296.

Ali Shah F., Mohmood Q., Maroof Shah M., Pervez A., Ahmad Asad S. (2014): Microbial Ecology and Anaerobic Digester: The Key Players of Anaerobiosis. The Scientific World Journal. Volume 2014 (2014), Article ID 183752, 21 pages. Available online at: http://www.hindawi.com/journals/tswj/2014/183752/

Alters S. (2000): Biology: Understanding Life; 3<sup>rd</sup> Edition. Copyright © 2000 by Jones and Bartlett Publication, Inc. Pg. 90-96.

Amani T., Nosrati M., Mousavi S. M.; Kermanshahi R. K. (2011): Study of syntrophic anaerobic digestion of volatile fatty acids using enriched cultures at mesophilic conditions. Int. J. Environ. Sci. Tech., 8 (1), 83-96.

Amevi Acakpovi , Essel Ben Hagan, "Novel Photovoltaic Module Modelling using MATLAB/ Simulink", International Journal of Computer Application, Volume 83 – No.16, December 2013.

André M. J. (2013): Modelling 18O2 and 16O2 unidirectional fluxes in plants: IV. Role of conductance and laws of its regulation in C3 plants. J Biosyst 113:104–114 http://dx.doi.org/ 10.1016/j.biosystems.2012.11.002.

Angelidaki I.; Petersen S. P. and Ahring, B. K. (1990): Effects of lipids on thermophilic anaerobic digestion and reduction of lipid inhibition upon addition of bentonite. Applied microbiology and biotechnology. Vol. 33 (4): 469-472.

Angelidaki I. I., Ellegaard L., Ahring B. K. (1999): A comprehensive model of anaerobic bioconversion of complex substrates to biogas Biotechnol Bioeng 63: 363-372.

Angelidaki I. and Ellegaard L. (2003): Co-digestion of manure and organic wastes in centralised biogas plants. *Applied biochemistry and biotechnology*. Vol. 109 (1-3): 95-105.

Anscombe F. J. (1973): Graphs in Statistical Analysis. *The American Statistician*, Vol. 27, No. 1. (Feb., 1973), pp. 17-21.

Antonio MASSARUTTO (2001): Municipal Waste Management in Italy. Working paper CIRIEC No. 2010/01. Available online at: http://www.ciriec.ulg.ac.be/fr/telechargements/WORKING\_PAPERS/WP10-01.pdf

Appels L., Lauwers J., Degrève J., Helsen L., Lievens B., Willems K., Van Impe J., Dewil R. (2011): Anaerobic digestion in global bio-energy production: potential and research challenges. Renewable and Sustainable Energy Reviews 15, 4295e 4301.

Arlot A. (2010): A survey of cross-validation procedures for model selection. Statistical surveys, Vol. 4 (2010) 40-79, ISSN: 1935-7516; DOI: 10.1214/09-SS054.

Ascari S., Di Marzio T., Massarutto A., (1992): L'igiene urbana: economia e politica ambientale, Franco Angeli, Milano.

ATSDR. Atlanta, GA (2001): Agency for Toxic Substances and Disease Registry; 2001. Landfill Gas Primer: An Overview for Environmental Health Professionals. Available online at: http://www.atsdr.cdc.gov/HAC/landfill/html/intro.html.

Ayomoh M.K.O., Oke S.A., Adedeji W.O., Charles-Owab O.E. (2008): An approach to tackling the environmental and health impacts of municipal solid waste disposal in developing countries. Journal of Environmental Management 88 (2008) 108–114.

Babaee A., Shayegan J., Roshani A. (2013): Anaerobic slurry co-digestion of poultry manure and straw: effect of organic loading and temperature. Journal of Environmental Health Sciences & Engineering 2013 11:15. Available online at: http://www.ijehse.com/content/pdf/2052-336X-11-15.pdf

Barbier E. (2002): Geothermal energy technology and current status: an overview. Renewable and Sustainable Energy Review. Volume 6, Issues 1-2, 2002, Pages 3-63.

Barker N., Evans L. (2009): Evaluation of Opportunities for Converting Indigenous UK Wastes to Fuels and Energy (Report), NNFCC 09-012. Available online at: http://www.nnfcc.co.uk/tools/evaluation-of-opportunities-for-converting-indigenous-uk-wastes-to-fuels-and-energy-report-nnfcc-09-012

Barnett W11. 92007): Dimensions and Economies: Some Problems. Quarterly Journal of Austrian Economics, 7(1).

Batstone D. J., Keller J., Angelidaki I., Kalyuzhnyi S., Pavlostathis S. G., Rozzi A., Sanders W., Siegrist H., Vavilin V. (2002): IWA Task Group on Modelling of Anaerobic Digestion Processes. *Anaerobic Digestion Model No. 1 (ADM1)*. IWA Publishing, London

Batstone D. J., Keller J., Angelidaki I., Kalyuzhnyi S. V., Pavlostathis S. G. (2002): The IWA Anaerobic Digestion Model No 1 (ADM1). Water Sci Technol 45: 65-73.

Batstone D. J., Keller J. (2003): Industrial applications of the IWA anaerobic digestion model No. 1 (ADM1). Water Sci Technol 47: 199-206.

Batstone D. J., Keller J., Steyer J. P. (2006): A review of ADM1 extensions, applications, and analysis: 2002-2005. Water Sci Technol 54: 1-10.

Bauman R. (2007): Microbiology with Diseases by Taxonomy. Pearson Benjamin Cummings.

Beck M. B. (1980): Model structure identification from experimental data. Research report/International Institute for Applied Systems Analysis, no. 80-4.

Bemporad A., Morari M., Ricker N. L (2012): Model Predictive Control Toolbox: User's Guide; The Math Works, Inc., Natick, 2012b.

Bernard O., Hadj-Sadok Z., Dochain D. (2000): Software sensors to monitor the dynamics of microbial communities: application to anaerobic digestion. Acta Biotheoretica 48, 197–205.

Bernard O., Pollt M., Hadj-Sadok Z., Pengov M., Dochain D., Estaben M., Labat P. (2001): Advanced monitoring and control of anaerobic wastewater treatment plants: software sensors and controllers for an anaerobic digester. Water Science and Technology 43, 175–182.

Bernard B. S. (2012): Analysis and Troubleshooting to put a Digester back on Track. BioCycle St. Louise, Missouri October 30<sup>th</sup> 2012. Available online at: http://www.biocyclerefor.com/2012/presents\_tuesday/Sheff\_2\_s.pdf

Bien J. B., Malina G., Bien J. D., Wolny L. (2004): Enhancing anaerobic fermentation of sewage sludge for increasing biogas generation, J. Environ. Sci. Health Part A-Toxic/Hazard. Subst. Environ. Eng. 39 (4) (2004) 939–949.

Bilitewski B., Härdtle G., Marek K., (1997): Waste Management. Springer, Berlin, ISBN: 3-540-59210-5.

Bishnoi P. (2012): Effects of thermal hydrolysis pretreatment on anaerobic digestion of sludge. Copyright © 2012, Play Bishnoi.

Blumensaat F., Keller J. (2005): Modelling of two-stage anaerobic digestion using the IWA Anaerobic Digestion Model No. 1 (ADM1). Water Res 39: 171-183.

Boadi K. O., Kuitunen M. (2005): Environmental and health impacts of household solid waste handling and disposal practices in third world cities: the case of the Accra Metropolitan Area, Ghana. J. Environ Health. 2005 Nov;68(4):32-6.

Boghossian A., Brown J., Zak S. (2006): MIMO using Neural Network. Available online at: https://controls.engin.umich.edu/wiki/index.php/NN

Bohn J. P. (2010): Food Waste Diversion and Utilisation in Humboldt County. 2010.

Bouallagui H., Touhami Y., Cheikh R. B. and Hamdi M. (2005): Bioreactor performance in anaerobic digestion of fruit and vegetable wastes, Process Biochemistry 40, pp. 989–995.

Bougrier C., Carrère H., Delgenès J. P. (2005): Solubilisation of waste-activated sludge by ultrasonic treatment, Chem. Eng. J. 106 (2) (2005) 163–169.

Boyle G., Ed. (1996): Renewable Energy, Power for a Sustainable Future. The Open University, Oxford Press, London.

Braguglia C. M., Mininni G., Gianico A. (2008): Is sonication effective to improve biogas production and solids reduction in excess sludge digestion? Water Sci. Technol. 57 (4) (2008) 479–483.

Braun R., Wellinger A. (2009): Potential of Co-digestion. Internationa Energy Agency (IEA) Bioenergy.

Braun R. (2002): Potential of Co-digestion. Available online at: http://www.novaenergie.ch/iea-bioenergytask37/Dokumente/final.PDF

Brebbia C. A. (2010): The Sustainable World. ISBN: 1743-3541 (online); Copyright @ WIT Press 2011.

Bree M. A., Dhurjati P., Georghegan R. F., Robnett (1988): Kinetic modelling of hybridoma cell growth and immunoglobulin production in a large-scale suspension culture. Biotechnol Bioeng. 1988;32:1067–1072.

Brown, T. L., LeMay H. E., Bursten B. E. (2009): Chemical Kinetics. *Chemistry: The Central Science*. Upper Saddle River, NJ: Pearson Prentice Hall, 2009. 608-12. Print.

Bruun S., Jensen L. S., Vu T. K. V., Sommer S. G. (2014): Small-scale household biogas digesters: An option for global warming mitigation or a potential climate bomb? Renew Sust Energy Rev. 2014;33:736–741.

Buclet N., Godard O. (eds.) (2000): European Policy between Harmonisation and

*Subsidiarity :* The Evolution of Municipal Waste Management in Europe: How Different are National Regimes. Dordrecht : Kluwer Academic Publishers, Coll. Environment and Management.

Budde, J., Quinones S. T., Plöchl M., Heiermann M. (2008): Methods of pre-treatment of less fermentable material and their applicability on anaerobic digestion. In *Proc. International Conference on Agricultural Engineering & Industry Exhibition*, Crete (Greece). CD-version 1130243; 20 p.

Brus L. (2005): Nonlinear Identification of an Anaerobic Digestion Process. Proceedings of IEEE International Conference on Control Applications, Toronto, Canada, Aug. 2005, pp. 137-142.

Burton C. H., Turner C. (2003): Manure Management Treatment Strategies for Sustainable Agriculture, second ed. Silsoe Research Institute.

Buswell A. M. and Hatfield, W.D. (1930): State Water Survey, Bulletin No. 29, State of Illinois, Urbana, IL.

Buswell A. and M., Neave, S. L. (1930): Laboratory studies on sludge digestion III, State Water Survey. Bull. 1930, *30*, 1–84.

Buswell, A. M. and Sollo, F. W. (1948): Am. Chem. Soc. J. 70:1778. Butler, L. (1960) Phytopathol. 55:1262-1268.

Buswell A. M. and Mueller H. F. (1952): Mechanism of methane fermentation, *Ind. Eng. Chem.* 1952, 44, 550–552.

Çalli B. (2012): ENVE 424. Anaerobic Treatment, chapter 8. Marmara University, Department of Environmental Engineering. Available online at: http://mebig.marmara.edu.tr/Enve424/Chapter1.pdf - accessed on 05/11/12

Cannon M. (2016): Nonlinear Systems: Introduction and Concepts of Stability. Hilary Team 2016; University of Oxford. Available online at: http://www.eng.ox.ac.uk/~conmrc/nlc/nlc\_lec1.pdf

Caputo A. C., Palumbo M., Pelagagge P. M., Scacchia F. (2005): Economics of biomass energy utilization in combustion and gasification plants: effects of logistic variables. Biomass and Bioenergy. Volume 28, Issue 1, January 2005, Pages 35-51.

Carrèrea H., Dumasa C., Battimellia A., Batstoneb D. J, Delgenèsa J. P., Steyera, J .P; Ferrer I. (2010): Pre-treatment methods to improve sludge anaerobic degradability: A review. Journal of Hazardous Materials 183 (2010) 1–15.

Castillo E. F. M., Cristancho D. E., Arellano V. A. (2006): Study of the operational conditions for anaerobic digestion of urban solid wastes. Waste Manage. 26, 546–556.

Chae K. J., Jang A., Yim S. K., Kim I. S. (2008): The effects of digestion temperature and temperature shock on the biogas yields from the mesophilic anaerobic digestion of swine manure. *Bioresource Technol* 2008, 99:1-6.

Chai T. and Draxler R. R. (2014): Root mean square error (RMSE) or mean absolute error (MAE)? – Arguments against avoiding RMSE in the literature. Geosci. Model Dev., 7, 1247-1250, 2014; doi: 10.5194/gmd-7-1247-2014. Available online at: www.geosci-model-dev.net/7/1247/2014.

Chen Y., Rößler B., Zielonka S., Wonneberger A. M., Lemmer A. (2014): Effects of organic loading rate on the performance of a pressurized anaerobic filter in two-phase anaerobic digestion. *Energies* 2014, *7*, 736-750; doi:10.3390/en7020736.

Chen Y. R., Hashimoto A. G. (1978): Kinetics of methane fermentation. Biotech Bioeng Symp. 1978;8:269–282.

Cheremisinoff N., Rosenfield P., Davletshin A. (2008): Responsible Care: Anew Strategy for Pollution Prevention and Waste Reduction through Environmental Management. ISBN-10: 1-933762-16-0. Copyright @ 2008 by Gulf Publishing Company.

Choo H. P., Sahlan S., Wahab N. A. (2012): System Identification of Activated Sludge Process. 2012 IEEE Conference on Control, Systems and Industrial Informatics (ICCSII) Bandung, Indonesia, September 23-26, 2012.

Choorit W., Wisarnwan P. (2007): Effect of temperature on the anaerobic digestion of palm oil mill effluent. Electron J Biotechn 2007, 10(No.3, Issue of July 15):376.

Christie A. M., Glimm J., Grove W. J., Higdon D. M., Sharp D. H., Wood-Schultz M. M. (2005): Error Analysis and Simulation of Complex Phenomena. Los Alomos Science; Number 29 2005.

Chu C. P., Lee D. J., Chang B. V., You C. S., Tay J. H. (2002) "Weak" ultrasonic pretreatment on anaerobic digestion of flocculated activated biosolids, Water Res. 36 (11) (2002) 2681–2688.

Cioabla A. E., Ionel I., Dumitrel G., Popescu F. (2012): Comparative study on factors affecting anaerobic bigestion of agricultural vegetal residues. Biotechnology for Biofuels 2012, 5:39. DOI: 10. 1186/1754-6834-5-39. Available online at: http://www.biotechnologyforbiofuels.com/content/5/1/39

Ciubota-Rosie C., Gavrilescu M., and Macoveanu M. (2008): BIOMASS – AN IMPORTANT RENEWABLE SOURCE OF ENERGY IN ROMANIA. Environmental Engineering and Management Journal. September/October 2008, Vol.7, No.5, 559-568. Available online at: http://omicron.ch.tuiasi.ro/EEMJ/

Columbia Electronic Encyclopaedia (CEE), 6th ed. Copyright © 2012, Columbia University Press. All rights reserved. Available online at: http://www.infoplease.com/encyclopedia/science/viscosity-the-velocity-gradient.html#ixz23Fr4uY1Hh

Copot D., Ionescu C. M., Vannecke T. P. W., Volcke E. I. P., De Keyser R. (2013): An Approach towards Multivariable Control of Anaerobic Digeastion using the EPSAC Preditive Control Strategy. 11<sup>th</sup> IWA Conference on Instrumentation Control and Automation. Narbonne, France, 18 – 20 September 2013. Poster Session. Cordaro M. (2008): Understanding Base Load Power. October 7, 2008. Published by Dr. Matthew in conjunction with New York Affordable Reliable Electricity Alliance (New York AREA) pg. 2.

Cornell M., Banks C. J., Heaven S. (2012): Effect of increasing the organic loading rate on the co-digestion and mono-digestion of cattle slurry and maize. *Water Science & Technology* 66, no. 11 (2012): 2336-2342. doi:10.2166/wst.2012.459. Available online at: http://www.iwaponline.com/wst/06611/wst066112336.htm

Cu T. T. T., Pham H. C., Le T. H., Nguyen V. C., Le X. A., Nguyen X. T., Sommer S. G. (2012): Manure management practices on biogas and non-biogas pig farms in developing countries - using livestock farms in Vietnam as an example. J Clean Prod. 2012;27:64–71.

Cuff P. (2012): Lecture 3 - Signal and Systems. Princeton University Fall 2011-2012. Available online at: https://www.princeton.edu/~cuff/ele301/files/lecture3\_2.pdf

Cybulski A. (2001): Fine Chemical Manufacture: Technology and Engineering. Moulijn Gulf Professional publishing, 10 Dec 2001; ISBN: 044482202X, 978044822024 – Technology and Engineering – 551 pages

Dalemo M., Sonesson U., Bjorklund A., Mingarini K., Frostll B., Jonsson H., Nybrant T. and Thyselius L., (1997), ORWARE – A simulation model for organic waste handling systems. Part 1: Model description. Resources, Conservation and Recycling 21 (1997) 17 - 37

Davoli E., Fattore E., Paiano V., Colombo A., Palmiotto M., Rossi A. N., Grande M., Fanelli R. (2010): Waste management health risk assessment: a case study of a solid waste landfill in South Italy. *Waste Management* 30, 1608-1613.

de Baere, L. (2000): Anaerobic digestion of solid waste: state-of-the-art. Water science and technology. Vol 41 (3): 283-290.

de Bok F. A. M., Plugge C., Stams A. J. M. (2004): Interspecies electron transfer in methanogenic propionate degrading consortia. Water Res., 38, 1368–1375 (8 pages).

DEFRA (2004): Review of Environmental and Health Effects of Waste Management: Municipal Solid Waste and Similar Wastes. Available online at: http://archive.defra.gov.uk/environment/waste/statistics/documents/health-summary.pdf

DEFRA (2009): Anaerobic digestion – shared goal, February 2009. Available online at: http://www.biogas.org.nz/Publications/Resources/ad-sharedgoals-090217.pdf

Demirbas A. (1998): Yields of oil products from thermochemical biomass conversion processes. Energy Conversion and Management 1998; 39: 685–90.

Demirbas A. (2001): Biomass resource facilities and biomass conversion processing for fuels and chemicals. Energy Conversion and Management 2001; 42:1357–78.

Demirbas A. (2004): Combustion characteristics of different biomass fuels, Progress in Energy and Combustion Science 30, 219–230.

Denholm, P., Hand M., Jackson M., and Ong S. (2009): Land-use requirements of modern wind power plants in the United States. Golden, CO: National Renewable Energy Laboratory.

Desai M, Patel V., Madamwar D. (1994): Effect of temperature and retention time on biomethanation of cheese whey–poultry waste–cattle dung. *Environ Pollut*. 83:311-315.

Deublein D. and Steinhauser (2008): A. *Biogas from Waste and Renewable Resources*; Wiley Online Library: Weinheim, Germany, 2008.

Dlabaja T., Malaťák J. (2013): Optimization of anaerobic fermentation of kitchen waste. Res. Agr. Eng., *59*: 1–8.

Dominiczak M. H. (2007): Flesh and Bones of Metabolism. ELSERVIER MOSBY © 2007, Elservier Limited. All rights reserverd. ISBN-13:978-0-7234-3368-2. Pg. 22-23.

Donoso-Bravo A., Fdz-Polanco M. (2013): Anaerobic co-digestion of sewage sludge and grease trap: Assessment of enzyme addition. Process Biochemistry, Volume 48, Issues 5-6 (2013) 936–940.

EEA (European Environment Agency), 2003. *Europe's Environment: The third assessment*, Environmental Assessment Report No 10. Kopenhagen: EEA

EEA (European Environment Agency), 2009. *Diverting waste from landfill - Effectiveness of waste-management policies in the European Union*. EEA Report No 7/2009 http://www.articlesbase.com/environment-articles/problem-of-india-waste-management-4836058.html#comments

El-Mashad H. M., Zeeman G., Van Loon, Wilko K. P., Bot G. P. A., Lettinga G. (2004): Effect of temperature and temperature fluctuation on thermophilic anaerobic digestion of cattle manure. Bioresource Technology, 95:191-201.

Environment Agency (2010): Renewable Energy Potential for the Water Industry.

Erden G., Filibeli A. (2009); Ultrasonic pre-treatment of biological sludge: consequences for disintegration, anaerobic biodegradability, and filterability, J. Chem. Technol. Biotechnol. 85 (1) (2009) 145–150.

Espinoza-Escalante F. M., Pelayo-Ortız C., Navarro-Corona J., Gonzalez-Garcıa Y., Bories A., Gutierrez-Pulido H. (2009): Anaerobic digestion of the vinasses from the fermentation of Agave tequilana Weber to tequila: The effect of pH, temperature and hydraulic retention time on the production of hydrogen and methane. Journal of Biomass and Bioenergy 33:14–20.

Eun, J. S., Beauchemin K. A, Hong, Bauer M. W. (2006): Exogenous enzymes added to untreated or ammoniated rice straw: Effects on *in vitro* fermentation characteristics and degradability. *Animal Feed Science and Technology*. 131: 87-102.

EU-AGRO-BIOGAS (2008): European Biogas Initiative to improve the yield of agricultural biogas plants. Available online at: http://www.rtd-services.com/euagrobiogas/images/d09.pdf

European Commission (1995), DGXII, Science, Research and Development, JOULE (1995). Externalities of Energy 'ExternE' Project. Report Number EUR 16520 EN, Summary.

European Commission (1997), Energy for the future: Renewable Sources of Energy. White Paper for a Community Strategy and Action Plan. COM(97)599 final (26/11/1997).

Eurostat, (2012): Environment in the EU28. Eurostat news release, 48/2014 – 25 March 2014. Aviable online at: http://epp.eurostat.ec.europa.eu/cache/ITY\_PUBLIC/8-25032014-AP/EN/8-25032014-AP-EN.PDF

Fang H. H. P. (2010): Environmental Anaerobic Technology: Application and Now Development. ISBN: 1848165420, 978184165427; World Scientific, 2010 – Science – 404 pages.

Feng Y., Behrendt J., Wendland C., Otterpohl R. (2006): Implementation of the IWA anaerobic digestion model No.1 (ADM1) for simulating digestion of blackwater from vacuum toilets. Water Sci Technol 53: 253-263.

Fernandes T. V.; Klaasse-Bos G.J.; Zeeman G.; Sanders J. P.M. and van Lier, J.B. (2009): Effects of thermo-chemical pre-treatment on anaerobic biodegradability and hydrolysis of lignocellulosic biomass. Bioresource technology. Vol. 100: 2575-2579.

Fewtrell L. (2012): Municipal Solid Waste and Health. Research report for Regional Visions of Integrated Sustainable Infrastructure Optimised for Neighbourhoods (ReVISIONS), CREH, Aberystwyth University.

FOA (1992): Biogas process for sustainable development. Available online at: http://www.fao.org/docrep/t0541e/T0541E06.htm#Influence of carbonnitrogen ratio on digestion

Fernandes, T. V., Klaasse G. J., Zeeman G., Sanders J. P. M., van Lier J. B. (2009): Effects of thermo-chemical pre-treatment on anaerobic biodegradability and hydrolysis of lignocellulosic biomass. May 2009, Bioresource technology. Volume 100, Issue 9, May 2009, Pages 2575-2579.

Forastiere F., Badaloni C., de Hoogh K., von Kraus M. K., Martuzzi M., Mitis F., Palkovicova L., Porta D., Preiss P., Ranzi A., Perucci C. A., Briggs D. (2011): Health impact assessment of waste management facilities in three European countries. *Environmental Health* 10: 53.

Fridleifsson I. B. (2003), Status of geothermal energy amongst the world's energy sources.

Fridleifsson I. B. (1996), Present Status and Potential of Geothermal Rol in the World. United Nations University Geothermal Training Programme, WREC 1996.

Fridleifsson I. B. (2001): Geothermal energy for the benefit of the people. Renewable and Sustainable Energy Review. Volume 5, Issue 3, September 2001, Pages 229-312.

Frock A. D. and Kelly R. M. (2012): Extreme thermophiles: moving beyond single-enzyme biocataysis. Current Opinion in Chemical Engineering (COCHE) 2012, COCHE-51; 1: 1-10.

Gagel A., Bates D. and Gawell Karl (2007), A Guide to Geothermal Energy and the Environment. Geothermal Energy Association, Updated April 2007.

Galindo E., Ramirez O. R. (2013): Advance in Bioprocess Engineering. Springer Science & Business Media, 2013. ISBN: 940170617, 9789401706414. Science – 542 pages.

Gallert C. and Winter J., (2002), Solid and liquid residues as raw materials for Biotechnology. *Naturwissenschaften*. Vol. 89: 483-496.

Gallert C. and Winter J., (2005). Bacterial metabolism in wastewater treatment systems. In: Environmental biotechnology – Concepts and applications (H.\_J. Jordening and J. Wintereds.). Weinheim: Wiley\_VCH.

Gao W. J., Leung K. T. and Liao B. Q. (2011): Effects of temperature and temperature shock on the performance and microbial community structure of a submerged anaerobic membrane bioreactor. Bioresource Technology 102 (2011) 8733–8740.

Garcíaa M. R., Carlos Vilas C., Santosc L. O., Alonsob A. A. (2012): A robust multi-model predictive controller for distributed parameter systems Journal of Process Control 22 (2012) 60–71.

Gavala H., Angelidaki I., Ahring B. (2003): Kinetics and modeling of anaerobic digestion process. In: Ahring B, Angelidaki I, Macario EC, Gavala HN, Hofman-Bang J, Macario AJI, Elferink SJWHO, Raskin L, Stams AJM, Westermann P, Zheng D, editors. Biomethanation I. Vol. 81. Springer; Berlin Heidelberg, Germany: 2003. pp. 57–93.

Gerardi M. H. (2003): *The Microbiology of Anaerobic Digesters*; Wastewater Microbiology Series. John Wiley & Sons: Chichester, UK, 2003; Volume 3.

Gibbings J. C. (2001): Dimensional Analysis; Springer; ISBN: 1-84996-316-9.

Gijzen, H.J., 2002. Anaerobic digestion for sustainable development: a natural approach. *Wastewater and Technology*. Vol. 45 (10): 321\_328.

Glacken M. W., Huang C., Sinskey A. J. (19890: Mathematical descriptions of hybridoma culture kinetics III. Simulation of fed-batch bioreactors. J Biotechnol. 1989;10:39–66.

Goldberg M. S., Siemiatyck J., DeWar R., Desy M., and Riberdy H. (1999), Risk of developing cancer relative to living near a municipal solid waste site in Montreal, Quebec, Canada. Arch Environ Health. 1999 Jul-Aug; 54(4):291-6.

Goldberg M. S., al-Homsi N, Goulet L and Riberdy H. (1995), Incidence of cancer among persons living near a municipal solid waste site in Montreal, Quebec, Canada. Arch Environ Health. 1995 Nov-Dec;50(6):416-24.

Goldberg M. S., Goulet L, Riberdy H., and Bonvalot Y. (1995), Low birth weight and preterm births among infants born to women living near a municipal solid waste site in Montreal, Quebec, Canada. Arch Environ Health. 1995 Apr; 69(1):37-50.

Greenpeace (2005), Solar Generation: Solar energy for over one billion people and two million jobs by 2020. Available online at: http://www.cap.lmu.de/download/2005/greenpeace-solar-report-2005.pdf

Güçlü D., Yılmaz N., Ozkan-Yucel U. G. (2011): Application of neural network prediction model to full-scale anaerobic sludge digestion. Journal of Chemical Technology and Biotechnology. Volume 86, Issue 5, pages 691-698, May 2011

Hajji A., Rhachi M. (2013): The Influence of Particle Size on the Performance of Anaerobic Digestion of Municipal Solid Waste. Energy Procedia. Volume 36, 2013, Pages 515-520.

Hanaki K.; Matsuo T. and Nagase M. (1981): Mechanism of inhibition caused by longchain fatty acids in anaerobic digestion process. Biotechnology and bioengineering. Vol. 23: 1591-1610.

Hartmann H., Moller H. B., Ahring B. K. (2004), Efficiency of the anaerobic treatment of the organic fraction of municipal solid waste: collection and pre-treatment. Waste Management Research 2004 Feb; 22(1): 35-41.

Hartmann H. and Ahring B. K. (2006): Strategis for the anaerobic digestion of the organic fraction of municipal solid waste: an overview. Water science and technology. Vol. 53 (8): 7-22.

Hayes T. D., Isaacson H. R., Pfeffer J. T., Liu Y. M. (1990): In situ methane enrichment in anaerobic digestion. *Biotechnol. Bioeng.* 1990, *35*, 73–86.

Heams E. C. (2006): Focus on Biotechnology Research. Nova Publisher, 2006; ISBN: 1594548633, 97815948635. Medical/Biotechnology, 246 pages.

Hestnes A.G. (1999), Building integration of solar energy systems. Solar Energy 67,181–187. Jefferson M., (2006), Sustainable energy development: performance and prospects, Renew Energy, 31, 71-82.

Haug R. T., LeBrun T. J., Tortorici L. D. (1983): Thermal pre-treatment of sludges–a field demonstration. J Water Pollut Control Fed (JWPCF), Vol. 55, pp 23–34, 1983.

Hayes T. D., Isaacson H. R., Pfeffer J. T., Liu Y. M. In situ methane enrichment in anaerobic digestion. *Biotechnol. Bioeng.* 1990, *35*, 73–86.

Hills D. J., Nakano K. (1984): Effects of particle size on anaerobic digestion of tomato solid wastes. Agricultural Wastes. Volume 10, Issue 4, 1984, pages 285-295.

Hiremath R. B., Kumar B., Balachandra P., Ravindranath N. H. and Raghunandan B. N. (2009): Decentralised renewable energy: Scope, relevance and applications in the Indian context. *Energy Sustain. Dev.* 2009, *13*, 4–10.

Hoegh-Guldberg O., and Bruno J. F. (2010): The impact of climate change on the world's marine ecosystems. Science 18 June 2010: Vol. 328 no. 5985 pp. 1523-1528. DOI: 10.1126/science. 1189930.

Holm-Nielsen J. B., Seadi T. A., Oleskowicz-Popiel, P. (2009): The future of anaerobic digestion and biogas utilization. *Bioresource Technology* 100 (2009), 5478–5484.

Holubar P., Zani L., Hager M., Froschl W., Radak Z., Braun R. (2002): Advanced controlling of anaerobic digestion by means of hierarchical neural networks. Water Research 36, 2582–2588.

Holubar P., Zani L., Hager M., Frochl W., Radak Z., Braun R. (2003): Start-up and recovery of a biogas-reactor using a hierarchical neural network-based control tool. Journal of Chemical Technology and Biotechnology 78, 847–854.

Hoyle B. D. (2003): Fundamental of Bacteria Biology; Encyclopedia of Espionage, Intelligence, and Security [online]. Available: http://www.faqs.org/espionage/An-Ba/Bacterial-Biology.html#ixz2JYTaYiN2

Hooshyari B., Azimi A. Mehrdadi N. (2009): Kinetic analysis of enhanced biological phosphorus removal in a hybrid integrated fixed film activated sludge process. Int. J. Environ. Sci. Tech. 6 (1), 149-158 (10 Pages).

HPA (2011): Impact on health of emissions from landfill sites. RCE-18. Health Protection Agency, London.

Huber H., Thomm M., Konig H., Thies G., Stetter K.O. (1982): Methanococeus thermolithotrophicus, a novel thermophilic lithotrophic methanogen. Arch. Microbiol. 132, 47–50.

Iacovidou E., Ohandja D-G., Voulvoulis N. (2012): Food waste co-digestion with sewage sludge e Realising its potential in the UK. Journal of Environmental Management 112 (2012) 267-274.

IEO (2013). Report Number: DOE/EIA-0484(2013). Available online at: http://www.eia.gov/forecasts/ieo/pdf/0484(2013).

Japan Agency for Marine-Earth Science and Technology (JAMSTEC) (2007): Exploration of the effects of high hydrostatic pressure on microbial growth, physiology and survival: perspectives from piezophysiology. Biosci Biotechnol Biochem, 2007 Oct:71(10):2347-57. Epub 2007 Oct7.

Jiang Y., Heaven S., Banks C. J. (2012): Strategies for stable anaerobic digestion of vegetable waste. Renewable Energy 44(2012) 206-214.

Johansen A., Nielsen H. B., Hansen C. M., Andreasen C., Carlsgart J., Hauggard-Nielsen H., (2013): Survival of weed seeds and animal parasites as affected by anaerobic digestion at meso- and thermophilic conditions. *Waste Manag.* 33, 807–812. doi:10.1016/j.wasman.2012.11.001.

Jolis D. (2008): High-Solids Anaerobic Digestion of Municipal Sludge Pretreated by Thermal Hydrolysis. *Water Environment Research*, Vol. 80, No. 7, pp. 654-662, 2008.

Johansson T. B., Ed., Laurie Burnham, ex. Ed (1993) 'Renewable energy. Sources for Fuels and Electricity', Island Press.

Kalfas H., Skiadas I. V., Gavala H. N., Stamatelatou K., Lyberatos G. (2006): Application of ADM1 for the simulation of anaerobic digestion of olive pulp under mesophilic and thermophilic conditions. Water Sci Technol 54: 149-156.

Kangle K. M., Kore S. V., Kore V. S., Kulkarni G. S. (2012): Recent Trends in Anaerobic Codigestion: A Review. Universal Journal of Environmental Research and Technology All Rights Reserved Euresian Publication c 2012 eISSN 2249 0256 Available Volume 2, Issue 4: 210-219 Available online at: www.environmentaljournal.org

Kaparaju P., Buendia I., Ellegaard L., and Angelidaki I. (2007): Effects of mixing on methane production during thermophilic anaerobic digestion of manure: Labscaleand pilot-scale studies. Bioresource technology. Vol. 99: 4919–4928.

Karim K., Hoffman R., Klasson K. T., Al-Dahhan, M. H. (2005a): Anaerobic digestion of animal waste: Effect of mode of mixing. *Water Research*, 39, 3597-3606.

Karim K., Hoffman R., Klasson K. T., Al-Dahhan, M. H. (2005b): Anaerobic digestion of animal waste: Waste strength versus impact of mixing. *Bioresource Technology*, 96, 1771-1781.

Karim K., Klasson T., Hoffmann R., Drescher S. R., DePaoli D. W., and Al-Dahhan M.H. (2005c): Anaerobic digestion of animal waste: Effect of mixing. Bioresource technology. Vol. 96: 1607–1612.

Kenealy W. and Zeikus J. G. (1981): Appl. Bacteriol. 146:138- 140. Khaleel, R. Reddy, K.R. and Overcash, M.R. (1980) Water Res. 14:421-436. Kent M. (2000): Advanced Biology. Oxford University Press. ISBN 0-19-914195-9

Kim J., Park C., Kim T. H., Lee M., Kim S., Kim S. W., Lee J. (2003): Effects of various pretreatments for enhanced anaerobic digestion with waste activated sludge. J. Biosci. Bioeng. 95, 271–275.

Khalid A., Arshad M., Anjum M., Mahmood T. and Dawson L. (2011): The Anaerobic digesteion of solid organic waste. Waste Management 31 (2011) 1737-1744.

Khanal S. K. (2008): Anaerobic biotechnology for bioenergy production: principles and applications. Wiley-Blackwell, Ames.

Kheiredine B., Derbal K., Bencheikh-Lehocine M. (2014): Effect of Starting pH on the Produced Methane from Dairy Wastewater in Thermophilic Phase. Chemical Engineering Transactions. A publication of The Italian Association of Chemical Engineering. Guest Editors: Enrico Bardone, Marco Bravi, Taj Keshavarz. Copyright © 2014, AIDIC Servizi S.r.l., ISBN 978-88-95608-29-7; ISSN 2283-9216.

Knudsen M. (2004): Experimental modelling of dynamic systems. Department of Control Engineering, Aalborg University. 2004. Available online at: http://www.control.auc.dk/~pa/kurser/PR6model/LectureNote02pdf.pdf Koppar A. and Pullammanappallil P. (2008): Single-stage, batch, leach-bed, thermophilic anaerobic digestion of spent sugar beet pulp. *Bioresources technology*. Vol. 99 (8): 2831-2839

Koqak U., Palmerius K. L., Cooper M. (2012): An Error Analysis Model for Adaptive Simulation. C-Research, Linkoping University, ITN Norrkoping, SWEDEN.

Korres N., O'Kiely P., Benzie J. A. H., West J. S. (2013): Bioenergy production by anaerobic digestion: Using agricultural biomass and organic waste. Routledge, 29 Aug 2013 – Technology & Engineering – 472 pages.

Kossmann W., Pönitz U., Habermehl S. (1997): Biogas digest: Biogas application and product development (Volume II) Information and Advisory Service on Appropriate Technology (ISAT) & GATE in Deutsche Gesellschaft für Technische Zusammenarbeit (GTZ) 1997.

Kozani S. J. (2014): Basics of the biogas production process. Available online at http://bee-res.gr/wp-content/uploads/2014/06/Kozani\_Workshop\_SJunne.pdf

Kramarz F. and Visser M. (2012): The linear regression model. Lectures 1 and 2. Available online at: http://catalogue.polytechnique.fr/site.php?id=289&fileid=4318

Krishna K. M., Verma A. K., Srivdya A., Papic L. (2010): INTEGRATION OF BLACK-BOX AND WHITE-BOX MODELING APPROACHES FOR SOFTWARE RELIABILITY ESTIMATION. International Journal of Reliability, Quality and Safety Engineering; Volume 17, Issue 03, June 2010.

Krustok I., Nehrenheim E., Odlare M., Liu X., Li S. (2013): Cultivation of Indigenous Algea for Increased Biogas Production. International Conference on Applied Energy; ICAE 2013, Jul 1-4, 2013, Pretoria, South Africa; Paper ID: ICAE2013-126.

Kumar J. L. G., Zhao Y. Q. (2011): A review on numerous modelling approaches for effective, economical and ecological treatment wetlands. J Environ Manage 92:400–40

Kumar A., Kumar A., Kumar R., Sudarson J. S (2010): Biogas manure (BGM) from mixed kitchen waste: A trail study. Journal of Environmental Research and Development, Vol. 5 No. 1, July –September 2010.

Kumar S. (2010): Performance study for anaerobic digestion of municipal solid waste in a single phase reactor. *Intl. J Envirn Pollut* 2010, 43(Nos. 1/2/3):16. Kumar S. (2012): Biogas. Croatia. ISBN 978-953-51-0204-5. Available online at www.intechopen.com

Kwon H. and Choi S. (2015): A tended Kriging model with R<sup>2</sup> indicator and design optimisation. Aerospace Science and Technology. Volume 43, June 2015, doi:10.1016/j.ast.2015.02.021; Pages 111-125.

Labatut R. A. and Gooch C. A. (2012): Monitoring of anaerobic digestion process to optimise performance and prevent system failure. Department of Biological and Environmental

Engineering, Cornell University, Ithaca, NY. Available online at: http://ansci1.abc.cornell.edu/prodairy/gotmanure/2012proceedings/21.Rodrigo.Labatut.pdf

Lee D. H., Behera S. K., Kim J., Park H. S. (2009b). Methane production potential of leachate generated from Korean food waste recycling facilities: a lab scale study. Waste Manage. 29, 876–882.

Lee D., Naseradinmousavi P., Nataraj C. (2012): Nonlinear Dynamic Model-Based Adaptive Control of a Solenoid-Valve System. Journal of Control Science and Engineering. Volume 2012 (2012), Article ID 846458, 13 pages. http://dx.doi.org/10.1155/2012/846458

Lettinga G., van Lier J.B., van Buuren J.C.L. and Zeemanpollution (2001): control and the role of anaerobic digestion, *Water Sci. Technol.* 44 (6) (2001), pp. 181–188.

Lindeboom R. E. F., Fermoso F. G., Weijma J., Zagt K., van Lier J. B. (2011): Autogenerative high pressure digestion: Anaerobic digestion and biogas upgrading in a single step reactor system. *Water Sci. Technol.* 2011, *64*, 647–653.

Lindmark J. (2014): The effects of different mixing intensities during anaerobic digestion of the organic fraction of municipal solid waste. Waste Management (Impact Factor: 3.16). 05/2014; DOI: 10.1016) wasman 2014.04.006 Sorce Pubmed.

Linke B., Heiermann M., Mumme J. (2006): Results of monitoring the pilot plants Pirow and Clausnitz. In: Rohstoffe, F.N. (Ed.), Solid-State Digestion–State of the Art and Further R&D Requirements, vol. 24. Gülzower Fachgespräche, pp. 112–130.

Li Y. F., Nelson M. C., Chen P. H., Graf J., Li Y., Yu Z. (2014): Comparison of the microbial communities in solid-state anaerobic digestion (SS-AD) reactors operated at mesophilic and thermophilic temperatures. Appl Microbiol Biotechnol. 2014 Sep. 7. Available online at: http://www.ncbi.nlm.nih.gov/pubmed/25194839

Lissens G., Vandevivere P., De Baere L., Biey E. M. and Verstraete W. (2001): Solid waste digestors: biogas production and practice for municipal solid waste digestion. *Water science and technology*. Vol 44 (8): 91-102.

Liu D. W., Liu D. P., Zeng R. J. and Angelidaki I. (2006): Hydrogen and methane production from household solid waste in the two-stage fermentation process, Water Research 40 (2006a), pp. 2230–2236.

Liu G. T., Peng X. Y., Long, T.R., (2006b): Advance in high-solid anaerobic digestion of organic fraction of municipal solid waste. Journal of Central South University of Technology 13, 151–157. Luenberger, D.G., 1966.

Liu R. R., Tian Q., Yang B., Chen J. H. (2009): Hybrid anaerobic baffled reactor for treatment of desizing wastewater. Int. J. Environ. Sci. Tech., 7 (1), 111-118 (8 pages).

Liu Y. H., Bi J. X., Zeng A. P., Yuan J. Q. (2008): A simple kinetic model for myeloma cell culture with consideration of lysine limitation. Bioprocess Biosyst Eng. 2008;31:569–577.

 $\begin{array}{c|c} Ljung \ L. \ (2015): \ System \ Identification \ Toolbox^{TM} - Getting \ Started \ Guide. \ @ \ copyright \ 1988-2015 \ by \ The \ Mathworks, \ Inc. \ Available \ online \ at: \ http://uk.mathworks.com/help/pdf_doc/ident/ident_gs.pdf \end{array}$ 

Ljung L. (2010): "Perspectives on System Identification," *Annual Reviews in Control*, vol. 34, no. 1, pp. 1–12.

 $\begin{array}{c|c} Ljung \ L. \ (2008): \ System \ Identification \ Toolbox^{TM} - Getting \ Started \ Guide. \ @ \ copyright \ 1988-2008 \ by \ The \ Mathworks, \ Inc. \ Available \ online \ at: \ http://uk.mathworks.com/help/pdf_doc/ident/ident_gs.pdf \end{array}$ 

Ljung L. (2001): Black-box Model from Input-output Measurement. IEEE Instrumentation and Measurement Technology Conference Budapost, Hungary, May 21-23, 2001.

Lu J., Ahring B. K. (2005): Effects of temperature and hydraulic retention time on thermophilic anaerobic pre-treatment of sewage sludge. ADSW 2005 Conference Proceedings – Volume 1 – Session 3a: Collection & pre-treatment.

Luo K., Yang Q., Li X., Yang G., Liu Y., Wang D., Zheng W., Zeng G. (2012): Hydrolysis kinetics in anaerobic digestion of waste activated sludge enhanced by  $\alpha$ -amylase. Biochemical Engineering Journal 62 (2012) 17–21.

Luostarinen S. A. (2005): Anaerobic on-site wastewater treatment at low temperature. Department of Biological I and Environmental Science, University of Jyvaskyla, Finland. 85p.

Lusk P. (1999) Latest Progress in Anaerobic Digestion. Biocycle, 40.

Maass B. (2011): The application of black-box models to combustion processes in the internal combustion engine. Copyright © Bastian Maass; 2011. Available online at: https://dspace.lboro.ac.uk/2134/8490

Ma J., Yu L., Frear C., Zhao Q., Li X., Chen S. (2013): Kinetics of psychrophilic anaerobic sequencing batch reactor treating flushed dairy manure. Bioresour Technol. 2013 Mar; 131:6-12.

Macias-Corral M., Samani Z., Hanson A., Smith G., Funk P., Yu H. (2008): Anaerobic digestion of municipal solid waste and agricultural waste and the effect of codigestion with dairy cow manure. Biores. Technol. 99, 8288–8293.

Manyi-Loh C. E., Mamphweli S. N., Meyer E. L., Okoh A. I., Makaka G., Simon M. (2013): Microbial Anaerobic Digestion (Bio-Digesters) as an Approach to the Decontamination of Animal Wastes in Pollution Control and the Generation of Renewable Energy. Int J Environ Res Public Health. Sep 2013; 10(9): 4390–4417 doi: 10.3390/ijerph10094390.

Marchaim U. (1992): Biogas processes for sustainable development. FAO 1992, Coperate Document Repository. Available online at: http://agrienvarchive.ca/bioenergy/download/Marchaim\_biogas\_processes\_fao\_1992.pdf

Mata-*Alvarez* J., Mace S., Llabres, P. (2000): Anaerobic digestion of organic solid wastes: an overview of research achievements and perspectives. *Bioresource technology*. Vol. 74: 3 – 16.

Mata-Alvarez, J., (2002): Fundamentals of the anaerobic digestion process (in: Biomethanisation of organic fraction of municipal solid wastes. Editor: Mata-Alvarez, J.). Amsterdam: IWA publishing company.

Mavropoulos A. (2013): The future of waste management in an overcrowded planet. Waste Management 2030+. *Waste Management World*. Volume 11, Issue 2. Available online at: http://www.waste-management-world.com/articles/print/volume-11/issue-2/features/waste-management-2030.html

Mbachu H. I., Nduka E. C., Nja M. E. (2012): Designing a Pseudo R-Squared Goodness-of-Fit Measure in Generalised Linear Models. Journal of Mathematics Research. Vol 4, No. 2; April 2012. Available online at:

http://www.ccsenet.org/journal/index.php/jmr/article/viewFile/16073/10827

McCabe J., Eckenfelder W. eds. (1957): *Biological Treatment of Sewage and Industrial Wastes*. Two volumes. New York: Reinbold Publishing.

McCarty P. L., Smith D. P. (1986): Anaerobic Wastewater treatment. *Environ. Sci. Technol.*, Vol. 20, No. 12, 1986.

McInerney M. J., Bryant M. P., Hespell R. B., and Costerton J. W. (1981): Syntrophomonas wolfei gen. nov. sp. nov., an anaerobic, syntrophic, fatty acid-oxidizing bacterium. Appl. Environ. Microbiol. 41:1029–1039.

McKendry P. (2002): Energy production from biomass (part 2): conversion technologies. Bioresource Technology, Volume 83, Issue 1, May 2002, Pages 47-54.

Melosi M. (2005): Garbage in the Cities: Refuse, Reform and the Environment. Pittsburgh:University of Pittsburgh Press; 2005.

Meroney R. N. and Colorado P. E. (2009): CFD simulation of mechanical draft tube mixing in anaerobic digester tanks. Water research. Vol. 43: 1040–1050.

Meynell P-J. (1976): Methane: Planning a Digester. New York: Schocken Books. pp. 3.

Michel J., Dunagan H., Boring C., Healy E., Evans W., Dean J., McGillis A., Hain J. (2007): Worldwide Synthesis and Analysis of Existing Information Regarding Environmental Effects of Alternative Energy Uses on the Outer Continental Shelf. MMS 2007-038. Prepared by Research Planning and ICF International. Herndon, VA: U.S. Department of the Interior, Minerals Management Service.

Montgomery F. R and Bochmann G. (2014): Pre-treatment of feedstock for enhanced biogas production: Technical Brochure. ISBN 978-1-910154-052 (electronic version); Copyright @ 2014 1EA Bioenergy.

Morari M., Lee J. H., Garc´ıa C. E. (2002): MODEL PREDICTIVE CONTROL. March 15, 2002. Available online at: http://www.kasr.elf.stuba.sk/predmety/pmr/Ucebnice%20MPC%20,vyznamne%20clanky,%2 0manualy,%20programy%20k%20MPC/knihaucebmpc.pdf

Moreira A. R., Wallace K. K. (2012): Computer and Information Science Application in Bioprocess Engineering. Vaolume 305 of Nato Science SERIES e. Springer Science &Business Media 2012. ISBN: 9400901771, 978940090177; 477 pages.

Morita R. (1975): Psychrophilic Bacteria. Bacteriological Reviews (1975); 39(2):144-167.

Mosier N., Wyman C., Dale B., Elander R., Lee Y. Y, Holtzapple M. and Ladisch M. (2005): Features of promising technologies for pre-treatment of lignocellulosic biomass. Bioresource technology. Vol. 96 (6): 673-686.

Motte J. C., Escudie R., Bernet N., Delgenes J. P., Steyer J. P., Dumas C. (2013): Dynamic effect of total solid content, low substrate/inoculum ratio and particle size on solid-state anaerobic digestion. Bioresource Technology. Volume 144, September 2013, Pages 141-148.

Myung J. (2000): The importance of complexity in model selection. Journal of Mathematical Psychology, 44(1), 190-204.

Najafpour G. (2006): Biochemical Engineering and Biotechnology. Elsevier, 2006. ISBN: 0080468020, 9780080468020; 438 pages.

Nasir A., Bala K. C., Mohammed S. N., Mohammed A., Umar I. (2015): Experimental Investigation on the Effects of Digester Size on Biogas Production from Cow Dung. American Journal of Engineering Research (AJER); e-ISSN : 2320-0847 p-ISSN : 2320-0936.;Volume-04, Issue-01, pp-181-186. Available online at: http://www.ajer.org/papers/v4(01)/X040101810186.pdf

National Instrument (NI) (2010): Selecting a Model Structure in the System Identification Process. Published Date: Jun 17, 2010. Available online at: http://www.ni.com/white-paper/4028/en/

National Instrument (2009): Improving PID controller performance. Published Date: Dec. 03, 2009. Available online at: http://www.ni.com/white-paper/7438/en/

National Non-Food Crop Centre (NNFCC) (2011): Renewable Fuels and Energy Factsheet: Anaerobic Digestion. Available online at: http://www.nnfcc.co.uk/publications/nnfccrenewable-fuels-and-energy-factsheet-anaerobic-digestion

National Renewable Energy Laboratory (NREL). 2012. Renewable Electricity Futures Study. Hand, M.M.; Baldwin, S.; DeMeo, E.; Reilly, J.M.; Mai, T.; Arent, D.; Porro, G.; Meshek, M.; Sandor, D. eds. 4 vols. NREL/TP-6A20-52409. Golden, CO: National Renewable Energy Laboratory.

National Wind Coordinating Committee (NWCC) (2010): Wind turbine interactions with birds, bats, and their habitats: A summary of research results and priority questions.

Nayono S. E., Gallert C., Winter J. (2010): Co-digestion of press water and food waste in a biowaste digester for improvement of biogas production. Bioresour. Technol. 101, 6987–6993.

Nayono S. E. (2010): Anaerobic digestion of organic solid waste for energy production. KIT Scientific Publishing 2010; ISBN: 978-3-86644-464-5.

Neis U., Nickel K., Lunden A. (2008): Improving anaerobic and aerobic degradationby ultrasonic disintegration of biomass, J. Environ. Sci. Health Part A-Toxic/Hazard. Subst. Environ. Eng. 43 (13) (2008) 1541–1545.

Neves L.; Goncalo E.; Oliveira R., Alves M. M. (2008): Influence of composition on the biomethanation potential of restaurant waste at mesophilic temperatures. Waste management. Vol. 28: 965-972.

Nielsen H. B., Mladenovska Z., Westermann P., Ahring B.K. (2004): Comparison of twostage thermophilic (68 °C/55 °C) anaerobic digestion with one-stage thermophilic (55 °C) digestion of cattle manure, Biotechnology and Bioengineering 86 (2004), pp. 291–300.

Nightingale F. (1954): *Selected Writings of Florence Nightingale*, ed. Lucy Ridgely Seymer (New York: The Macmillan Co., 1954), pp. 38287.

NIST/SEMATECH (2012): Engineering Statistics Handbook. *e-Handbook of Statistical Methods*Available online at: http://www.itl.nist.gov/div898/handbook/pmd/section4/pmd44.htm

Noike T., Endo G., Chang J., Yaguchi J., Matsumoto J. (1985): Characteristics of Carbohydrate Degradation and the Rate-limiting Step in Anaerobic Digestion. *Biotechnology and Bioengineering*. 27: 1482-1489.

Norton J. M, Wing S., Lipscomb H. J., Marshall S. W., and Cravey A. J. (2007), Race, Wealth, and Solid Waste Facilities in North Carolina. Environ Health Perspect. 2007 September; 115(9): 1344–1350.

NREL (2010): Brownfields' Bright Spot: Solar and Wind Energy. Available Online at: http://www.nrel.gov/news/features/feature\_detail.cfm/feature\_id=1530

NRC (2010), Advancing the Science of Climate Change. National Research Council. The National Academies Press, Washington, DC, USA. Available online at: http://dels.nas.edu/resources/static-assets/materials-based-on-reports/reports-in-brief/Science-Report-Brief-final.pdf

Nwuche C. O., Ugoji E. O. (2010): Effect of co-existing plant species on soil microbial activity under heavy metal stress. Int. J. Environ. Sci. Tech., 7 (4), 697-704 (8 pages).

Oberkampf W. L., DeLand S. M., Rutherford B. M., Diegert K. V., Alvain K. F. (2002): Error and uncertainty in modelling and simulation. Reliability Engineering and System Safety 75 (2002) 333-357.

OECD/IEA (1998): Benign energy? The environmental implications of renewable international energy agency, www. iea.org.Geothermics 32 (2003) 379–388.

OECD (2004): Towards waste prevention performance indicators. Available online at: http://search.oecd.org/officialdocuments/displaydocumentpdf/?doclanguage=en&cote=env/ep oc/wgwpr/se(2004)1/final

OECD (2010): Bioheat, Biopower and Bioga Development and Implications for Agriculture. ISBN 978-92-64-08586-2 (PDF), DOI 10.1787/9789264085862-em. Copyright @ 2010

OECD (2011): GREEN GROWTH STUDIES: ENERGY OECD 2011 Available online t: http://www.oecd.org/greengrowth/greening-energy/49157219.pdf

OECD (Organisation for economic co-operation and development), 2013. Work on Sustainable Materials & Waste Management. Available online at: http://www.oecd.org/env/waste/OECD%20Work%20on%20SMM\_update%2011-02-2013.pdf

Omstead D. R., Jeffries T. W., Naughton R. and Harry P. (1980) Membrane -Controlled Digestion: Anaerobic Production of Methane and Organic Acids, Biotechnology and Bioengineering Symposium No 10, 247-258.

Onyeche T. I., Schlafer O., Bormann H., Schroder C., Sievers M. (2002): Ultrasonic cell disruption of stabilised sludge with subsequent anaerobic digestion, Ultrasonics 40 (2002) 31–35.

Oppng G., McEwan M. I., Montague G. A., Martin E. B. (2013): Towards Advanced Control for Anaerobic Digesters: Sludge Inventory Optimisation. 13<sup>th</sup> World Congress on Anaerobic Digestion -IWA. Santiago De Compostola, Spain, 25 – 28 June 2013. Poster Session.

Ostrem, K. (2004): Greening Waste: Anaerobic Digestion For Treating The Organic Fraction Of Municipal Solid Wastes. Earth Engineering Center Columbia University.

Ouedraogo A. (1999): Pilot scale two-phase anaerobic digestion of the biodegradable organic fraction of Bamako district municipal solid waste. In *II Int. Symp. Anaerobic Dig. Solid Waste*, held in Barcelona, June 15-17, 1999 (eds. J. Mata-Alvarez, A. Tilche and F. Cecchi), vol. 2, pp. 73-76, Int. Assoc. Wat. Qual.

Overend R. P. (1998), Biomass gasification: a growing business. Renewable Energy World 1998; 1: 59–63.

Palmowski L. M., Müller J.A. (2000): Influence of the size reduction of organic waste on their anaerobic digestion. Water Sci Technol. 2000; 41(3): 155-62.

Pandey P. K. and Michelle L. Soupir M. L. (2012): Impacts of Temperatures on Biogas Production in Dairy Manure Anaerobic Digestion. *IACSIT International Journal of Engineering and Technology*, Vol. 4, No. 5, October 2012. Available online at: http://www.ijetch.org/papers/448-S20011.pdf Parawira W., Murto M., Zvauya R., Mattiasson B. (2004): Anaerobic batchdigestion of solid potato waste alone and in combination with sugar beet leaves. Renew. Energy 29, 1811–1823.

Parker W. J. (2005): Application of the ADM1 model to advanced anaerobic digestion. Bioresour Technol 96: 1832-1842.

Perendeci A., Arslan S., Çelebic S. S., Tanyolaç A. (2008): Prediction of effluent quality of an anaerobic treatment plant under unsteady state through ANFIS modeling with on-line input variables. Chem Eng J 145: 78-85.

Perez-Elvira S., Fdz-Polanco M., Plaza F. I., Garralon G., Fdz-Polanco F. (2009) Ultrasound pre-treatment for anaerobic digestion improvement, Water Sci. Technol. 60 (6) (2009) 1525–1532.

Petr H. (2010) "Fuzzy Logic", *The Stanford Encyclopedia of Philosophy* (Fall 2010 Edition), Edward N. Zalta (ed.). Available online at: <a href="http://plato.stanford.edu/archives/fall2010/entries/logic-fuzzy/">http://plato.stanford.edu/archives/fall2010/entries/logic-fuzzy/</a>.

Pfeffer J. T. (1974): Temperature Effects on Anaerobic Fermentation of Domestic Refuse. Biotechnology and Bioengineering, Vol. 16, 771 – 787.

Pham C. H., Vu C. C., Sommer S. G., Bruun S. (2014): Factors Affecting Process Temperature and Biogas Production in Small-scale Rural Biogas Digesters in Winter in Northern Vietnam. Asian-Australas J Anim Sci. Jul 2014; 27(7): 1050–1056. Doi: 10.5713/ajas.2013.13534. PMCID: PMC4093558. Available online at: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC4093558/

Pillonetto, G.; Tianshi Chen; Ljung, L. "Kernel-based model order selection for identification and prediction of linear dynamic systems", *Decision and Control (CDC)*, 2013 IEEE 52nd Annual Conference on, On page(s): 5174 – 5179.

Ploechl, M., Hilse, A., Heiermann, M., Quinones, T. S., Budde, J., and Prochnow, A. (2010): Application of hydrolytic enzymes for improving biogas feedstock fluidity. Agricultural Engineering International, 11:1529.

Polit M., Genovesi A., Claudet B. (2001): Fuzzy logic observers for a biological wastewater treatment process. Applied Numerical Mathematics 39, 173–180.

Polprasert C, Van der Steen NP, Veenstra S, and Gijzen HJ, 2001. *Wastewater Treatment II: Natural System for Wastewater Management*. Delft: International Institute for Infrastructure, Hydraulics and Environmental Engineering (IHE Delft).

Premier G. C., Dinsdale R., Guwy A. J., Hawkes F. R., Hawkes D. L., Wilcox S. J. (1997): Simple black box models predicting potential control parameters during disturbances to a fluidised bed anaerobic reactor. Proc 8<sup>th</sup> International Conf. on Anaerobic Digestion, May 25-29, Sendai, Japan, Vol 1, 189-196.

Pullammanappallil P. C., Svoronos S. A., Chynoweth D. P., Lyberatos G. (1998): Expert system for control of anaerobic digesters. Biotechnology and Bioengineering 58, 13–22.

Quiñones, T. S., Plöchl, M., Budde, J., Heiermann, M. (2012): Results of batch anaerobic digestion test - effect of enzyme addition. Agric. Eng. Int. 14 (1), 38-50.

Rahiman M. H. F., Taib M. N., Adnan R., Salleh Y. M. (2009): Analysis of weight decay regularisation in NNARX nonlinear identification; 2009 5<sup>th</sup> International Colloquium on Signal Processing & Its Applications, pp. 355–361, Mar. 2009.

Rahiman M. H. F. (2009): System Identification of Steam Distillation Essential Oil Extraction System. PhD Dissertation, Universiti Teknologi MARA Malaysia, 2009.

Rajendran K., Aslanzadeh S. and Taherzadeh M. J. (2012): Household Biogas Digesters—A Review. *Energies* 2012, *5*, 2911-2942; doi:10.3390/en5082911.

Rankovi'c V., Radulovi'c J., Grujovi'c N., Divac D. (2012): Neural Network Model Predictive Control of Nonlinear Systems Using Generic Algorithms. INT J COMPUT COMMUN, ISSN: 1841-9836; Vol.7 (2012), No. 3 (September), pp. 540-549.

Rapport J., Zhang R., Jenkins B. M., Williams R. B. (2008): Current anaerobic digestion technologies used for treatment of municipal organic waste. California Integrated Waste Management Board (March 2008). Available online at: http://www.calrecycle.ca.gov/Publications/Documents/Organics%5C2008011.pdf

Rasmussen K. (2010): A Relational Look at Renewable Energy and the Implication of Intermittent Power. Deseret Power. November2010, Edition 1.2. Available online at: http://docs.wind-watch.org/Rational-Look-Renewables.pdf

Residua, 2009, Information sheet of anaerobic digestion of solid waste. Available online at: http://www.waste.nl/content/download/472/3779/file/WB89\_InfoSheet(Anaerobic% 20 Digestion).pdf

Ritchie M. E., (2014): Plant compensation of grazing and soil carbon dynamics in tropical grassland. PeerJ. 2014; 2: e233. Available online at: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3912448/

Roberts R. J., Chen M. (2006): Waste incineration – how big is the health risk? A quantitative method to allow comparison with other health risks. *Journal of Public Health* 28(3), 261-266.

Robu B., Petruc V., Macoveanu M., (2005), Integrated Environmental Impact and Risk Assessment of Emissions Resulted from Oil Distribution, Environmental Engineering and Management Journal, 4, 499-513.

Romano R. T., Zhang R. (2011): Anaerobic digestion of onion residuals using a mesophilic Anaerobic Phased Solids Digester. *Biomass Bioenerg* 2011, 35(Issue 10):4174-4179 Salhofer, S., Obersteiner, G., Schneider, F., and Lebersorger, S., 2007. Potentials for the prevention of municipal solid waste. *Waste management*. Vol. 28: 245-259.

Salsabil M. R., Prorot A., Casellas M., Dagot C. (2009): Pre-treatment of activated sludge: Effect of sonication on aerobic and anaerobic digestibility, Chem. Eng. J. 148 (2–3) (2009) 327–335.

Sangodoyin, A.Y., Olorunfemi, A.I., 1996. Nutrient benefits and environmental aspects of land disposal of livestock waste. Environmental Management and Health 7 (1), 33–38.

Schmidt T. M., and Schaecheter M. (2012): Topic in Ecological and Environmental Microbilogy, Edited by T. M. Schmidt and M. Schaecheter Copyright @ 2012, Elsever Inc. All rights reserved.

Schober, G. Schafer, J., Schmid-Staiger, U. and Trosch, W., (1999): One and twostage digestion of solid organic waste. *Water research*. Vol. 33 (3): 854 – 860.

Schoukens J. Marconato A., Sjoberg J., Suykens J. (2012): Identification of the Silverbox Benchmark Using Nonlinear State-Space Models. 16th IFAC Symposium on System Identification, The International Federation of Automatic Control Brussels, Belgium. July 11-13, 2012.

Schunn, C. D., & Wallach, D. (2001). Evaluating goodness-of-fit in comparison of models to data. Online manuscript. Retrieved 1 April 2004 from www.lrdc.pitt.edu/schunn/gof/GOF.doc.

Scottish Natural Heritage (SNH) (2014): Siting and Designing Wind Farm in the Landscape. Version 2, May 2014. Available online at:

http://www.snh.org.uk/pdfs/strategy/renewables/Guidance\_Siting\_Designing\_wind\_farms.pd f

Shah B. A., Shah A. V., Singh R. R. (2009): Sorption isotherms and kinetics of chromium uptake from wastewater using natural sorbent material. Int. J. Environ. Sci. Tech., 6 (1), 77-90 (14 Pages).

Shah F. A., Mohamood Q., Shah M. M., Perivez A., Asad S. A. (2014): Microbial Ecology of Anaerobic Digesters: The Key Players of Anaerobiosis. The Scientific World Journal Volume 2014 (2014), Article ID 183752, 21 pages. doi.org/10.1155/2014/183752.

Shang Y., Johnson B. R., Sieger R. (2005): Application of the IWA anaerobic digestion model (ADM1) for simulating full-scale anaerobic sewage sludge digestion. Water Sci Technol 52: 487-492.

Shariff H. M. D., Rahiman M. H. F, Yassin I., Tajjudin M. (2014): System Identification of a Steam Distillation Pilot Scale using ARX and NARX Approach. International Journal of Research in Engineering and Technology; eISSN: 2319-1163; pISSN: 2321-7308. Volume: 03, Issue: 01; January 2014.

Sharvelle S., Keske C., Davis J., Lasker J. (2012): Guide for Assessing Feasibility of AD in Colorado. Colorado Engineering, Technical Note 28, NRCS CO-ENG-TN-28, July 2012 [online]. Available: http://efotg.sc.egov.usda.gov/references/public/CO/ENG-TN-28\_Guide\_for\_Feasibility\_of\_Anaerobic\_Digestion\_in\_Colorado.pdf.

Shekiro 111 J., Kuhu E. M., Nagle N. J., Tucker M. P., Elander R. T., Schell D. J. (2014): Characherzation of pilot-scale dilute acid pre-treatment performance using deacetylated corn stover. Biotechnol Biofuels. 2014; 7: 23. Available online at; http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3942107/

Simpson R., Sastry S. K. (2013): Chemical and Bioprocess Engineering: Fundamental Concept for First-Year Student. Springer Science & Business Media, 2013. ISBN: 1461491266, 9781461491262. 352 pages.

Sindall R. C. (2014): Increasing the efficiency of anaerobic waste digestion by optimising flow pattern to enhance biogas production, A PhD thesis, University of Birmingham Research Archive e-thesis repository. Available online at: http://etheses.bham.ac.uk/5636/1/Sindall15PhD.pdf

Sindall R., Bridgeman J., Carliell-Marquet C. (2013): Velocity gradient as a tool to characterise the link between mixing and biogas production in anaerobic waste digesters. Water Sci Technol. 2013;67(12):2800-6. doi: 10.2166/wst.2013.206.

Sjoberg J., and Zhang Q., Juditsky A., Hjalmarsson H., Benveniste A., Delyon B., Ljung L. (1995): "Nonlinear black-box modeling in System Identification: Mathematical foundations," *Automatica*, vol. 31, no. 12, pp. 1724–1750, 1995.

Sluiter A., Hames B., Hyman D., Payne C., Ruiz R., Scarlata C., Sluiter J., Templeton D., Wolfe J. (2008): Determination of Total Solids in Biomass and Total Dissolved Solids in Liquid Process Samples Laboratory Analytical Procedure (LAP). Available online at: http://www.nrel.gov/biomass/pdfs/42621.pdf

Sluiter B. J., Ruiz R. O., Scarlata C. J. Sluiter A. D., Templeton D. W. (2010): Compositional Analysis of Lignocellulosic Feedstock. 1. Review and Description of Methods. Journal of Agricultural and Food Chemistry. Aug 25, 2010; 58(16): 9043-9053. Available online at: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2923870/

Solantausta Y, Bridgwater A. T., Beckman D. (1995), Feasibility of power production with pyrolysis and gasification systems. Biomass and Bioenergy 1995; 9: 257–69.

Spanjers H., Bouvier J. C., Steenweg P., Bisschops 1., van Gils W., Versprille B (2006): Implementation of in-line infrared monitor in full-scale anaerobic digestion process. Water Sci Technol 2006, 53(4-5):55-61.

Speece R. E. and McCarty P. L. (1964): Advances in Water Poll. Res. 2, Pergamon Press, London, p. 305.

Spiess, A-N., Neumeyer N. (2010): An evaluation of R<sup>2</sup> as an inadequate measure for nonlinear models in pharmacological and biochemical research: a Monte Carlo approach. *BMC Pharmacology*. 2010; 10: 6. DOI:10.1186/1471-2210-10-6. Available online at: http://www.biomedcentral.com/1471-2210/10/6

Stamatelatou K., Syrou L., Kravaris C, Lyberatos G. (2009): An invariant manifold approach for CSTR model reduction in the presence of multi-step biochemical reaction schemes. Application to anaerobic digestion. Chem Eng J 150: 462- 475.

Starr C. and McMillan B. (2012): Human Biology (10<sup>th</sup> edition). © 2014, 2012 Brooks/Cole, Cengage Learning. ISBN-13:978-1-133-59916-6.

Steel D. (2007): Across the Boundaries: Extrapolation in Biology and Social Science. Oxford University Press, USA, 2007. ISBN: 0198044143, 9780198044147, 256 pages.

Stefansson, V., (1998), Estimate of the world geothermal potential. In: Geothermal Training in Iceland 20<sup>th</sup> Anniversary Workshop. United Nations University Geothermal Training Programme, Reykjavic, pp. 111–120.

Stein O. R., Towler B. W., Hook P. B., Biederman J. A. (2007): On fitting the k-C\* first order model to batch loaded sub-surface treatment wetlands. Water Sci Technol 56(3):93–99

Steyer J. P., Bouvier J. C., Conte T., Gras P., Harmand J., Delgenes J. P. (2002): On-line measurements of COD, TOC, VFA, total and partial alkalinity in anaerobic digestion processes using infra-red spectrometry. Water Science and Technology 45, 133–138.

Strik, D.P.B.T.B., Domnanovich, A.M., Zani, L., Braun, R., Holubar, P., 2005. Prediction of trace compounds in biogas from anaerobic digestion using the MATLAB neural network toolbox. Environmental Modelling & Software 20, 803–810.

Sugiki A., Furuta K., Ohata A., Nita H. (2014): Nonlinear variable structure adaptive control. American Control Conference (ACC), 2014. DOI:10.1109/ACC.2014.6858623, ISSN: 0743-1303; ISBN: 978-1-4799-3272-6; IEEE, Pages 1298 – 1303.

Switzenbaum M. S., Giraldo-Gomez E. and Hickey R. F. (1990): Monitoring of the anaerobic methane fermentation process. Enzyme and Microbial Technology, 12, 722-730.

Syngellakis S. (2014): Biomass to Biofuels. eISBN: 978-1-78466-036-9; Copyright @ WIT Press 2015.

Tabatabaei, M., Abdul Rahim, R., Abdullah, N., Wright, A.D.G., Shirai, Y., Sakai, K., Sulaiman, A., Hassan, M.A. Importance of the methanogenic archaea populations in anaerobic wastewater treatments, Process Biochemistry, In Press, Corrected Proof, Available online 10 June 2010, ISSN 1359-5113, DOI: 10.1016/j.procbio.2010, 05.017.

Tangirala A. K. (2015): Principles of System Identification: Theory and Practice. ©2015 by Taylor & Francis Group, LLC. NSBN: 13: 978-1-4398-9602-0 (eBook-PDF).

Taylor D. (1999): Talking trash: the economic and environmental issues of landfills. Environmental Health Perspectives. Volume 107, Number 8, August 1999.

te Boekhorst R. H., Ogilvie J. R., Pos J. (1981): An Overview of Current Simulation Models for Anaerobic Digesters. Livestock Waste: A renewable resource, ASAE American Society of Agricultural Engineers, No. 2, 105 – 108.

Tejada M., and Gonzalez J. L. (2006): Effects of two beet vinasse forms on soil physical properties and soil loss. Catena 68:41–50.

Themelis, N. J. and Verma S. (2004): "The Better Option: Anaerobic Digestion of Organic Waste in MSW." Waste Management World January/February 2004.

Themelis J. N., Ulloa P. A. (2007): Methane generation in landfills. Renewable Energy. Volume 32, Issue 7, June 2007, Pages 1243-1257.

Themelis N. J., Bourtsalas A. (2013): UK Waste management: Growing old or growing clean. Waste Management World, volume 14, issue 3. Available online at: http://www.waste-management-world.com/articles/print/volume-14/issue-3/features/uk-waste-management-growing-old-or-growing-clean.html

Tiehm A., Nickel K., Neis U., (1997): The use of ultrasound to accelerate the anaerobic digestion of sewage sludge, Water Sci. Technol. 36 (11) (1997) 121–128.

Tiehm A., Nickel K., Zellhorn M., Neis U. (2001): Ultrasonic waste activated sludge disintegration for improving anaerobic stabilization, Water Res. 35 (8) (2001) 2003–2009.

Tsoutsos T., Frantzeskaki N. and Gekas V. (2005): Environmental impacts from the solar energy technologies. Energy Policy 33 (2005) 289–296.

Tsoutsos T., Gekas, V., Marketaki K. (2003a): Technical and economical evaluation of Stirling dish solar thermal power generation. Renewable Energy 28 (6) 873–886.

Tsoutsos T. (2001): 'Environmental Impact Assessment for Energy Projects', educational notes, Environmental Department, Technical University of Crete, Chania.

Tudorache T,. Kreindler L (2010): "Design of a Solar Tracker System for PV Power Plants", Acta Polytechnica Hungarica, Vol. 7, No. 1, pp. 23-39, 2010.

Union of concerned scientists (UCS 2013): Clean energy. Available online at: http://www.ucsusa.org/clean\_energy/

United Nations, Department of Economic and Social Affairs, Population Division (2007): World Population Prospects: The 2006 Revision, Highlights, Working Paper No. ESA/P/WP.202. Available online at: http://www.un.org/esa/population/publications/ wpp2006/WPP2006\_Highlights\_rev.pdf.

United Nations Fund for Population Activities, UNFPA (2007): State of world population 2007: Unleashing the potential of urban growth. Available online at: http://www.unfpa.org/swp/2007/english/introduction.html.

United Tech, Inc. (2003). Anaerobic Digestion, UTI Web Design. 2003.

USEPA, (1991): Code of Federal Regulations; Title 40: Protection of Environment, part 258-Criteria for municipal solid waste landfills. Available online at: http://www.epa.gov/epacfr40/chapt-I.info/chi-toc.htm

USEPA, (2003): Municipal solid waste generation, recycling, and disposal in the United States, Facts and Figures; 2003. Available online at: www.epa.gov/epaoswer/non-hw/muncpl/pubs/msw05rpt.pdf.

USEPA, (2008): Anaerobic Digestion of Food Waste. Final Report, March 2008, Prepared by East Bay Municipal Utility Districtl Available online at: http://www.epa.gov/region9/organics/ad/EBMUDFinalReport.pdf

U.S. Food and Drug Administration, USFDA (2012): Bad Bug Book - Foodborne Pathogenic Microorganisms and Natural Toxins - Second Edition [online]. Available: http://www.fda.gov/downloads/Food/FoodborneIllnessContaminants/UCM297627.pdf

Usman M. A., Olanipekun O. O., Ogunbanwo O. A. (2012): Effect of Temperature on Biogas generation from Lignocellulosic Substrate. International Journal of Research in Chemistry and Environment *Vol. 2 Issue 2 April 2012(68-71)* ISSN 2248-9649. Available online at: http://www.ijrce.org/download.php?file=68-71.pdf

Valencia-Palomo G., Hilton K., Rossiter J. (2009): Pre-dictive control implementation in a PLC using the IEC 1131.3 programming standard. In Proceedings of the European Control Conference 2009. Available online at: https://www.sheffield.ac.uk/polopoly\_fs/1.60748!/file/report11.pdf

van Foreest F. (2012): Perspectives for Biogas in Europe. Oxford Institute for Energy Studies; NG70, December 2012; ISBN 978-1-907555-63-3. Available online at: http://www.oxfordenergy.org/wpcms/wp-content/uploads/2012/12/NG-70.pdf

Vandevivere P., De Baere L., Verstraete W. (2002): Types of anaerobic digesters for solid wastes, in Biomethanization of the Organic Fraction of Municipal Solid Wastes, J. Mata-Alvarez, Editor. 2002, IWA Publishing: Barcelona. p. 111-140.

Vaneeckhaute C., Meers E., Michels E., Ghekiere G., Accoe F., Tack F. M. G. (2013): Closing the nutrient cycle by using bio-digestion waste derivatives as synthetic fertilizer substitutes: a field experiment. *Biomass Bioenergy* 55, 175–189. doi:10.1016/j.biombioe.2013.01.032.

van Lier' J. B., Tilche A., Ahring B. K., acarie H., Moletta R., Dohanyos M., Hulshoff Pol L.W, Lens P., Verstraete W. (2001): New perspectives in anaerobic digestion. Water Science and Technology Vil 43, No.1, pp 1-18. IWA Publishing 2001.

van Liere J., Boosten G., van Dijk L., Hemke G., Verschoor A. (2011): Ecoferm! – de kringloopboerderij. Juni 2011, Innovatienetwerk.

Vázquez-Cruz M. A, Guzman-Cruz R., Espinnosa-Calderon A., Jiménez-Sánchez A. R., Guzmán-Cruz R. (2014): Biosystems Engineering: Biofactories for Food Production in the Century. Chapter 2; XXI, DOI: 10.1007/978-3-319-03880-3\_2, Springer International Publishing Switzerland 2014.

Veenstra S. (2000): *Wastewater Treatment I*. Delft: International Institute for Infrastructure, Hydraulics and Environmental Engineering (IHE Delft).

Vidal G.; Carvalho A.; Mendez R.; Lema J.M (2000): Influence of the content in fats and proteins on the anaerobic biodegradability of dairy wastewaters. Bioresource Technology. Volume 74, Number 3, September 2000, pp. 231-239(9)
Villanverde A. F. and Banga J. R. (2013): Reverse engineering and identification in system biology: Strategies, perspectives and challenges. Journal of The Royal Society Interface. DOI: 10.1098/rsif.2013.0505. Available online at: http://rsif.royalsocietypublishing.org/content/11/91/20130505.short

Ward A. J., Hobbs P. J., Hollimana P. J. and Jones D. L. (2008): Optimisation of the anaerobic digestion of agricultural resources, *Bioresource Technology*, Volume 99, Issue 17, November 2008, Pages 7928-7940.

Wang J. Y., Hobbs P., Chadwick D., Butler A. (2011): Modelling of N<sub>2</sub>O Emissions of Biogas Digestate for Use as Fertiliser Using DNDC. London: Agricultural Research Modellers' Group.

Wang J. (2014): Decentralized biogas technology of anaerobic digestion and farm ecosystem: opportunities and challenges. Front. Energy Res., 26 March 2014. Doi: 10.3389/fenrg.2014.00010 Available online at: http://journal.frontiersin.org/Journal/10.3389/fenrg.2014.00010/full#sthash.bCjiNstv.dpuf

Wang L. K., Shammas N. K., Hung Y-T (2010): Advance Biological Treatment Processes. E-ISBN: 978-1-60327-170-7. DOI: 10.1007/978-1-60327-170-7: Copyright @ 2009 Hamana Press

Wang Q, Kuninobo M., Kakimoto K., Ogawa H. I., Kato Y. (1999): Upgrading of anaerobic digestion of waste activated sludge by ultrasonic pre-treatment, Bioresour. Technol. 68 (1999) 309–313.

Weiland P. (2006): State of the art of solid-state digestion–recent developments. In: Rohstoffe, F.N. (Ed.), Solid-State Digestion–State of the Art and Further R&D Requirements, vol. 24. Gulzower Fachgespräche, pp. 22–38.

Willey J. M., Sherwood L., Woolverton C. J. and Prescott L. M. (2008): Prescott, Harley, and Klein's Microbiology. New York: McGraw-Hill Higher Education, 2008. Print.

Wilson Christopher. A., Murthy, Sudhir N., Novak John T. (2008): Digestibility Study of Wastewater Sludge Treated by Thermal Hydrolysis. *Residuals and Biosolids*, 2008, pp. 374-386.

Whitmore T. N., Lloyd D., Jones G., Williams T. N. (1987): Hydrogen-dependent control of the continuous anaerobic-digestion process. Appl. Microbiol. Biotechnol. 26, 383–388.

Wong B. T.; Show K. Y., Lee D. J., Lai J. Y. (2009): Carbon balance of anaerobic process: Carbon credit. Bioresource. Tech., 100, 1734–1739 (6 pages).

Word Energy Council (WEC) (1994): New Renewable Energy Resources, London, 1994.

Wu L., Hao Y., Sun C., Liu R. (2009): Effect of different solid concentration on biogas yield and composition during anaerobic fermentation process. International Journal Global Energy Issues 2009, 31(Nos. 3/4):240.

Wu, Wei. (2007): Anaerobic co-digestion of biomass for methane production: recent research achievements. Iowa State University. Available online at: home.eng.iastate.edu/~tge/ce421-521/wei.pdf.

Xie R., Xing Y., Ghani Y.A., Ooi K.E., Ng S.W. (2007): Full-scale demonstration of an ultrasonic disintegration technology in enhancing anaerobic digestion of mixed primary and thickened secondary sewage sludge, J. Environ. Eng. Sci. 6 (5) (2007) 533–541.

Yadvika, Santosh, Sreekrishnan T. R., Kohli S. and Rana V. (2004): Enhancement of biogas production from solid substrates using different techniques—A review. *Bioresour. Technol.* 2004, *95*, 1–10.

Yalkowsky S. H., He Y., Jain P. (2010): Hankbook of Aqueous Solubility Data. Second Edition; CRC Press; Boca Raton, FL 2010.

Yang S. T., Okos M. R. (1987): Kinetic study and a mathematical modeling of methanogenesis of acetate using pure cultures of methanogens. Biotechnol. Bioeng. 30, 661–667.

Yu L., Wensel P. C., Ma J., Chen S. (2013): A mathematical modelling in Anaerobic Digestion (AD). J Bioremed Biodeg S4: 003. doi:10.4172/2155-6199.S4-003

Zaher U., Cheong, D-Y., Wu. B., and Chen S., (2007): Producing energy and fertilizer from organic municipal solid waste. Olympia, WA: Department of Biological Systems Engineering, WSU. Also available online at: http://www.ecy.wa.gov/programs/ swfa/solidwastedata/

Zeng A. P., Hu W. S., Deckwer W. D. (1998): Variation of stoichiometric ratios and their correlation for monitoring and control of animal cell cultures. Biotechnol Prog. 1998;14:434–441.

Zhao C. (2011): Effect of Temperature on Biogas Production in Anaerobic Treatment of Domestic Wastewater UASB System in Harnmarby Sjostadsverk. TRITA LWR Degree Project 11:35.

Zhang B., He P., Lü F., Li-ming S., Wang P. (2007): Extracellular enzyme activities during regulated hydrolysis of high-solid organic wastes. *Water Research*. 41: 4468-4478.

Zhang Q. (1997): Using wavelet network in nonparametric estimation. IEE Trans. On Neural Network, Vol. 8, March 1997, pp. 227-236.

Zhang Q. and Cai W. (2008): Enzymatic hydrolysis of alkali-pretreated rice straw by Trichoderma reesei ZM4-F3. *Biomass Bioenergy*. 32: 1130–1135.

Zheng Y. H., Li Z. F., Feng S. F., Lucas M., Wu G. L., Li, Y. (2010): Biomass energy utilization in rural areas may contribute to alleviating energy crisis and global warming: a case study in a typical agro-village of Shandong, China. *Renew. Sustain. Energy Rev.* 14, 3132–3139. doi:10.1016/j.rser.2010.07.052.

Ziadat A. H., Mott H. (2005): Assessing solid waste recycling opportunities for closed campuses. Management of Environmental Quality: *An International Journal* 16 (3), 250–256.

Zoetemeyer R. J., Vandenheuvel J. C., Cohen A., (1982): pH influence on acidogenic dissimilation of glucose in an anaerobic digester. *Water Research* 16, 303–311.

Zorzuta S., Jovana V., Gradisara D., Musicb G. (2009): Closed-loop control of a polymerisation plant using production performance indicators (PIs). International Journal of Computer Integrated Manufacturing; Vol. 22, No. 12, December 2009, 1128–1143.

Zupančič G. D., Uranjek-Ževart N., Roš M. (2008): Full-scale anaerobic co-digestion of organic waste and municipal sludge. *Biomass and bioenergy*. Vol. 32: 163-167.

Zupančič G. D. and Grilc V. (2012): Anaerobic Treatment and Biogas Production from Organic Waste, Management of Organic Waste, Dr. Sunil Kumar (Ed.), ISBN: 978-953-307-925-7, InTech, Available online at:

http://www.intechopen.com/books/management-of-organic-waste/anaerobic-treatment-and-biogas-productionfrom-organic-wastes

# Appendices

## **Appendix A: Raw Experimental Data**

The following tables are the experimental data generated from the various experiments conducted in this research work:

A	verage exp	erimental	data for	temperature	variables for	· biogas	production:
				· · · · · · · · · · · · · · · · · · ·			<b>r</b>

Time (h)	32 (°C)	34 (°C)	36 (°C)	38 (°C)	40 (°C)	42 (°C)
	Avera	age biogas	productio	on (mL)		
1	0	0	0	0	0	0
2	0	0	10.2	56	75.6	3.999
3	0	40.6	75.6	101	200.4	40.6
4	0	80.8	164.2	162.8	270.6	93.8
5	0	100	220.6	232.4	350.8	123
6	0	137.2	270.6	274.2	490.8	205.2
7	20.4	194.8	351	352	900.4	247.8
8	49.6	211	610.4	587.2	1208.4	284
9	80	504.6	921	1210.8	2150.8	640.9
10	100	716.4	1245	1720	2299.8	903.4
11	170.8	813.8	1664.8	1901	2391.6	1223.4
12	249.6	1300.4	2130.8	2191.4	2509.2	1554.4
13	480	1520.4	2217.8	2380.8	2608	1954
14	713.8	1751.4	2251	2433.8	2658.4	2173.8
15	951	1851	2316	2566	2674.4	2214.4
16	1312	1870.8	2345.8	2581.2	2687.8	2264
17	1450	1911.4	2382.6	2601.6	2796.2	2286.8

18	1551	1961	2398.8	2659.2	2833.6	2304
19	1721.8	1983.8	2404	2703.6	2848.4	2316
20	1760.6	2001	2410.4	2750.8	2856.2	2354.4
21	1772.8	2053	2417.4	2756	2902	2273.8
22	1808.2	2036	2428	2760.4	2910.6	2234.4
23	1803	2010.6	2441.4	2766.4	2896.8	2214
24	1751.2	2001.6	2417.6	2755.8	2882.6	2202.6

Average experimental data for pH variables for biogas production: •

Time (h)	pH5	pH6	pH7	pH8
	A	verage biogas j	production (mI	_)
1	0	0	0	0
2	0	0	0	0
3	0	221.4	111.2	65.4
4	0	501.8	280.8	111.2
5	0	556.4	880	276
6	0	667.2	1500.6	609
7	0	776.4	2100	723.6
8	0	890	2165	889
9	0	1001	2334.4	1000.2
10	0	1276.4	2444.8	1160.6
11	0	1336	2447.8	1221.2
12	0	1443.2	2451.2	1444.8
13	0	1555.8	2455.4	1555.2
14	0	1610.8	2501.2	1612.4

15	0	1666	2515	1779.4
16	0	1775.8	2555.4	1890
17	0	1835.6	2561.4	2111.2
18	0	1887.6	2570.4	2164.6
19	0	2000.2	2571.8	2223.8
20	0	2111	2572	2223.6
21	0	2175.5	2570.6	2228.4
22	0	2176.4	2570	2230.8
23	0	2180.4	2568.4	2235.2
24	0	2180.4	2564.4	2241.2

рН	5	6	7	8
Time (h)	Avera	ige methane pro	oportion (%)	)
	pH5	pH6	pH7	pH8
1	0	0	0	0
2	0	0	0	0
3	0	20.4	22.4	13.2
4	0	21.3	25.1	15.3
5	0	21.9	27.5	18.4
6	0	22.5	35.4	20.5
7	0	23.4	42.6	22.4
8	0	23.7	46.3	24.4
9	0	26.1	50.4	25.6
10	0	28.1	53.1	27.5
11	0	28.7	55.9	28.1
12	0	32.1	58.1	29.7
13	0	32.4	60.2	30.2
14	0	33.8	63.6	31.2
15	0	35.5	64.8	33.1
16	0	36.1	66.1	33.9
17	0	37.7	66.8	34.3
18	0	39.8	67.2	34.7
19	0	41.1	67.3	35.2
20	0	42	67.5	36
21	0	42.8	67	37.1

• The experimental data for pH variables for methane proportion:

22	0	43.2	67	38.6
23	0	43.6	67.1	41.6
24	0	43.4	66.8	40.1

Time (h)	60 rpm	300 rpm	600 rpm	2000 rpm
	Average	biogas produc	ction (mL)	
1	0	0	0	0
2	0	0	0	0
3	9.835	9.9	13.886	0
4	27.337	28.426	22.713	0.725
5	53.25	62.417	30.864	1.607
6	85.01	100.495	40.003	44.695
7	125.039	131.037	60.295	105.943
8	170.98	187.919	78.635	280.469
9	246.189	270.157	108.927	403.921
10	477.932	551.351	216.99	624.04
11	662.972	699.288	374.17	775.983
12	1248.44	1299.48	566.154	923.088
13	1386.243	1448.427	712.228	1176.463
14	1506.908	1572.857	960.577	1342.453
15	1614.934	1671.636	1170.302	1358.886
16	1690.114	1756.958	1400.976	1345.772
17	1721.566	1775.247	1462.721	1322.994
18	1739.69	1808.921	1595.478	1286.17
19	1763.341	1865.618	1672.81	1178.759
20	1851.276	1906.031	1784.045	1103.005
21	1871.194	1921.918	1794.752	1093.271

• Average experimental data for mixing speed variables for biogas production:

22	1876.462	1949.288	1769.024	1086.152
23	1859.193	1937.858	1734.061	1078.495
24	1849.748	1918.488	1668.545	1063.987

Time (h)	60 rpm	300 rpm	600 rpm	2000 rpm		
Average methane proportion (%)						
1	0	0	0	0		
2	0	0	0	0		
3	11.7	13.2	25.3	0		
4	27.1	28.2	28.2	9.8		
5	35.5	38.7	30.7	15.2		
6	42.4	43.6	33.1	27.4		
7	46.1	48.1	35.3	40.6		
8	48.7	51.3	40.1	47.8		
9	50.1	53.6	43	53.3		
10	53.1	55.1	48.2	56.7		
11	55.7	56.2	51.2	59.1		
12	58	59.5	53.2	61.4		
13	60.2	61.5	55.3	61.9		
14	63	64.7	57.3	63.7		
15	64.3	65.4	60.3	64.5		
16	64.8	65.7	62.1	64.1		
17	64.7	66	63.6	63.4		
18	65	66.5	64.1	62.7		
19	65.6	66.9	65.1	60.9		
20	66.2	67.2	65.9	59.1		
21	66.6	67.5	65.2	58.5		

•	Average experimental	data for mixing	speed variables for	or methane proportion:
---	----------------------	-----------------	---------------------	------------------------

\_

22	66.7	68	65.5	58.2
23	66.2	67.5	64.5	57.9
24	66	67.1	64.2	57.6

Time (h)	0.7 bar	0.5 bar	0.3 bar	0.1 bar
Average biogas production (mL)				
1	0	0	0	0
2	0	0	0	0
3	21.86	31.37	83.2	102.73
4	62.707	93.878	115.6416	118.41
5	119.45	123.28	179.6012	217.84
6	157.56	303.44	377.3172	403.82
7	176.64	354.72	391.5	423.31
8	196.2	402.89	422.864	501.3
9	308.67	501.24	678.5653	931.77
10	683.65	966.06	1178.766	1329.4
11	809.09	1260	1486.471	1758
12	842.11	1281	1560.58	1864.9
13	1011.1	1490.4	1771.199	2093.7
14	1154.8	1688.9	1951.596	2213.7
15	1232.4	1849.4	2149.482	2408.4
16	1454.7	2095.3	2429.584	2637
17	1566.8	2247.4	2616.3	2850.1
18	1751.4	2597.5	2946.807	3118.8
19	1901.9	2707.5	3088.064	3266.2
20	1950.1	2778	3146.516	3337.1
21	1990.1	2831	3191.4	3443.8

Average experimental data for pressure variables for biogas production: ٠

-

22	2010.3	2969.1	3408.339	3621.4
23	2060.1	2931.1	3425.823	3630
24	2099.3	2893.3	3430.801	3608

Time	0.7 bar	0.5 bar	0.3 bar	0.1 bar
(h)				
	Average	methane prop	portion (%)	
1	0	0	0	0
2	0	0	0	0
3	32.2	28.2	20.5	15.6
4	35.5	30.5	23.1	17.6
5	42.6	34.3	25.5	20.9
6	46.3	37.7	29	23
7	49.5	43.2	33.5	24.8
8	52.3	46.6	36.4	36.3
9	57.6	49.7	40.6	32.6
10	59.1	52.9	44.2	35.5
11	61.2	55.4	47.3	39.5
12	63	58.4	51.2	43.6
13	63.7	58	53.9	46.4
14	65	59.8	56	49.3
15	66.7	61.5	58	52.6
16	68.4	63.1	60.2	54.2
17	71.5	64.4	61.6	56.5
18	73.2	66.9	62.3	59.9
19	75.4	68.3	62.9	60.4
20	75.7	69.4	63.5	61.2

• Average experimental data for pressure variables for methane proportion:

21	76.1	70.6	63.8	61.9
22	76.3	70.8	63.7	62.7
23	76.5	71.2	63.6	62.4
24	76.8	71.5	63.3	61.6

### **Appendix B: MATLAB Modelling Codes**

#### MATLAB modelling codes:

%Data processing

%Load data containing all the parameters from mat files into your own variables.

load('Biogas\_data.mat');

%extract from the K\_data.mat the time vector. Time=parameters(2:end,1);

Time=3600\*Time; %the time vector values in hours.

%From the data loaded extract production for temperatures and pH, mixing speed and pressure.

Temperature32=parameters(2:end,2);	%Extract the measurements for the Temperature at 32
nU5-noromotors(2) and 3).	% Extract the measurements for pU of 5
pris-parameters(2:end,5),	<sup>6</sup> Extract the measurements for ma 60
$\frac{11800}{2} = \text{parameters}(2.\text{end}, 4); \qquad \%$	$^{\circ}$ Extract the measurements for fits 60.
pressure/00=parameters(2:end,5);	% Extract the measurements for pressure 700 mbar
Temperature34=parameters(2:end,6); degrees.	%Extract the measurements for the Temperature at 34
pH6=parameters(2:end,7);	%Extract the measurements for pH of 6
ms270=parameters(2:end,8); pressure600=parameters(2:end,9);	%Extract the measurements for ms 300
Temperature36=parameters(2:end,10) pH65=parameters(2:end,11); ms300=parameters(2:end,12); pressure500=parameters(2:end,13);	);
Temperature38=parameters(2:end,14) pH7=parameters(2:end,15); ms500=parameters(2:end,16); pressure300=parameters(2:end,17);	);
Temperature40=parameters(2:end,18) pH75=parameters(2:end,19); ms600=parameters(2:end,20); pressure200=parameters(2:end,21);	);
Temperature42=parameters(2:end,22)	);

```
pH8=parameters(2:end,23);
ms2000=parameters(2:end,24);
pressure100=parameters(2:end,25);
Temperature35=parameters(2:end,26);
Temperature37=parameters(2:end,27);
% Creation of an iddata object for each of the experiments.
%Experiment 1
InputExp1=ones(25,4);
InputExp1(1:end,1)=32.*InputExp1(1:end,1);
                                             %Temperature (C)
InputExp1(1:end,2)=7.1.*InputExp1(1:end,2);
                                             %pH
InputExp1(1:end,3)=250.*InputExp1(1:end,3);
                                              %Mixing speed (rpm)
InputExp1(1:end,4)=0.7.*InputExp1(1:end,4);
                                             %Pressure (bar)
OutputExp1=Temperature32;
SystemEperiment1=iddata(OutputExp1,InputExp1,3600);
SystemEperiment1.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment1.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment1.OutputName=";
SystemEperiment1.OutputUnit='mL';
%Experiment 2
InputExp2=ones(25,4);
InputExp2(1:end,1)=34.*InputExp2(1:end,1);
                                             %Temperature (C)
InputExp2(1:end,2)=7.1.*InputExp2(1:end,2);
                                             %pH
InputExp2(1:end,3)=250.*InputExp2(1:end,3);
                                              %Mixing speed (rpm)
InputExp2(1:end,4)=0.7.*InputExp2(1:end,4);
                                             %Pressure (bar)
OutputExp2=Temperature34;
SystemEperiment2=iddata(OutputExp2,InputExp2,3600);
SystemEperiment2.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment2.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment2.OutputName=";
SystemEperiment2.OutputUnit='mL';
%Experiment 3
InputExp3=ones(25,4);
InputExp3(1:end,1)=36.*InputExp3(1:end,1);
                                             %Temperature (C)
InputExp3(1:end,2)=7.1.*InputExp3(1:end,2);
                                             %pH
InputExp3(1:end,3)=250.*InputExp3(1:end,3);
                                              %Mixing speed (rpm)
InputExp3(1:end,4)=0.7.*InputExp3(1:end,4);
                                             %Pressure (bar)
OutputExp3=Temperature36;
SystemEperiment3=iddata(OutputExp3,InputExp3,3600);
SystemEperiment3.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment3.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment3.OutputName=";
SystemEperiment3.OutputUnit='mL';
%Experiment 4
InputExp4=ones(25,4);
InputExp4(1:end,1)=38.*InputExp4(1:end,1);
                                             %Temperature (C)
InputExp4(1:end,2)=7.1.*InputExp4(1:end,2);
                                             %pH
InputExp4(1:end,3)=250.*InputExp4(1:end,3);
                                              %Mixing speed (rpm)
```

InputExp4(1:end,4)=0.7.\*InputExp4(1:end,4); %Pressure (bar) OutputExp4=Temperature38; SystemEperiment4=iddata(OutputExp4,InputExp4,3600); SystemEperiment4.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment4.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment4.OutputName="; SystemEperiment4.OutputUnit='mL'; %Experiment 5 InputExp5=ones(25,4); InputExp5(1:end,1)=40.\*InputExp5(1:end,1); %Temperature (C) InputExp5(1:end,2)=7.1.\*InputExp5(1:end,2); %pH InputExp5(1:end,3)=250.\*InputExp5(1:end,3); %Mixing speed (rpm) InputExp5(1:end,4)=0.7.\*InputExp5(1:end,4); %Pressure (bar) OutputExp5=Temperature40; SystemEperiment5=iddata(OutputExp5,InputExp5,3600); SystemEperiment5.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment5.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment5.OutputName="; SystemEperiment5.OutputUnit='mL'; %Experiment 6 InputExp6=ones(25,4); InputExp6(1:end,1)=42.\*InputExp6(1:end,1); %Temperature (C) InputExp6(1:end,2)=7.1.\*InputExp6(1:end,2); %pH InputExp6(1:end,3)=250.\*InputExp6(1:end,3); % Mixing speed (rpm) InputExp6(1:end,4)=0.7.\*InputExp6(1:end,4); %Pressure (bar) OutputExp6=Temperature42; SystemEperiment6=iddata(OutputExp6,InputExp6,3600); SystemEperiment6.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment6.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment6.OutputName="; SystemEperiment6.OutputUnit='mL'; %Experiment 7 InputExp7=ones(25,4); InputExp7(1:end,1)=39.\*InputExp7(1:end,1); %Temperature (C) InputExp7(1:end,2)=5.\*InputExp7(1:end,2); %pH InputExp7(1:end,3)=250.\*InputExp7(1:end,3); %Mixing speed (rpm) InputExp7(1:end,4)=0.7.\*InputExp7(1:end,4); %Pressure (bar) OutputExp7=pH5; SystemEperiment7=iddata(OutputExp7,InputExp7,3600); SystemEperiment7.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment7.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment7.OutputName="; SystemEperiment7.OutputUnit='mL'; %Experiment 8 InputExp8=ones(25,4);

InputExp8(1:end,1)=39.\*InputExp8(1:end,1); %Temperature (C)

InputExp8(1:end,2)=6.\*InputExp8(1:end,2); %pH InputExp8(1:end,3)=250.\*InputExp8(1:end,3); %Mixing speed (rpm) InputExp8(1:end,4)=0.7.\*InputExp8(1:end,4); %Pressure (bar) OutputExp8=pH6; SystemEperiment8=iddata(OutputExp8,InputExp8,3600); SystemEperiment8.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment8.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment8.OutputName="; SystemEperiment8.OutputUnit='mL'; %Experiment 9 InputExp9=ones(25,4); InputExp9(1:end,1)=39.\*InputExp9(1:end,1); %Temperature (C) InputExp9(1:end,2)=7.\*InputExp9(1:end,2); %pH InputExp9(1:end,3)=250.\*InputExp9(1:end,3); % Mixing speed (rpm) InputExp9(1:end,4)=0.7.\*InputExp9(1:end,4); %Pressure (bar)

OutputExp9=pH7;

SystemEperiment9=iddata(OutputExp9,InputExp9,3600);

SystemEperiment9.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment9.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment9.OutputName=";

SystemEperiment9.OutputUnit='mL';

```
%Experiment 10
InputExp10=ones(25,4);
InputExp10(1:end,1)=39.*InputExp10(1:end,1);
                                               %Temperature (C)
InputExp10(1:end,2)=8.*InputExp10(1:end,2);
                                              %pH
InputExp10(1:end,3)=250.*InputExp10(1:end,3);
                                                %Mixing speed (rpm)
InputExp10(1:end,4)=0.7.*InputExp10(1:end,4);
                                               %Pressure (bar)
OutputExp10=pH8;
SystemEperiment10=iddata(OutputExp10,InputExp10,3600);
SystemEperiment10.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment10.inputUnit={'C', '', 'rpm', 'bar'};
SystemEperiment10.OutputName=";
SystemEperiment10.OutputUnit='mL';
```

```
%Experiment 11
InputExp11=ones(25,4);
InputExp11(1:end,1)=39.*InputExp11(1:end,1);
                                                %Temperature (C)
InputExp11(1:end,2)=7.1.*InputExp11(1:end,2);
                                                %pH
                                               % Mixing speed (rpm)
InputExp11(1:end,3)=60.*InputExp11(1:end,3);
InputExp11(1:end,4)=0.7.*InputExp11(1:end,4);
                                                %Pressure (bar)
OutputExp11=ms60;
SystemEperiment11=iddata(OutputExp11,InputExp11,3600);
SystemEperiment11.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment11.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment11.OutputName=";
SystemEperiment11.OutputUnit='mL';
```

%Experiment 12

InputExp12=ones(25,4); InputExp12(1:end,1)=39.\*InputExp12(1:end,1); %Temperature (C) InputExp12(1:end,2)=7.1.\*InputExp12(1:end,2); %pH InputExp12(1:end,3)=300.\*InputExp12(1:end,3); % Mixing speed (rpm) %Pressure (bar) InputExp12(1:end,4)=0.7.\*InputExp12(1:end,4); OutputExp12=ms300; SystemEperiment12=iddata(OutputExp12,InputExp12,3600); SystemEperiment12.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment12.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment12.OutputName="; SystemEperiment12.OutputUnit='mL'; %Experiment 13 InputExp13=ones(25,4); InputExp13(1:end,1)=39.\*InputExp13(1:end,1); %Temperature (C) InputExp13(1:end,2)=7.1.\*InputExp13(1:end,2); %pH InputExp13(1:end,3)=600.\*InputExp13(1:end,3); %Mixing speed (rpm) InputExp13(1:end,4)=0.7.\*InputExp13(1:end,4); %Pressure (bar) OutputExp13=ms600; SystemEperiment13=iddata(OutputExp13,InputExp13,3600); SystemEperiment13.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment13.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment13.OutputName="; SystemEperiment13.OutputUnit='mL'; %Experiment 14 InputExp14=ones(25,4); InputExp14(1:end,1)=39.\*InputExp14(1:end,1); %Temperature (C) InputExp14(1:end,2)=7.1.\*InputExp14(1:end,2); %pH InputExp14(1:end,3)=2000.\*InputExp14(1:end,3); % Mixing speed (rpm) InputExp14(1:end,4)=0.7.\*InputExp14(1:end,4); %Pressure (bar) OutputExp14=ms2000; SystemEperiment14=iddata(OutputExp14,InputExp14,3600); SystemEperiment14.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment14.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment14.OutputName="; SystemEperiment14.OutputUnit='mL'; %Experiment 15 InputExp15=ones(25,4); InputExp15(1:end,1)=39.\*InputExp15(1:end,1); % Temperature (C) InputExp15(1:end,2)=7.1.\*InputExp15(1:end,2); %pH InputExp15(1:end,3)=250.\*InputExp15(1:end,3); %Mixing speed (rpm) InputExp15(1:end,4)=0.1.\*InputExp15(1:end,4); %Pressure (bar) OutputExp15=pressure100; SystemEperiment15=iddata(OutputExp15,InputExp15,3600); SystemEperiment15.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment15.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment15.OutputName="; SystemEperiment15.OutputUnit='mL';

%Experiment 16 InputExp16=ones(25,4); InputExp16(1:end,1)=39.\*InputExp16(1:end,1); %Temperature (C) InputExp16(1:end,2)=7.1.\*InputExp16(1:end,2); %pH InputExp16(1:end,3)=250.\*InputExp16(1:end,3); %Mixing speed (rpm) InputExp16(1:end,4)=0.3.\*InputExp16(1:end,4); %Pressure (bar) OutputExp16=pressure300; SystemEperiment16=iddata(OutputExp16,InputExp16,3600); SystemEperiment16.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment16.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment16.OutputName="; SystemEperiment16.OutputUnit='mL'; %Experiment 17 InputExp17=ones(25,4); InputExp17(1:end,1)=39.\*InputExp17(1:end,1); %Temperature (C) InputExp17(1:end,2)=7.1.\*InputExp17(1:end,2); %pH InputExp17(1:end,3)=250.\*InputExp17(1:end,3); %Mixing speed (rpm) InputExp17(1:end,4)=0.5.\*InputExp17(1:end,4); %Pressure (bar) OutputExp17=pressure500; SystemEperiment17=iddata(OutputExp17,InputExp17,3600); SystemEperiment17.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment17.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment17.OutputName="; SystemEperiment17.OutputUnit='mL'; %Experiment 18 InputExp18=ones(25,4); InputExp18(1:end,1)=39.\*InputExp18(1:end,1); %Temperature (C) InputExp18(1:end,2)=7.1.\*InputExp18(1:end,2); %pH InputExp18(1:end,3)=250.\*InputExp18(1:end,3); %Mixing speed (rpm) InputExp18(1:end,4)=0.7.\*InputExp18(1:end,4); %Pressure (bar) OutputExp18=pressure700; SystemEperiment18=iddata(OutputExp18,InputExp18,3600); SystemEperiment18.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment18.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment18.OutputName="; SystemEperiment18.OutputUnit='mL'; %Experiment 19 InputExp19=ones(25,4); InputExp19(1:end,1)=39.\*InputExp19(1:end,1); %Temperature (C) InputExp19(1:end,2)=7.1.\*InputExp19(1:end,2); %pH InputExp19(1:end,3)=270.\*InputExp19(1:end,3); %Mixing speed (rpm) InputExp19(1:end,4)=0.7.\*InputExp19(1:end,4); %Pressure (bar) OutputExp19=ms270; SystemEperiment19=iddata(OutputExp19,InputExp19,3600); SystemEperiment19.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment19.inputUnit={'C', ' ', 'rpm', 'bar'};

SystemEperiment19.OutputName="; SystemEperiment19.OutputUnit='mL'; %Experiment 20 InputExp20=ones(25,4); InputExp20(1:end,1)=39.\*InputExp20(1:end,1); %Temperature (C) InputExp20(1:end,2)=7.1.\*InputExp20(1:end,2); %pH InputExp20(1:end,3)=250.\*InputExp20(1:end,3); %Mixing speed (rpm) %Pressure (bar) InputExp20(1:end,4)=0.6.\*InputExp20(1:end,4); OutputExp20=pressure600; SystemEperiment20=iddata(OutputExp20,InputExp20,3600); SystemEperiment20.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment20.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment20.OutputName="; SystemEperiment20.OutputUnit='mL'; %Experiment 21 InputExp21=ones(25,4); InputExp21(1:end,1)=39.\*InputExp21(1:end,1); %Temperature (C) InputExp21(1:end,2)=(6.5).\*InputExp21(1:end,2); %pH InputExp21(1:end,3)=250.\*InputExp21(1:end,3); %Mixing speed (rpm) InputExp21(1:end,4)=0.7.\*InputExp21(1:end,4); %Pressure (bar) OutputExp21=pH65; SystemEperiment21=iddata(OutputExp21,InputExp21,3600); SystemEperiment21.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment21.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment21.OutputName="; SystemEperiment21.OutputUnit='mL'; %Experiment 22 InputExp22=ones(25,4); InputExp22(1:end,1)=39.\*InputExp22(1:end,1); %Temperature (C) InputExp22(1:end,2)=(7.5).\*InputExp22(1:end,2); %pH InputExp22(1:end,3)=250.\*InputExp22(1:end,3); %Mixing speed (rpm) InputExp22(1:end,4)=0.7.\*InputExp22(1:end,4); %Pressure (bar) OutputExp22=pH75; SystemEperiment22=iddata(OutputExp22,InputExp22,3600); SystemEperiment22.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment22.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment22.OutputName="; SystemEperiment22.OutputUnit='mL'; %Experiment 23 InputExp23=ones(25,4); InputExp23(1:end,1)=39.\*InputExp23(1:end,1); % Temperature (C) InputExp23(1:end,2)=7.1.\*InputExp23(1:end,2); %pH InputExp23(1:end,3)=500.\*InputExp23(1:end,3); %Mixing speed (rpm) InputExp23(1:end,4)=0.7.\*InputExp23(1:end,4); %Pressure (bar) OutputExp23=ms500; SystemEperiment23=iddata(OutputExp23,InputExp23,3600);

SystemEperiment23.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment23.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment23.OutputName="; SystemEperiment23.OutputUnit='mL'; %Experiment 24 InputExp24=ones(25,4); InputExp24(1:end,1)=39.\*InputExp24(1:end,1); %Temperature (C) InputExp24(1:end,2)=7.1.\*InputExp24(1:end,2); %pH InputExp24(1:end,3)=250.\*InputExp24(1:end,3); %Mixing speed (rpm) InputExp24(1:end,4)=0.2.\*InputExp24(1:end,4); %Pressure (bar) OutputExp24=pressure200; SystemEperiment24=iddata(OutputExp24,InputExp24,3600); SystemEperiment24.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment24.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment24.OutputName="; SystemEperiment24.OutputUnit='mL'; % Experiment 25 InputExp25=ones(25,4); InputExp25(1:end,1)=35.\*InputExp25(1:end,1); %Temperature (C) InputExp25(1:end,2)=7.1.\*InputExp25(1:end,2); %pH InputExp25(1:end,3)=250.\*InputExp25(1:end,3); %Mixing speed (rpm) InputExp25(1:end,4)=0.7.\*InputExp25(1:end,4); %Pressure (bar) OutputExp25=Temperature35; SystemEperiment25=iddata(OutputExp25,InputExp25,3600); SystemEperiment25.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment25.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment25.OutputName="; SystemEperiment25.OutputUnit='mL'; % Experiment 26 InputExp26=ones(25,4); InputExp26(1:end,1)=37.\*InputExp26(1:end,1); % Temperature (C) InputExp26(1:end,2)=7.1.\*InputExp26(1:end,2); %pH InputExp26(1:end,3)=250.\*InputExp26(1:end,3); %Mixing speed (rpm) InputExp26(1:end,4)=0.7.\*InputExp26(1:end,4); %Pressure (bar) OutputExp26=Temperature37; SystemEperiment26=iddata(OutputExp26,InputExp26,3600); SystemEperiment26.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment26.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment26.OutputName="; SystemEperiment26.OutputUnit='mL'; %Using KI, K2, K3 as estimation data sets and K6 as validation data set % merging all estimation iddata to a single iddata object; % This is for all the measurements that will be considered for the model extraction. Overall\_iddata=merge(SystemEperiment1,SystemEperiment2, SystemEperiment3, SystemEperiment8, SystemEperiment11, SystemEperiment12, SystemEperiment17, SystemEperiment18, SystemEperiment19, SystemEperiment20, SystemEperiment21, SystemEperiment25, SystemEperiment16, SystemEperiment22, SystemEperiment23, SystemEperiment26);

#### % VALIDATION PROCESS

% merging data to a single iddata object;

% This is for all the measurements that will be considered for the model extraction.

Overall\_iddata\_val=merge(SystemEperiment4,SystemEperiment5, SystemEperiment6, SystemEperiment10, SystemEperiment13, SystemEperiment14, SystemEperiment15);

%Estimation data and validation data have been entered and two iddata objects have been created:

%Overall\_iddata contains the experiment results for the model identification process.

%Overall\_iddata\_val contains the experiment results for the model validation.

de=(Overall\_iddata);

dv=(Overall\_iddata\_val);

%NALRX MODEL

%mx1 = nlarx(de, [ones(1,1), ones(1,4), ones(1,4)], wavenet) %mx2 = nlarx(de, [ones(1,1), ones(1,4), 2\*ones(1,4)], wavenet) %mx3 = nlarx(de, [2\*ones(1,1), ones(1,4), 2\*ones(1,4)], wavenet) %mx4 = nlarx(de, [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)], wavenet) %mx5 = nlarx(de, [3\*ones(1,1), ones(1,4), ones(1,4)], wavenet) %mx6 = nlarx(de, [3\*ones(1,1), 3\*ones(1,4), ones(1,4)], wavenet)

%compare(dv,mx1, mx2, mx3, mx4, mx5, mx6)

%The numbers of units (wavelets) of the two WAVENET estimators have been automatically chosen by the estimation algorithm. These numbers are displayed below. Notice the abbreviations 'nl'='Nonlinearity' and 'num'='NumberOfUnits' %mx4.Nonlinearity(1).NumberOfUnits %using full property names

%nanbnk = [2\*ones(1,1), ones(1,4), 2\*ones(1,4)];

%The number of units in the WAVENET estimators can be explicitly specified instead of being automatically chosen by the estimation algorithm: %mx7 = nlarx(de, nanbnk, [wavenet('num',6)]); %mx8 = nlarx(de, nanbnk, [wavenet('num',8)]); %mx9 = nlarx(de, [2\*ones(1,1), ones(1,4), 2\*ones(1,4)], [wavenet('num',9)])

%compare(dv,mx7, mx8, mx9)

%Nonlinear ARX Model - Trying Other Nonlinearity Estimators

%mx10 = nlarx(de, nanbnk, treepartition);

%The SIGMOIDNET estimator can also be used. Estimation options such as maximum iterations (MaxIter) and iteration display can be specified using NLARXOPTIONS command. %opt = nlarxOptions('Display','on'); %opt.SearchOption.MaxIter = 2; %mx11 = nlarx(de, nanbnk, sigmoidnet);

```
%compare(dv, mx12, mx13)
```

%Function PLOT may be used to view the nonlinearity responses of various models. %plot(mx7, mx12, mx13)

%Estimation of Hammerstein-Wiener Model (Both Input and Output Nonlinearities)

%Indicate both input and output nonlinearities for a Hammerstein-Wiener model. As in the case of Nonlinear ARX models, we can use a string (rather than object) to specify the nonlinearity estimator.

%mhw1 = nlarx(de, [1 5 3], wavenet); %ws = warning('off','Ident:estimation:NparGTNsamp'); %mhw1 = nlhw(de, [4, 8\*ones(1,4), 8\*ones(1,4)],pwlinear, pwlinear);

%mhw1 = nlhw(de, [ones(1,4), ones(1,4), zeros(1,4)], 'pwlinear', 'pwlinear'); mhw2 = nlhw(de, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'pwlinear', 'pwlinear'); %mhw3 = nlhw(de, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'pwlinear', 'pwlinear');

```
%compare(dv, mhw2)
%compare(dv,mhw1, mhw2, mhw3)
```

%mhw4 = nlhw(de, [ones(1,4), ones(1,4), zeros(1,4)], 'unitgain', 'deadzone'); %mhw5 = nlhw(de, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'unitgain', 'deadzone') %mhw6 = nlhw(de, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'unitgain', 'deadzone');

%compare(dv,mhw4, mhw5, mhw6)

%mhw7 = nlhw(de, [ones(1,4), ones(1,4), zeros(1,4)], 'saturation', 'deadzone'); %mhw8 = nlhw(de, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'saturation', 'deadzone') %mhw9 = nlhw(de, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'saturation', 'deadzone');

%compare(dv,mhw2, mhw5, mhw8)

%compare(dv, mx9, mhw2)

%AIC = aic (mx9, mhw2)

%The limit values of the SATURATION estimators can be accessed. The short-hands

'u'='input', 'y'='output', and 'nl'='Nonlinearity' are available

%Data processing

%Load data containing all the parameters from mat files into your own variables.

load('Methane\_data.mat');

%extract from the K\_data.mat the time vector. Time=parameters(2:end,1);

Time=3600\*Time; %the time vector values in hours.

%From the data loaded extract production for temperatures and pH, mixing speed and pressure.

Temperature32=parameters(2:end,2); %Extract the measurements for the Temperature at 32 degrees. pH5=parameters(2:end,3); %Extract the measurements for pH of 5. ms60=parameters(2:end,4); %Extract the measurements for ms 60. pressure700=parameters(2:end,5); %Extract the measurements for pressure 700 mbar Temperature34=parameters(2:end,6); %Extract the measurements for the Temperature at 34 degrees. pH6=parameters(2:end,7); %Extract the measurements for pH of 6 ms270=parameters(2:end,8); %Extract the measurements for ms 300 pressure600=parameters(2:end,9); Temperature36=parameters(2:end,10); pH65=parameters(2:end,11); ms300=parameters(2:end,12); pressure500=parameters(2:end,13);

Temperature38=parameters(2:end,14); pH7=parameters(2:end,15); ms500=parameters(2:end,16); pressure300=parameters(2:end,17);

Temperature40=parameters(2:end,18);

```
pH75=parameters(2:end,19);
ms600=parameters(2:end,20);
pressure200=parameters(2:end,21);
Temperature42=parameters(2:end,22);
pH8=parameters(2:end,23);
ms2000=parameters(2:end,24);
pressure100=parameters(2:end,25);
Temperature35=parameters(2:end,26);
Temperature37=parameters(2:end,27);
% Creation of an iddata object for each of the experiments.
%Experiment 1
InputExp1=ones(25,4);
InputExp1(1:end,1)=32.*InputExp1(1:end,1);
                                             %Temperature (C)
InputExp1(1:end,2)=7.1.*InputExp1(1:end,2);
                                             %pH
InputExp1(1:end,3)=250.*InputExp1(1:end,3);
                                              %Mixing speed (rpm)
InputExp1(1:end,4)=0.7.*InputExp1(1:end,4);
                                             %Pressure (bar)
OutputExp1=Temperature32;
SystemEperiment1=iddata(OutputExp1,InputExp1,3600);
SystemEperiment1.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment1.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment1.OutputName=";
SystemEperiment1.OutputUnit='mL';
%Experiment 2
InputExp2=ones(25,4);
InputExp2(1:end,1)=34.*InputExp2(1:end,1);
                                             %Temperature (C)
InputExp2(1:end,2)=7.1.*InputExp2(1:end,2);
                                             %pH
InputExp2(1:end,3)=250.*InputExp2(1:end,3);
                                              % Mixing speed (rpm)
InputExp2(1:end,4)=0.7.*InputExp2(1:end,4);
                                             %Pressure (bar)
OutputExp2=Temperature34;
SystemEperiment2=iddata(OutputExp2,InputExp2,3600);
SystemEperiment2.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment2.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment2.OutputName=";
SystemEperiment2.OutputUnit='mL';
%Experiment 3
InputExp3=ones(25,4);
InputExp3(1:end,1)=36.*InputExp3(1:end,1);
                                             %Temperature (C)
InputExp3(1:end,2)=7.1.*InputExp3(1:end,2);
                                             %pH
InputExp3(1:end,3)=250.*InputExp3(1:end,3);
                                              %Mixing speed (rpm)
                                             %Pressure (bar)
InputExp3(1:end,4)=0.7.*InputExp3(1:end,4);
OutputExp3=Temperature36;
SystemEperiment3=iddata(OutputExp3,InputExp3,3600);
SystemEperiment3.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment3.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment3.OutputName=";
SystemEperiment3.OutputUnit='mL';
```

```
%Experiment 4
InputExp4=ones(25,4);
InputExp4(1:end,1)=38.*InputExp4(1:end,1);
                                             %Temperature (C)
InputExp4(1:end,2)=7.1.*InputExp4(1:end,2);
                                             %pH
InputExp4(1:end,3)=250.*InputExp4(1:end,3);
                                              % Mixing speed (rpm)
InputExp4(1:end,4)=0.7.*InputExp4(1:end,4);
                                             %Pressure (bar)
OutputExp4=Temperature38;
SystemEperiment4=iddata(OutputExp4,InputExp4,3600);
SystemEperiment4.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment4.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment4.OutputName=";
SystemEperiment4.OutputUnit='mL';
%Experiment 5
InputExp5=ones(25,4);
InputExp5(1:end,1)=40.*InputExp5(1:end,1);
                                             %Temperature (C)
InputExp5(1:end,2)=7.1.*InputExp5(1:end,2);
                                             %pH
InputExp5(1:end,3)=250.*InputExp5(1:end,3);
                                              % Mixing speed (rpm)
InputExp5(1:end,4)=0.7.*InputExp5(1:end,4);
                                             %Pressure (bar)
OutputExp5=Temperature40;
SystemEperiment5=iddata(OutputExp5,InputExp5,3600);
SystemEperiment5.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment5.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment5.OutputName=";
SystemEperiment5.OutputUnit='mL';
%Experiment 6
InputExp6=ones(25,4);
InputExp6(1:end,1)=42.*InputExp6(1:end,1);
                                             %Temperature (C)
InputExp6(1:end,2)=7.1.*InputExp6(1:end,2);
                                             %pH
InputExp6(1:end,3)=250.*InputExp6(1:end,3);
                                              % Mixing speed (rpm)
InputExp6(1:end,4)=0.7.*InputExp6(1:end,4);
                                             %Pressure (bar)
OutputExp6=Temperature42;
SystemEperiment6=iddata(OutputExp6,InputExp6,3600);
SystemEperiment6.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment6.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment6.OutputName=";
SystemEperiment6.OutputUnit='mL';
%Experiment 7
InputExp7=ones(25,4);
InputExp7(1:end,1)=39.*InputExp7(1:end,1);
                                             %Temperature (C)
InputExp7(1:end,2)=5.*InputExp7(1:end,2);
                                            %pH
InputExp7(1:end,3)=250.*InputExp7(1:end,3);
                                              % Mixing speed (rpm)
InputExp7(1:end,4)=0.7.*InputExp7(1:end,4);
                                             %Pressure (bar)
OutputExp7=pH5;
SystemEperiment7=iddata(OutputExp7,InputExp7,3600);
SystemEperiment7.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment7.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment7.OutputName=";
```

SystemEperiment7.OutputUnit='mL';

%Experiment 8 InputExp8=ones(25,4); InputExp8(1:end,1)=39.\*InputExp8(1:end,1); %Temperature (C) InputExp8(1:end,2)=6.\*InputExp8(1:end,2); %pH InputExp8(1:end,3)=250.\*InputExp8(1:end,3); % Mixing speed (rpm) InputExp8(1:end,4)=0.7.\*InputExp8(1:end,4); %Pressure (bar) OutputExp8=pH6; SystemEperiment8=iddata(OutputExp8,InputExp8,3600); SystemEperiment8.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment8.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment8.OutputName="; SystemEperiment8.OutputUnit='mL'; %Experiment 9 InputExp9=ones(25,4); InputExp9(1:end,1)=39.\*InputExp9(1:end,1); %Temperature (C) InputExp9(1:end,2)=7.\*InputExp9(1:end,2); %pH InputExp9(1:end,3)=250.\*InputExp9(1:end,3); %Mixing speed (rpm) InputExp9(1:end,4)=0.7.\*InputExp9(1:end,4); %Pressure (bar) OutputExp9=pH7; SystemEperiment9=iddata(OutputExp9,InputExp9,3600); SystemEperiment9.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment9.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment9.OutputName="; SystemEperiment9.OutputUnit='mL'; %Experiment 10 InputExp10=ones(25,4); InputExp10(1:end,1)=39.\*InputExp10(1:end,1); %Temperature (C) InputExp10(1:end,2)=8.\*InputExp10(1:end,2); %pH InputExp10(1:end,3)=250.\*InputExp10(1:end,3); %Mixing speed (rpm) InputExp10(1:end,4)=0.7.\*InputExp10(1:end,4); %Pressure (bar) OutputExp10=pH8; SystemEperiment10=iddata(OutputExp10,InputExp10,3600); SystemEperiment10.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment10.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment10.OutputName="; SystemEperiment10.OutputUnit='mL'; %Experiment 11 InputExp11=ones(25,4); InputExp11(1:end,1)=39.\*InputExp11(1:end,1); %Temperature (C) InputExp11(1:end,2)=7.1.\*InputExp11(1:end,2); %pH InputExp11(1:end,3)=60.\*InputExp11(1:end,3); % Mixing speed (rpm) InputExp11(1:end,4)=0.7.\*InputExp11(1:end,4); %Pressure (bar) OutputExp11=ms60; SystemEperiment11=iddata(OutputExp11,InputExp11,3600);

```
SystemEperiment11.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
```

SystemEperiment11.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment11.OutputName="; SystemEperiment11.OutputUnit='mL'; %Experiment 12 InputExp12=ones(25,4); InputExp12(1:end,1)=39.\*InputExp12(1:end,1); %Temperature (C) InputExp12(1:end,2)=7.1.\*InputExp12(1:end,2); %pH InputExp12(1:end,3)=300.\*InputExp12(1:end,3); %Mixing speed (rpm) InputExp12(1:end,4)=0.7.\*InputExp12(1:end,4); %Pressure (bar) OutputExp12=ms300; SystemEperiment12=iddata(OutputExp12,InputExp12,3600); SystemEperiment12.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment12.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment12.OutputName="; SystemEperiment12.OutputUnit='mL'; %Experiment 13 InputExp13=ones(25,4); InputExp13(1:end,1)=39.\*InputExp13(1:end,1); %Temperature (C) InputExp13(1:end,2)=7.1.\*InputExp13(1:end,2); %pH %Mixing speed (rpm) InputExp13(1:end,3)=600.\*InputExp13(1:end,3); InputExp13(1:end,4)=0.7.\*InputExp13(1:end,4); %Pressure (bar) OutputExp13=ms600; SystemEperiment13=iddata(OutputExp13,InputExp13,3600); SystemEperiment13.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment13.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment13.OutputName="; SystemEperiment13.OutputUnit='mL'; %Experiment 14 InputExp14=ones(25,4); InputExp14(1:end,1)=39.\*InputExp14(1:end,1); %Temperature (C) InputExp14(1:end,2)=7.1.\*InputExp14(1:end,2); %pH InputExp14(1:end,3)=2000.\*InputExp14(1:end,3); %Mixing speed (rpm) InputExp14(1:end,4)=0.7.\*InputExp14(1:end,4); %Pressure (bar) OutputExp14=ms2000; SystemEperiment14=iddata(OutputExp14,InputExp14,3600); SystemEperiment14.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment14.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment14.OutputName="; SystemEperiment14.OutputUnit='mL'; %Experiment 15 InputExp15=ones(25,4); InputExp15(1:end,1)=39.\*InputExp15(1:end,1); %Temperature (C) InputExp15(1:end,2)=7.1.\*InputExp15(1:end,2); %pH InputExp15(1:end,3)=250.\*InputExp15(1:end,3); %Mixing speed (rpm) InputExp15(1:end,4)=0.1.\*InputExp15(1:end,4); %Pressure (bar) OutputExp15=pressure100;

```
SystemEperiment15=iddata(OutputExp15,InputExp15,3600);
SystemEperiment15.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment15.inputUnit={'C', '', 'rpm', 'bar'};
SystemEperiment15.OutputName=";
SystemEperiment15.OutputUnit='mL';
%Experiment 16
InputExp16=ones(25,4);
InputExp16(1:end,1)=39.*InputExp16(1:end,1);
                                               %Temperature (C)
InputExp16(1:end,2)=7.1.*InputExp16(1:end,2);
                                               %pH
InputExp16(1:end,3)=250.*InputExp16(1:end,3);
                                                %Mixing speed (rpm)
InputExp16(1:end,4)=0.3.*InputExp16(1:end,4);
                                               %Pressure (bar)
OutputExp16=pressure300;
SystemEperiment16=iddata(OutputExp16,InputExp16,3600);
SystemEperiment16.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment16.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment16.OutputName=";
SystemEperiment16.OutputUnit='mL';
%Experiment 17
InputExp17=ones(25,4);
InputExp17(1:end,1)=39.*InputExp17(1:end,1);
                                               %Temperature (C)
InputExp17(1:end,2)=7.1.*InputExp17(1:end,2);
                                               %pH
InputExp17(1:end,3)=250.*InputExp17(1:end,3);
                                                %Mixing speed (rpm)
InputExp17(1:end,4)=0.5.*InputExp17(1:end,4);
                                               %Pressure (bar)
OutputExp17=pressure500;
SystemEperiment17=iddata(OutputExp17,InputExp17,3600);
SystemEperiment17.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment17.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment17.OutputName=";
SystemEperiment17.OutputUnit='mL';
%Experiment 18
InputExp18=ones(25,4);
InputExp18(1:end,1)=39.*InputExp18(1:end,1);
                                               %Temperature (C)
InputExp18(1:end,2)=7.1.*InputExp18(1:end,2);
                                               %pH
InputExp18(1:end,3)=250.*InputExp18(1:end,3);
                                                %Mixing speed (rpm)
InputExp18(1:end,4)=0.7.*InputExp18(1:end,4);
                                               %Pressure (bar)
OutputExp18=pressure700;
SystemEperiment18=iddata(OutputExp18,InputExp18,3600);
SystemEperiment18.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'};
SystemEperiment18.inputUnit={'C', ' ', 'rpm', 'bar'};
SystemEperiment18.OutputName=";
SystemEperiment18.OutputUnit='mL';
%Experiment 19
InputExp19=ones(25,4);
InputExp19(1:end,1)=39.*InputExp19(1:end,1);
                                               %Temperature (C)
InputExp19(1:end,2)=7.1.*InputExp19(1:end,2);
                                               %pH
InputExp19(1:end,3)=270.*InputExp19(1:end,3);
                                                %Mixing speed (rpm)
```

InputExp19(1:end,4)=0.7.\*InputExp19(1:end,4); %Pressure (bar) OutputExp19=ms270; SystemEperiment19=iddata(OutputExp19,InputExp19,3600); SystemEperiment19.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment19.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment19.OutputName="; SystemEperiment19.OutputUnit='mL'; %Experiment 20 InputExp20=ones(25,4); InputExp20(1:end,1)=39.\*InputExp20(1:end,1); %Temperature (C) InputExp20(1:end,2)=7.1.\*InputExp20(1:end,2); %pH InputExp20(1:end,3)=250.\*InputExp20(1:end,3); %Mixing speed (rpm) InputExp20(1:end,4)=0.6.\*InputExp20(1:end,4); %Pressure (bar) OutputExp20=pressure600; SystemEperiment20=iddata(OutputExp20,InputExp20,3600); SystemEperiment20.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment20.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment20.OutputName="; SystemEperiment20.OutputUnit='mL'; %Experiment 21 InputExp21=ones(25,4); InputExp21(1:end,1)=39.\*InputExp21(1:end,1); %Temperature (C) InputExp21(1:end,2)=(6.5).\*InputExp21(1:end,2); %pH InputExp21(1:end,3)=250.\*InputExp21(1:end,3); %Mixing speed (rpm) InputExp21(1:end,4)=0.7.\*InputExp21(1:end,4); %Pressure (bar) OutputExp21=pH65; SystemEperiment21=iddata(OutputExp21,InputExp21,3600); SystemEperiment21.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment21.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment21.OutputName="; SystemEperiment21.OutputUnit='mL'; %Experiment 22 InputExp22=ones(25,4); InputExp22(1:end,1)=39.\*InputExp22(1:end,1); %Temperature (C) InputExp22(1:end,2)=(7.5).\*InputExp22(1:end,2); %pH InputExp22(1:end,3)=250.\*InputExp22(1:end,3); %Mixing speed (rpm) InputExp22(1:end,4)=0.7.\*InputExp22(1:end,4); %Pressure (bar) OutputExp22=pH75; SystemEperiment22=iddata(OutputExp22,InputExp22,3600); SystemEperiment22.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment22.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment22.OutputName="; SystemEperiment22.OutputUnit='mL'; %Experiment 23 InputExp23=ones(25,4);

%Temperature (C)

InputExp23(1:end,1)=39.\*InputExp23(1:end,1);

InputExp23(1:end,2)=7.1.\*InputExp23(1:end,2); %pH InputExp23(1:end,3)=500.\*InputExp23(1:end,3); %Mixing speed (rpm) InputExp23(1:end,4)=0.7.\*InputExp23(1:end,4); %Pressure (bar) OutputExp23=ms500; SystemEperiment23=iddata(OutputExp23,InputExp23,3600); SystemEperiment23.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment23.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment23.OutputName="; SystemEperiment23.OutputUnit='mL'; %Experiment 24 InputExp24=ones(25,4); InputExp24(1:end,1)=39.\*InputExp24(1:end,1); %Temperature (C) InputExp24(1:end,2)=7.1.\*InputExp24(1:end,2); %pH InputExp24(1:end,3)=250.\*InputExp24(1:end,3); %Mixing speed (rpm) InputExp24(1:end,4)=0.2.\*InputExp24(1:end,4); %Pressure (bar) OutputExp24=pressure200; SystemEperiment24=iddata(OutputExp24,InputExp24,3600); SystemEperiment24.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment24.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment24.OutputName="; SystemEperiment24.OutputUnit='mL'; % Experiment 25 InputExp25=ones(25,4); InputExp25(1:end,1)=35.\*InputExp25(1:end,1); %Temperature (C) InputExp25(1:end,2)=7.1.\*InputExp25(1:end,2); %pH InputExp25(1:end,3)=250.\*InputExp25(1:end,3); % Mixing speed (rpm) %Pressure (bar) InputExp25(1:end,4)=0.7.\*InputExp25(1:end,4); OutputExp25=Temperature35; SystemEperiment25=iddata(OutputExp25,InputExp25,3600); SystemEperiment25.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment25.inputUnit={'C', '', 'rpm', 'bar'}; SystemEperiment25.OutputName="; SystemEperiment25.OutputUnit='mL'; % Experiment 26 InputExp26=ones(25,4); InputExp26(1:end,1)=37.\*InputExp26(1:end,1); %Temperature (C) InputExp26(1:end,2)=7.1.\*InputExp26(1:end,2); %pH InputExp26(1:end,3)=250.\*InputExp26(1:end,3); %Mixing speed (rpm) InputExp26(1:end,4)=0.7.\*InputExp26(1:end,4); %Pressure (bar) OutputExp26=Temperature37; SystemEperiment26=iddata(OutputExp26,InputExp26,3600); SystemEperiment26.InputName={'Temperature', 'pH', 'Mixing speed', 'Pressure'}; SystemEperiment26.inputUnit={'C', ' ', 'rpm', 'bar'}; SystemEperiment26.OutputName="; SystemEperiment26.OutputUnit='mL';

% Using KI, K2, K3 as estimation data sets and K6 as validation data set % merging all estimation iddata to a single iddata object;

% This is for all the measurements that will be considered for the model extraction.

Overall\_iddata=merge(SystemEperiment1, SystemEperiment8, SystemEperiment11, SystemEperiment18, SystemEperiment19, SystemEperiment25, SystemEperiment16, SystemEperiment24, SystemEperiment26);

SystemEperiment2, SystemEperiment3, SystemEperiment12, SystemEperiment17, SystemEperiment20, SystemEperiment21, SystemEperiment22, SystemEperiment23,

#### %VALIDATION PROCESS

% merging data to a single iddata object:

00	0 3		
%This is for all the me	asurements that will be	considered for the model	extraction.
Overall_iddata_val=me	erge(SystemEperiment4	,SystemEperiment5,	SystemEperiment6,
SystemEperiment9,	SystemEperiment10,	SystemEperiment13,	SystemEperiment14,
SystemEperiment15);			
%Overall_iddata=merg	ge(SystemEperiment1,S	ystemEperiment2,	SystemEperiment3,
SystemEperiment7,	SystemEperiment8,	SystemEperiment11,	SystemEperiment12,
SystemEperiment17,	SystemEperiment18,	SystemEperiment19,	SystemEperiment20,
SystemEperiment21,Sy	ystemEperiment25,	SystemEperimer	nt4,SystemEperiment5,
SystemEperiment6,	SystemEperiment9,	SystemEperiment10,	SystemEperiment13,
SystemEperiment14,	SystemEperiment15,	SystemEperiment16,	SystemEperiment22,
SystemEperiment23, S	ystemEperiment24,Syst	emEperiment26);	

%Estimation data and validation data have been entered and two iddata objects have been created:

%Overall iddata contains the experiment results for the model identification process. %Overall\_iddata\_val contains the experiment results for the model validation.

ze= Overall iddata;

zv=(Overall\_iddata\_val);

#### %NALRX MODEL

%ms1 = nlarx(ze, [ones(1,1), ones(1,4), ones(1,4)], wavenet)

%ms2 = nlarx(ze, [ones(1,1), ones(1,4), 2\*ones(1,4)], wavenet)

%ms3 = nlarx(ze, [2\*ones(1,1), ones(1,4), 2\*ones(1,4)], wavenet);

%ms4 = nlarx(ze, [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)], wavenet)

%ms5 = nlarx(ze, [3\*ones(1,1), ones(1,4), ones(1,4)], wavenet)

%ms6 = nlarx(ze, [3\*ones(1,1), 3\*ones(1,4), ones(1,4)], wavenet)

%ompare(zv,ms1, ms2, ms3, ms4, ms5, ms6)

% The numbers of units (wavelets) of the two WAVENET estimators have been automatically chosen by the estimation algorithm. These numbers are displayed below. Notice the abbreviations 'nl'='Nonlinearity' and 'num'='NumberOfUnits' %mx4.Nonlinearity(1).NumberOfUnits %using full property names

%nanbnk = [2\*ones(1,1), 2\*ones(1,4), 2\*ones(1,4)];

%The numbers of units (wavelets) of the two WAVENET estimators have been automatically chosen by the estimation algorithm. These numbers are displayed below. Notice the abbreviations 'nl'='Nonlinearity' and 'num'='NumberOfUnits' %mx4.Nonlinearity(1).NumberOfUnits %using full property names

%The number of units in the WAVENET estimators can be explicitly specified instead of being automatically chosen by the estimation algorithm: %ms7 = nlarx(ze, nanbnk, [wavenet('num',9)]) %ms8 = nlarx(ze, nanbnk, [wavenet('num',11)]) %ms9 = nlarx(ze, nanbnk, [wavenet('num',12)])

%compare(zv,ms7, ms8, ms9)

%Nonlinear ARX Model - Trying Other Nonlinearity Estimators

%ms10 = nlarx(ze, nanbnk, treepartition);

%The SIGMOIDNET estimator can also be used. Estimation options such as maximum iterations (MaxIter) and iteration display can be specified using NLARXOPTIONS command. %opt = nlarxOptions('Display','on'); %opt.SearchOption.MaxIter = 2; %ms11 = nlarx(ze, nanbnk, sigmoidnet);

%compare(zv,ms7, ms10, ms11)

%Nonlinear ARX Model - Trying Other Nonlinearity Estimators

%ms2 = nlarx(ze, nanbnk, treepartition);

```
%The SIGMOIDNET estimator can also be used. Estimation options such as maximum iterations (MaxIter) and iteration display can be specified using NLARXOPTIONS command. %opt = nlarxOptions('Display','on');
%opt.SearchOption.MaxIter = 2;
%ms3 = nlarx(ze, nanbnk, sigmoidnet);
```

%compare(zv, ms2, ms3)

%ms4 = nlarx(ze, nanbnk, sigmoidnet ('NumberOfUnits',5));

%compare(zv, ms3, ms4)
%Function PLOT may be used to view the nonlinearity responses of various models. %plot(mx1, mx2, mx3)

%NLHW Model

%mhws1=nlhw(ze, [2\*ones(1,4), 3\*ones(1,4), zeros(1,4)], 'saturation', 'deadzone'); %compare(zv, ms1, mhws1)

%mhws1 = nlhw(ze, [ones(1,4), ones(1,4), zeros(1,4)], 'pwlinear', 'pwlinear'); mhws2 = nlhw(ze, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'pwlinear', 'pwlinear'); %mhws3 = nlhw(ze, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'pwlinear', 'pwlinear');

%compare(zv,mhws1, mhws2, mhws3) %compare(zv,mhws2) %mhws4 = nlhw(ze, [ones(1,4), ones(1,4), zeros(1,4)], 'unitgain', 'deadzone'); %mhws5 = nlhw(ze, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'unitgain', 'deadzone'); %mhws6 = nlhw(ze, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'unitgain', 'deadzone');

%compare(zv,mhws4, mhws5, mhws6)

%mhws7 = nlhw(ze, [ones(1,4), ones(1,4), zeros(1,4)], 'saturation', 'deadzone'); %mhws8 = nlhw(ze, [2\*ones(1,4), 2\*ones(1,4), 2\*zeros(1,4)], 'saturation', 'deadzone'); %mhws9 = nlhw(ze, [3\*ones(1,4), ones(1,4), zeros(1,4)], 'saturation', 'deadzone');

%compare(zv,mhws2, mhws5, mhws8) %compare(zv,ms7,mhws2)

%AIC = aic (ms7, mhws2)

Appendix C: Published Conference Paper